# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

# CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): December 21, 2012

# **Lehigh Gas Partners LP**

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

001-35711

(Commission File Number)

45-4165414

(IRS Employer Identification No.)

702 West Hamilton Street, Suite 203 Allentown, PA 18101

(Address of principal executive office) (Zip Code)

(610) 625-8000

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

# Item 1.01 Entry into a Material Definitive Agreement.

The information set forth under the caption "Acquisition of Express Lane, Inc." under "Item 2.01 Completion of Acquisition or Disposition of Assets" is incorporated into this Item 1.01 by reference.

#### Item 2.01 Completion of Acquisition or Disposition of Assets.

Acquisition of Express Lane, Inc.

On December 21, 2012, Lehigh Gas Wholesale Services, Inc. (the "Express Lane Purchaser"), a wholly-owned subsidiary of Lehigh Gas Partners LP (the "Partnership"), entered into a Stock Purchase Agreement (the "Express Lane Stock Purchase Agreement") with James E. Lewis, Jr., Lida N. Lewis, James E. Lewis, III and Reid D. Lewis (collectively, the "Express Lane Sellers"), pursuant to which the Express Lane Sellers agreed to sell to the Express Lane Purchaser all of the outstanding capital stock (collectively, the "Express Lane Shares") of Express Lane, Inc. ("Express Lane"), the owner and operator of various retail convenience stores, which include the retail sale of motor fuels and quick service restaurants, at various locations in Florida. In connection with the purchase of the Express Lane Shares, the Express Lane Purchaser agreed to acquire thirty-nine motor fuel service stations, one as a fee simple interest and thirty-eight as leasehold interests.

In connection with the purchase of the Express Lane Shares, on December 21, 2012, LGP Realty Holdings LP, a wholly-owned subsidiary of the Partnership ("LGP-R"), entered into a Purchase and Sale Agreement (the "Express Lane Purchase and Sale Agreement" and, together with the Express Lane Stock Purchase Agreement, the "Express Lane Agreements") with Express Lane. Under the Express Lane Purchase and Sale Agreement, LGP-R agreed to acquire from Express Lane, prior to the Express Lane Purchaser's acquisition of the Express Lane Shares, an additional fee simple interest in six properties and two fueling agreements (collectively, the "Express Lane Property").

On December 21, 2012, LGP-R completed the acquisition of the Express Lane Property from the Express Lane Sellers, as contemplated by the Express Lane Purchase and Sale Agreement. In addition, on December 22, 2012, the Express Lane Purchaser completed (the "Express Lane Closing") the acquisition of the Express Lane Shares from the Express Lane Sellers, as contemplated by the Express Lane Stock Purchase Agreement.

From and after the date of the Express Lane Closing, Lehigh Gas — Ohio, LLC ("LGO"), an entity managed by Joseph V. Topper, Jr., the Chief Executive Officer and the Chairman of the board of directors of Lehigh Gas GP LLC ("Lehigh Gas GP"), the general partner of the Partnership, will operate Express

Lane's gasoline and diesel retail outlet business and its related convenience store business (the "Express Lane Retail Business"). In addition, certain of the assets related to the Express Lane Retail Business and certain liabilities of the Express Lane Sellers were assigned by the Partnership to LGO. LGO paid the Partnership \$1,000,000 in exchange for such assets, subject to certain post-closing adjustments.

Pursuant to the PMPA Franchise Agreement (which was previously disclosed by the Partnership in its Current Report on Form 8-K filed with the U.S. Securities and Exchange Commission (the "SEC") on October 30, 2012) by and between LGO and the Express Lane Purchaser (the "Franchise Agreement"), the Partnership will be the exclusive distributor of motor fuels to all sites operated by LGO in connection with the Express Lane Retail Business. In addition, the Partnership will lease these sites to LGO pursuant to property lease agreements. The Partnership estimates it will receive from LGO aggregate rental income, net of expenses, of approximately \$4.6 million per year from such sites.

Under the Express Lane Agreements, the aggregate purchase price (the "Express Lane Purchase Price") for the Express Lane Property and the Express Lane Shares is \$43,000,000, subject to certain post-closing adjustments. Of the Express Lane Purchase Price, the Express Lane Purchaser paid an aggregate of \$41,868,500 to the Express Lane Sellers and placed an aggregate of \$1,754,849 into escrow, of which \$1,000,000 has been placed into escrow to fund any indemnification or similar claims made under the Express Lane Agreements by the parties thereto, and \$131,500 has been placed into escrow pending the completion by the Express Lane Sellers of certain environmental remediation measures. In addition to the Express Lane Purchase Price, the Express Lane Purchaser also placed \$623,349 (the "Tax Escrow") into escrow to indemnify the Express Lane Sellers for certain tax obligations resulting from the sale of the Express Lane Property, as more fully described below.

1

Pursuant to the Express Lane Stock Purchase Agreement, the Express Lane Purchaser is required to indemnify the Express Lane Sellers for any amount of federal income taxes incurred by Express Lane Sellers on their share of the income or gain resulting from the sale of the Express Lane Property that is in excess of fifteen percent of the Express Lane Purchase Price attributable to the Express Lane Property, plus an amount equal to the tax obligations of the Express Lane Sellers as the result of their receipt of such payments (collectively, the "Total Shareholder Taxes"). The Express Lane Purchaser paid the Tax Escrow into escrow, representing a preliminary estimate of the Total Shareholder Taxes. If the actual Total Shareholder Taxes calculated in accordance with the Express Lane Stock Purchase Agreement (the "Actual Total Shareholder Taxes") are equal to or exceed the Tax Escrow, then, on or before April 10, 2013, the Tax Escrow will be released to the Express Lane Sellers and the Express Lane Purchaser will be required to pay to the Express Lane Sellers any such excess. If the Tax Escrow exceeds the Actual Total Shareholder Taxes, then, on or before April 10, 2013, an amount equal to the Actual Total Shareholder Taxes will be released to the Express Lane Sellers from the Tax Escrow, and the remainder of the Tax Escrow will be returned to the Express Lane Purchaser.

Under the Express Lane Stock Purchase Agreement, the Express Lane Sellers have agreed not to compete in the retail motor fuel or convenience store business within the State of Florida for a period of four years following the Express Lane Closing. In addition, pursuant to the Express Lane Stock Purchase Agreement, each of the Express Lane Sellers executed a general release in favor of the Express Lane Purchaser, Express Lane and their respective affiliates.

The Express Lane Agreements contain customary representations, warranties, indemnity provisions, and agreements and obligations of the respective parties thereto.

All of the transactions between the Partnership and LGO that are described in this report under the caption "Acquisition of Express Lane, Inc." have been approved by the conflicts committee of the board of directors of Lehigh Gas GP.

The foregoing description of the Express Lane Stock Purchase Agreement and of the Express Lane Purchase and Sale Agreement does not purport to be complete and is qualified in its entirety by reference to the Express Lane Stock Purchase Agreement and to the Express Lane Purchase and Sale Agreement, which are attached as Exhibits 2.1 and 2.2, respectively, to this Current Report on Form 8-K and are incorporated into this report by reference. The registrant has omitted schedules, exhibits and similar attachments to the Express Lane Agreements pursuant to Item 601(b)(2) of Regulation S-K. The registrant will furnish a copy of any omitted schedule, exhibit or similar attachment to the SEC upon request.

The Express Lane Agreements and the above description thereof have been included to provide investors and securityholders with information regarding the terms of the Express Lane Agreements. It is not intended to provide any other factual information about the Partnership, the Express Lane Purchaser, LGP-R, Express Lane, the Express Lane Sellers, or their respective subsidiaries or affiliates. The Express Lane Agreements contain representations and warranties of each of the parties thereto, made solely for the benefit of the other parties thereto. The assertions embodied in those representations and warranties are qualified by information in disclosure schedules that the parties to the Express Lane Agreements have exchanged in connection with the execution of the Express Lane Agreements. The disclosure schedules to the Express Lane Agreements contain information that modifies, qualifies and creates exceptions to the representations and warranties set forth in the Express Lane Agreements. Moreover, the

2

representations and warranties in the Express Lane Agreements were used for the purpose of allocating risk among the parties thereto.

Acquisition of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc.

On December 21, 2012, the Partnership completed (the "Dunmore Closing") the acquisition of certain assets of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc. (together, the "Dunmore Sellers") as contemplated by the Asset Purchase Agreement (the "Dunmore Initial Agreement"), dated November 30, 2012, by and among the Partnership, LGP-R, the Dunmore Sellers, and, for limited purposes, Joseph Gentile, Jr., and the amendments thereto, dated December 4, 2012 (the "Dunmore First Amendment"), December 13, 2012 and December 21, 2012 (the Dunmore Initial Agreement, together with all amendments thereto, the "Dunmore Purchase Agreement"). Pursuant to the Dunmore Purchase Agreement, the Dunmore Sellers sold to the Partnership all of the assets (collectively, the "Dunmore Assets") held and used by the Dunmore Sellers in connection with their gasoline and diesel retail outlet business and their related convenience store business (the "Dunmore Retail Business").

From and after the date of the Dunmore Closing, LGO will operate the Dunmore Retail Business. In addition, as contemplated by the Dunmore Purchase Agreement, certain of the Dunmore Assets and certain liabilities of the Dunmore Sellers were assigned by the Partnership to LGO. LGO paid the Dunmore Sellers \$500,000 in exchange for such assets, subject to certain post-closing adjustments.

Pursuant to the Franchise Agreement, the Partnership will be the exclusive distributor of motor fuels to all sites operated by LGO in connection with the Dunmore Retail Business. In addition, the Partnership will lease these sites to LGO pursuant to property lease agreements. The Partnership estimates it will receive from LGO aggregate rental income, net of expenses, of approximately \$1.7 million per year from such sites.

As consideration for the Dunmore Assets, the Partnership paid (i) \$27,500,000 in cash to the Dunmore Sellers; (ii) \$500,000 in cash to Mr. Gentile as consideration for his agreeing, for a period of five years following the Dunmore Closing, to not compete in the Dunmore Retail Business, to not engage in the sale or distribution of branded motor fuels, and to not solicit or hire any of the Partnership's or its affiliates' employees; and (iii) \$500,000 in cash to be held in escrow and delivered to the Dunmore Sellers upon the Partnership's receipt of written evidence concerning the payment of certain of the Dunmore Sellers' pre-closing tax liabilities (collectively, the "Dunmore Purchase Price").

All of the transactions between the Partnership and LGO that are described in this report under the caption "Acquisition of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc." have been approved by the conflicts committee of the board of directors of Lehigh Gas GP.

Copies of the Dunmore Initial Agreement and the Dunmore First Amendment were previously filed as Exhibits 2.1 and 2.2, respectively, to the Partnership's Current Report on Form 8-K filed with the SEC on December 5, 2012.

3

#### Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

As previously disclosed by the Partnership in its Current Report on Form 8-K filed with the SEC on October 30, 2012, the Partnership entered into a Second Amended and Restated Credit Agreement, dated October 30, 2012, among the Partnership, as borrower, KeyBank National Association, as Administrative Agent, as Collateral Agent, as L/C Issuer, as Joint Lead Arranger and as Joint Book Runner, RBS Citizens, N.A., as Joint Lead Arranger and Joint Book Runner, and Citizens Bank of Pennsylvania, as Syndication Agent, and the other lenders party thereto (the "New Credit Agreement"). The New Credit Agreement includes a \$249 million senior secured revolving credit facility, expiring October 30, 2015 (the "Credit Facility").

Substantially all of the aggregate purchase price of \$71,500,000 for the acquisitions of Express Lane and the Dunmore Sellers, as described under Item 2.01 above, was funded by a drawing under the Credit Facility.

The foregoing description of the New Credit Agreement does not purport to be complete and is qualified in its entirety by reference to the New Credit Agreement, which is attached as Exhibit 10.1 to the Partnership's Current Report on Form 8-K filed with the SEC on October 30, 2012 and is incorporated into this report by reference.

#### Forward-Looking Statements

This Current Report on Form 8-K contains "forward-looking statements," such as statements regarding the anticipated impact of acquiring Express Lane and the Dunmore Sellers. These forward-looking statements are based on current plans and expectations and involve a number of risks and uncertainties that could cause actual results and events to vary materially. For a full discussion of these risks and uncertainties, please refer to the "Risk Factors" section of the Partnership's Registration Statement on Form S-1 filed with the SEC on May 11, 2012 and the information included in subsequent amendments and other filings. These forward-looking statements are based on and include the Partnership's expectations as of the date hereof. Subsequent events and market developments could cause the Partnership expectations to change. While the Partnership may elect to update these forward-looking statements at some point in the future, the Partnership specifically disclaims any obligation to do so, even if new information becomes available, except as may be required by applicable law.

# Item 7.01 Regulation FD Disclosure.

On December 24, 2012, the Partnership issued a press release related to the acquisition of the outstanding capital stock of Express Lane, Inc., as described under Item 2.01 above, which is furnished herewith as Exhibit 99.1. In addition, on December 24, 2012, the Partnership issued a press release related to the acquisition of certain assets of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc., as described under Item 2.01 above, which is furnished herewith as Exhibit 99.2.

The information in the preceding paragraph, as well as Exhibits 99.1 and 99.2 referenced therein, shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, nor shall it be deemed incorporated by reference in filings under the Securities Act of 1933, as amended.

#### Item 9.01 Financial Statements and Exhibits.

#### (a) Financial Statements of a Business Acquired

This Current Report on Form 8-K will be supplemented by an amendment to provide the required financial statements in connection with the Partnership's acquisition of Express Lane, as described under Item 2.01 above, not later than 71 days after the date that this Current Report on Form 8-K was required to be filed.

#### (b) Pro Forma Financial Information

This Current Report on Form 8-K will be supplemented by an amendment to provide the required pro forma financial information in connection with the Partnership's acquisition of Express Lane, as described under Item 2.01 above, not later than 71 days after the date that this Current Report on Form 8-K was required to be filed.

# (d) Exhibits

EXMIDIT	
Number	Description

2.2	Purchase and Sale Agreement, dated December 21, 2012, by and between Express Lane, Inc. and LGP Realty Holdings LP.
99.1	Press Release regarding the Acquisition of the Outstanding Capital Stock of Express Lane, Inc., dated December 24, 2012.
99.2	Press Release regarding the Acquisition of Certain Assets of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc., dated December 24, 2012.

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# Lehigh Gas Partners LP

By: Lehigh Gas GP LLC, its general partner

Dated: December 26, 2012

Lewis, James E. Lewis, III and Reid D. Lewis.

By: /s/ Mark L. Miller

Name: Mark L. Miller Title: Chief Financial Officer

5

# EXHIBIT INDEX

Exhibit Number	Description			
2.1	Stock Purchase Agreement, dated December 21, 2012, by and among Lehigh Gas Wholesale Services, Inc., James E. Lewis, Jr., Lida N. Lewis, James E. Lewis, III and Reid D. Lewis.			
2.2	Purchase and Sale Agreement, dated December 21, 2012, by and between Express Lane, Inc. and LGP Realty Holdings LP.			
99.1	Press Release regarding the Acquisition of the Outstanding Capital Stock of Express Lane, Inc., dated December 24, 2012.			
Press Release regarding the Acquisition of Certain Assets of Dunmore Oil Company, Inc. and Jojo Oil Company, Inc., dated 2012.				
	6			

#### STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT (the "Agreement") is entered into as of the 21st day of December, 2012 (the "Effective Date"), by and among Lehigh Gas Wholesale Services Inc., a Delaware corporation (the "Purchaser"), and James E. Lewis, Jr., Lida N. Lewis, James E. Lewis, III and Reid D. Lewis (collectively, the "Sellers").

#### WITNESSETH:

**WHEREAS,** Sellers own in the aggregate all the outstanding capital stock of Express Lane, Inc., a Florida corporation ("Company"), comprising 100 shares of voting common stock, \$10.00 par value, and 4,862 shares of non-voting common stock, \$10.00 par value (collectively, the "Shares");

**WHEREAS,** Company presently owns and operates various retail convenience stores, which include the retail sale of motor fuels and QSRs, at various locations in Florida (collectively the "Locations"); and

**WHEREAS,** Purchaser desires to purchase from Sellers, and Sellers desire to sell to Purchaser, the Shares, subject to terms and conditions contained herein.

NOW, THEREFORE, in consideration of the premises and the mutual representations, warranties, covenants and agreements set forth above and in the body of this Agreement, upon the terms and subject to the conditions hereinabove and hereinafter set forth, the Parties hereto agree as follows:

#### I. PURCHASE AND SALE OF SHARES

- 1.1 <u>Purchase and Sale of Shares</u>. Purchaser agrees to purchase from Sellers, and Sellers agree to sell to Purchaser, all the Shares of the Company for an aggregate purchase price set forth in Section 2.1 below.
  - 1.2 <u>Definitions</u>. Capitalized terms not defined elsewhere in this Agreement are defined in Article XIII.

#### II. PURCHASE PRICE AND ESCROW

2.1 <u>Purchase Price</u>. As consideration for the Shares, Purchaser shall pay to Sellers the sum of Twenty-five Million U.S. Dollars (\$25,000,000) plus a Tax Gross-Up Amount (the "Share Purchase Price") subject to adjustment as provided in Section 2.2 (the Share Purchase Price as adjusted shall be referred to as the "Adjusted Share Purchase Price").

#### 2.2 Post-Closing Adjustment of the Share Purchase Price.

(a) After the Closing, the Share Purchase Price shall be adjusted upward or downward based upon the difference between (A) the Current Assets at Closing and (B) the Current Liabilities at Closing determined as follows: (i) if the Current Assets at Closing shall exceed the Current Liabilities at Closing, then the Share Purchase Price shall be increased by the

amount of such difference; or (ii) if the Current Liabilities at Closing shall exceed the Current Assets at Closing, then the Share Purchase Price shall be reduced by such difference. Any post-Closing adjustment to the Share Purchase Price shall be paid within three (3) business days of final determination under Section 2.2(b).

Within sixty (60) days following the Closing Date, Purchaser shall deliver to Sellers a balance sheet of the Company as of the Closing Date (in its final and binding form, the "Closing Balance Sheet"), setting forth the Current Assets and Current Liabilities calculated in accordance with GAAP, consistently applied, and a certificate setting forth the resulting Share Purchase Price calculated with reference to such amounts (in its final and binding form, together with the Closing Balance Sheet, the "Share Purchase Price Calculation"). During the 20-day period immediately following the Sellers' receipt of the Share Purchase Price Calculation, the Sellers shall be permitted to review the Company's books and records and the Purchaser's working papers related to the preparation of the Share Purchase Price Calculation and determination of the Share Purchase Price. The Share Purchase Price Calculation shall become final and binding upon the Parties twenty (20) days following the Sellers' receipt thereof unless the Sellers give written notice of its disagreement (a "Notice of Disagreement") to Sellers prior to such date. Any Notice of Disagreement shall specify in reasonable detail the nature and dollar amount of any disagreement so asserted and shall be delivered only if (and to the extent that) the Sellers reasonably and in good faith determine that the Share Purchase Price Calculation and the resulting Share Purchase Price calculated with reference thereto delivered by Purchaser have not been determined in accordance with the guidelines and procedures set forth in this Agreement. If a timely Notice of Disagreement is received by Purchaser, then the Share Purchase Price Calculation (as revised in accordance with clause (x) or (y) below) shall become final and binding upon the Parties on the earlier of (x) the date the Parties resolve in writing any differences they have with respect to the matters specified in the Notice of Disagreement, or (y) the date all matters in dispute are finally resolved in writing by the Independent Auditor. During the twenty (20) days following delivery of a Notice of Disagreement, the Parties shall seek in good faith to resolve in writing any differences that they have with respect to the matters specified in the Notice of Disagreement. Following delivery of a Notice of Disagreement, Purchaser and its agents and representatives shall be permitted to review the Sellers' and their representatives' working papers relating to the Notice of Disagreement. At the end of the 20-day period referred to above, the Parties shall submit to the Independent Auditor for review and resolution all matters (but only such matters) that remain in dispute and that were properly included in the Notice of Disagreement. The Parties shall instruct the Independent Auditor to make a final determination of the Current Assets and Current Liabilities and the resulting Share Purchase Price calculated with reference to such amount to the extent such amount is in dispute, in accordance with the guidelines and procedures set forth in this Agreement. The Parties will cooperate with the Independent Auditor during the term of its engagement and shall instruct the Independent Auditor not to assign a value to any item in dispute greater than the greatest value for such item assigned by the Purchaser, on the one hand, or Sellers, on the other hand, or less than the smallest value for such item assigned by the Purchaser, on the one hand, or Sellers, on the other hand. The Parties also shall instruct the Independent Auditor to make its determination based solely on presentations by the Purchaser and Sellers that are in accordance with the guidelines and procedures set forth in this Agreement (i.e., not on the basis of an independent review). The Share Purchase Price Calculation and the determination of the Current Assets and Current Liabilities and the resulting Adjusted Share Purchase Price calculated with reference

thereto shall become final and binding on the Parties on the date the Independent Auditor delivers its final resolution in writing to the Parties (which final resolution shall be requested by the Parties to be delivered not more than 45 days following submission of such disputed matters). The fees and expenses of the Independent Auditor shall be allocated to the Parties as determined (and as set forth in the final determination) by the Independent Auditor based upon the relative success (in terms of percentages) of each Party's claims. For example, if the final determination reflects a 60/40 compromise of the Parties' claims, the Independent Auditor would allocate expenses 40% to the Party whose claims were determined to be 60% successful and 60% to the Party whose claims were determined to be 40% successful.

- 2.3 <u>Deposit</u>. Of the Share Purchase Price, the Company acknowledges receipt of an earnest money deposit in the amount of Two Million Dollars (\$2,000,000) (the "Initial Deposit"). Furthermore, within two (2) business days after receipt by Purchaser of this Agreement executed by Sellers, Purchaser shall deliver an additional deposit equal to Two Million Three Hundred Thousand Dollars (\$2,300,000) (the "Additional Deposit") (the Initial Deposit and the Additional Deposit shall be referred to hereinafter collectively as the "Deposit") to First American Title Insurance Company, Attention: Larry Deal, as escrow agent (the "Escrow Agent"), by wire transfer to an account designated by Escrow Agent. The Additional Deposit shall be held by the Escrow Agent in escrow, subject to the terms of an escrow agreement substantially in the form attached hereto as <u>Exhibit A</u> (the "Escrow Agreement"). If the Closing does not occur, for any reason, the Deposit will be distributed according to Article XI hereof. All accrued interest on the Additional Deposit will be delivered to Purchaser when the Deposit is distributed pursuant to Section 4.4 or Article XI.
- 2.4 <u>Form of Payment</u>. All payments by Purchaser to Sellers under this Agreement at Closing shall be made by wire transfer of immediately available funds to an account designated by Sellers before Closing for such purpose.
- 2.5 <u>Withholding.</u> Purchaser will be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any Seller such amounts as the Purchaser is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of applicable Law. To the extent that amounts are so withheld, such withheld amounts will be treated for all purposes of this Agreement as having been paid to the Seller in respect of which such deduction and withholding was made by the Purchaser.

#### III. PHYSICAL COUNT PROCEDURES AND VALUATION OF INVENTORY

3.1 <u>Physical Count Inventory Procedures.</u> Not more than five (5) days before the Closing, unless otherwise agreed by Sellers and Purchaser, a physical count of the Inventory at the Locations (the "Physical Inventory") shall be taken by an independent inventory company mutually agreed to by Sellers and Purchaser (the "Independent Auditor"). Unless otherwise agreed in writing by Sellers and Purchaser, the procedures for conducting the Inventory count and valuing the Inventory are set forth in this Article III.

#### 3.2 <u>Physical Inventory Procedures</u>.

- (a) <u>Observation Rights</u>. Both Sellers' and Purchaser's representatives can be present to observe the taking of any Physical Inventory. The representatives of Sellers and Purchaser shall execute an inventory completion certificate when the Physical Inventory is complete.
- (b) <u>Merchandise, Food Service and Supply Inventory.</u> A physical count of actual quantities of Merchandise Inventory, Food Service Inventory and Supplies Inventory will be taken by the Independent Auditor not more than five (5) days before or five (5) days after the Closing.
- (c) <u>Petroleum Inventory</u>. A measurement of the amount of the Petroleum Inventory at the Locations shall be made as close as practicable to the Closing. Arrangements will be made to have employees at the Locations take readings from the Veeder-Root Automatic Tank Gauge (ATG) and the VeriFone Ruby-Sapphire (Ruby) as of the Closing, as verified by the Independent Auditor. The Petroleum Inventory, water and sediment level readings will be measured by the ATG and printed out. Simultaneously, the employees shall cause the Ruby to close the shift totals for petroleum sales and print out the closing report. All tank inventories shall be taken at ambient conditions and calculated at total observed volume using the ATG at the Locations. A comparison of each Location's volumes as calculated from the Physical Inventory will be compared to the Location's book inventory and sales records. Excessive variances may be questioned by Purchaser and, if reasonably necessary, Purchaser may request that a second Physical Inventory be taken to ensure the accuracy of the reported readings.
- 3.3 <u>Valuation of Inventory</u>. For purposes of calculating the Adjusted Share Purchase Price, the book value of Inventory items which are considered Current Assets hereunder shall be adjusted as set forth in this Section 3.3.
- (a) <u>Merchandise Inventory</u>. Merchandise Inventory (other than cigarettes) will be valued at seventy percent (70%) of Company's retail price of each item, and cigarettes will be valued at Company's actual cost.
  - (b) <u>Supplies Inventory</u>. The normal operating level will be maintained and shall be valued at Company's book value at Closing.
  - (c) <u>Cash Drawer</u>. The amount of the Cash Drawer shall be included in the value of the Inventory for each Location.
- (d) <u>Food Service Inventory</u>. Food Service Inventory will be valued at Company's cost, as determined by Company's monthly foodservice workbook, a copy of which will be provided to Purchaser and the Independent Auditor before or at the Physical Inventory.
- (e) <u>Petroleum Inventory</u>. The Petroleum Inventory will be valued utilizing Company's cost of each grade of product for the last delivery to the Locations plus the current freight rate charged to transport the product to the Locations. The total value of Petroleum Inventory is to include all state sales and state and federal excise taxes ("Gasoline Taxes"), whether or not Purchaser holds an exemption certificate. The Petroleum Inventory will be

- (f) <u>Deliveries and Sales of Inventory</u>. All deliveries of Inventory to, and all sales of Inventory at, each Location after the Physical Inventory is completed but before the Closing and before the Physical Inventory is completed but after the Closing shall be taken into account for purposes of determining the Inventory at Closing.
- (g) <u>Disputes</u>. Should any disputes that cannot readily be settled between the Parties arise as to the count or pricing of any category of the Inventory, the amount then disputed shall be determined under the procedure set forth in Section 2.2(b).
- 3.4 <u>Costs.</u> Purchaser shall be solely responsible for all of the costs and expenses of the inventory audit taken at the Locations and shall indemnify and hold harmless Sellers and their Affiliates from and against all such costs and expenses. This obligation shall survive the Closing.

#### IV. <u>CLOSING, DELIVERIES AND ADJUSTMENTS</u>

- 4.1 <u>Closing</u>. The purchase and sale of the Shares (the "Closing") shall occur on a date and time as designated in writing by Purchaser to Sellers with at least five business days' notice, but in no event later than December 22, 2012 (the "Outside Date") or earlier than one (1) business day after the Closing (as defined in the Purchase and Sale Agreement). The Closing will be held at the offices of Holland & Knight LLP, in Orlando, Florida, and to the extent feasible, the Closing will be held by Federal Express or the electronic exchange of documents in PDF format or by facsimile, without the principals present. The actual date as of which the Closing occurs is referred to herein as the "Closing Date."
- 4.2 <u>Proceedings at Closing</u>. All proceedings to be taken and any documents to be executed and delivered by any of the Parties at the Closing shall be deemed to have been taken, executed and delivered simultaneously, and no proceedings shall be deemed taken nor any documents executed or delivered until all have been taken, executed and delivered.
  - 4.3 <u>Deliveries by Sellers to Purchaser</u>. At the Closing, Sellers shall deliver to Purchaser the following, duly executed:
    - (a) the stock certificates representing all the Shares endorsed in blank or accompanied by duly executed assignment documents;
    - (b) the various certificates, instruments and documents referred to in Section 9.2 below;
    - (c) resignations of the officers and directors of the Company as of the Closing Date;
- (d) a copy of a certificate of good standing for Company from the Florida Department of State dated as of a date reasonably close to the Closing Date;
- (e) a non-imputation affidavit and title affidavit, in forms reasonably required by Escrow Agent, to issue a title policy to the Company insuring its title to the Owned Real Properties, executed by Sellers; and
  - (f) the Environmental Remediation and Indemnity Agreement.
- 4.4 <u>Deliveries by Purchaser to Sellers</u>. At the Closing, Purchaser shall deliver or cause to be delivered, or direct the Escrow Agent to deliver, to Sellers the following, duly executed, as applicable:
- (a) a portion of the Deposit equal to One Million Dollars (\$1,000,000) (such portion referred to hereinafter as the "Escrow Amount") shall continue to be held in an account (the "Escrow Account") by the Escrow Agent pursuant to the terms of the Escrow Agreement;
- (b) a portion of the Deposit equal to the Environmental Escrow Amount shall continue to be held in an account by the Escrow Agent pursuant to the terms of the Environmental Escrow Agreement;
- (c) the difference between the Additional Deposit and the sum of (i) the Escrow Amount and (ii) the Environmental Escrow Amount shall be released by Escrow Agent to Sellers as further described on <u>Schedule 4.4(c)</u> hereof;
- (d) the Company shall distribute an amount equal to the Initial Deposit plus the net proceeds of the sale of the Property under the Purchase and Sale Agreement to the Sellers;
- (e) immediately available U.S. funds in the amount of the balance of the sum of the Share Purchase Price referred to in Section 2 hereof (excluding the Tax Gross-Up Amount) shall be paid by Purchaser to the Sellers as further described on <u>Schedule 4.4(c)</u> hereof;
- (f) Purchaser shall deposit an amount equal to the Tax Escrow with the Escrow Agent to be held by the Escrow Agent pursuant to the terms of the Escrow Agreement;
  - (g) the various certificates, instruments and documents referred to in Section 9.1 below;
  - (h) certified copies of the resolutions of the Board of Directors of Purchaser authorizing the transactions contemplated hereby; and
- (i) with respect to Purchaser, if applicable, a copy of a certificate of good standing or existence (as applicable) for Purchaser from the applicable Secretary of State (or other applicable governmental entity) dated as of a date reasonably close to the Closing Date.
- 4.5 <u>Property Tax Adjustments</u>. As applicable, all real estate taxes, ad valorem personal property and inventory taxes, fire district taxes, and other such charges constituting a lien or encumbrance on any of the Assets, including special or betterment assessments against any of the Assets which are payable in installments over more than one year (collectively "Property Taxes"), shall be adjusted on the basis of the fiscal year (regardless of the date of assessment) of the state, county, city or town in which any respective Assets are situated (the

"Fiscal Year"). All unpaid Property Taxes that relate to Fiscal Years ending on or before the Closing Date shall be the responsibility of Sellers. Property Taxes that relate to the Fiscal Year in which the Closing occurs shall be prorated between Sellers and Purchaser, with Sellers being responsible for the portion of such Fiscal Year up to and including the Closing Date and Purchaser being responsible for the portion of such Fiscal Year after the Closing Date. Property Taxes that relate to a Fiscal Year commencing after the Closing Date shall be the responsibility of the Purchaser. All installments of Property Taxes having a due date on or before the Closing Date shall be paid by Company on or before the Closing Date, and Purchaser shall pay all installments of Property Taxes having a due date after the Closing Date. At the Closing, the net amount of all Property Tax adjustments computed according to this Section, based upon which Party is responsible for such Property Taxes and which Party is to pay such Property Taxes, shall be added to or deducted from the Current Assets or Current Liabilities, as the case may be. If the amount of any Property Taxes is not fixed and determined as of the Closing, the foregoing Closing adjustment shall be based on the amount thereof as reasonably estimated at Closing. The foregoing adjustment will be calculated as part of the adjustment contemplated by Section 2.2.

- 4.6 Rents and Prepaid Expenses. Rents, prepaid expenses, and similar items relating to the Assets and benefiting Purchaser shall be prorated between the Parties as of the date of Closing. To the extent that any Leased Real Properties requires payment of additional rent, percentage rental, real estate taxes, insurance, common area maintenance and other lease charges, including any amounts calculated based on sales or gross or net income or cost of goods, Sellers shall be responsible for the same to the extent the same accrue related to periods before and ending as of the Closing Date for the particular Location, and Purchaser shall be responsible and agrees to pay the same for periods after the applicable Closing Date. At the Closing, an appropriate adjustment in the Current Assets or Current Liabilities, as the case may be shall be made as between Sellers and Purchaser to the extent there has been a prepayment or a shortfall in the payment of the same. The foregoing adjustment will be calculated as part of the adjustment contemplated by Section 2.2.
- 4.7 <u>Utilities</u>. Charges for water, gas, power, light and other utility service shall be the responsibility of Sellers with respect to service up to the Closing Date and shall be Purchaser's responsibility with respect to service on and after the Closing Date. The Parties shall endeavor to obtain meter readings or other evidence of the amounts due for utilities before Closing, but if such readings or evidence cannot be obtained before Closing, the Closing shall be completed without adjustment of the same, and upon obtaining such reading or evidence after Closing, Sellers shall pay Purchaser the charges incurred before the Closing based upon such reading. The foregoing adjustment will be calculated as part of the adjustment contemplated by Section 2.2.

# V. REPRESENTATIONS AND WARRANTIES OF SELLERS

Sellers hereby jointly and severally represent and warrant to Purchaser as follows:

5.1 <u>Authorization; Enforceability; Absence of Conflicts</u>. Each Seller has the legal capacity to own and hold the Shares. Each Seller has full power and authority to enter into and perform his or her obligations under this Agreement and each other agreement, document,

instrument or certificate contemplated by this Agreement or to be executed by such Seller in connection with the consummation of the transactions contemplated by this Agreement (all such other agreements, documents, instruments and certificates required to be executed by Sellers being hereinafter referred to, collectively, as the "Seller Documents"). This Agreement has been, and the Seller Documents will be at or before the Closing, duly executed and delivered by each Seller and, assuming the due authorization, execution and delivery by the other Parties hereto and thereto, this Agreement constitutes, and the Seller Documents when so executed and delivered will constitute, legal, valid and binding obligations of each Seller, enforceable against each Seller according to their respective terms, except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, or other laws of general application relating to or affecting the enforcement of creditors' rights generally, and (ii) as limited by laws relating to the availability of specific performance, injunctive relief, or other equitable remedies. None of the execution and delivery by Sellers of this Agreement and the Seller Documents, or the consummation of the transactions contemplated hereby or thereby, or compliance by Sellers with any of the provisions hereof or thereof, will (x) conflict with, violate, result in the breach or termination of, or constitute a default under, any agreement to which any Seller is a party or by which he or she or any of his or her properties or assets is bound or subject or (y) constitute a violation of any Law applicable to Sellers.

- 5.2 <u>Consents and Approvals</u>. Except as set forth on <u>Schedule 5.2</u>, no consent, waiver, approval, order, permit or authorization of, or declaration or filing with, or notification to, any person or governmental body is required on the part of Sellers or Company in connection with the execution and delivery of this Agreement or the Seller Documents or the compliance by Sellers with any of the provisions hereof or thereof.
- 5.3 <u>Organization and Good Standing</u>. Company is a corporation duly organized, validly existing and in good standing under the Laws of the State of Florida and has the corporate power and authority to own or lease its assets and to carry on its business as it is currently being conducted, except where the failure to have such requisite corporate or similar power and authority would not be reasonably likely to have a Material Adverse Effect. Company is not in default under or in violation of any provisions of its Articles of Incorporation or bylaws.
- 5.4 <u>Capitalization</u>. The entire authorized capital stock of Company consists of one hundred (100) shares of voting common stock of which one hundred (100) shares are issued and outstanding, and nine thousand nine hundred (9,900) shares of non-voting common stock of which four thousand eight hundred sixty-two (4,862) shares are issued and outstanding. All of the issued and outstanding Shares have been duly authorized, are validly issued, fully paid and nonassessable and are held of record by the Sellers, free and clear of all Liens. Except as set forth on <u>Schedule 5.4</u>, there is no outstanding Contract with any Person to purchase, redeem or otherwise acquire any outstanding Shares. The consummation of the transactions contemplated by this Agreement will convey to Purchaser good and valid title to the Shares, free and clear of all Liens. There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments that could require Company to issue, sell or otherwise cause to become outstanding any of its capital stock. There are no outstanding or authorized stock appreciation, phantom stock, profit

participation, or similar rights with respect to Company. There are no voting trusts, proxies, or other agreements or understandings with respect to the voting of the capital stock of Company.

5.5 <u>Subsidiaries</u>. Company does not own or control (directly or indirectly) any shares of capital stock of or other equity interest in any corporation, partnership, joint venture or other entity.

- 5.6 <u>Litigation</u>. Schedule 5.6 contains a summary of any action, suit, proceeding at law or in equity, or any arbitration or any administrative or other proceeding pending, or, to the Knowledge of Sellers, threatened, against or affecting the Company or any of its properties or rights which would have, individually or in the aggregate, a Material Adverse Effect on the Company or that seeks to enjoin or obtain damages in respect of the consummation of the transactions contemplated by this Agreement or that questions the validity of this Agreement, the Seller Documents or any action taken or to be taken by Sellers in connection with the consummation of the transactions contemplated hereby or thereby.
- 5.7 <u>Title to Assets</u>. Company has good and marketable title to, or a valid leasehold interest in, the Assets free and clear of all liens except the Permitted Encumbrances. NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, OF ANY NATURE IS BEING MADE BY ANY SELLER WITH RESPECT TO SUCH ASSETS, INCLUDING, BUT NOT LIMITED TO, SUCH ASSETS' MERCHANTABILITY, DESIGN OR USE FOR A PARTICULAR PURPOSE.
- 5.8 <u>Brokerage Fees</u>. Except as set forth in Section 8.5 below, Sellers or Company are not a party to any contract or undertaking to pay any broker's, finder's or financial advisor's fee in connection with the origin, negotiation, execution or performance of this Agreement.
- 5.9 <u>FIRPTA Status</u>. No Seller is a "foreign person" as defined in the Foreign Investment in Real Property Tax Act, 26 U.S.C. 1445(f)(3) ("FIRPTA"), and each Seller agrees to execute and deliver to Purchaser an affidavit to that effect, including such Sellers' federal tax identification number, before the Closing.
- 5.10 <u>Compliance With Laws; Permits</u>. Company holds in its name all applicable Permits required by Law for the operation of the Company's business, all of which Permits are listed on <u>Schedule 5.10</u>. Except as disclosed in <u>Schedule 5.10</u>, Company has not received official notice of any alleged violation of, nor citation for noncompliance with, any Law or Permit (except for minor violations or events of noncompliance which have been cured or remedied) including, without limitation, all applicable Laws relating to employment, employee benefits or the sale of tobacco, and the Americans With Disabilities Act, relating to the Business or the Assets and, to Sellers' Knowledge, except as set forth on <u>Schedule 5.10</u>, Company is in material compliance with all Laws and Permits. A list of all Permit-related violations or citations (including alcoholic beverage licenses) issued with respect to any Location in the last three (3) years or which are unresolved, regardless of when issued, is attached as <u>Schedule 5.10</u>.

#### 5.11 Financial Information.

(a) Company has furnished Purchaser with the audited balance sheet of the Company as at December 31, 2011 (the "Balance Sheet Date") and as at December 31, 2010, and

the related audited statements of operations, stockholders' equity and cash flows for the fiscal years ended December 31, 2011 and December 31, 2010, all certified by the Company's accountants, and the unaudited balance sheet of the Company as at September 30, 2012 and the related unaudited consolidated statements of operations and cash flows for the nine (9) months ended September 30, 2012. The financial statements referred to above, including the footnotes thereto (collectively, the "Financial Statements") are attached hereto as <u>Schedule 5.11</u> and, except as described therein or on <u>Schedule 5.11</u>, have been prepared in accordance with GAAP (except, in the case of the unaudited financial statements, for the absence of notes thereto and subject to year-end adjustments and as otherwise described therein). The audited balance sheets of the Company referred to in Section 5.11(a) fairly presents, in all material respects, the financial position of the Company as at December 31, 2011 and as at December 31, 2010 and the related statements of operations, stockholders' equity and cash flows of the Company for the fiscal years ended December 31, 2011 and December 31, 2010.

- (b) The Company does not have any Liabilities that are required to be set forth on an audited balance sheet prepared in accordance with GAAP, except (i) Liabilities reflected on the Financial Statements, (ii) Liabilities incurred in the ordinary course of business since the Balance Sheet Date, and (iii) Liabilities arising from performance obligations under any Contract.
- (c) Except as set forth on Schedule 5.11, during the period from the Balance Sheet Date to the date of this Agreement, (i) the business of Company has been conducted in all material respects in the ordinary course, (ii) Company has not materially increased the compensation of any officer or granted any general salary or benefits increase to their respective employees, other than in the ordinary course of business, (iii) Company has not acquired any business or Person, by merger or consolidation, purchase of substantial assets or equity interests, or by any other manner, in a single transaction or a series of related transactions, or entered into any Contract, letter of intent or similar arrangement with respect to the foregoing, (iv) Company has not sold, leased or otherwise disposed of any of its properties or assets that are material to its business and (v) there has been no material change by Company in accounting principles, practices or methods except as required by Law or GAAP.

# 5.12 Taxes.

- (a) Company has filed or caused to be filed, or shall file or cause to be filed, in accordance with applicable Law, all income, franchise and other material Tax Returns that are required by Law to be filed by, or with respect to, Company (taking into account any applicable extension of time within which to file) and all such Tax Returns were, or will be, true, correct and complete in all material respects. All Taxes owed by the Company (whether or not shown on any Tax Return) have been timely paid, other than any Taxes being contested in good faith by appropriate proceedings and identified on Schedule 5.12(a).
- (b) Except as set forth on <u>Schedule 5.12(b)</u>, Company is not currently the subject of an audit, administrative Tax proceedings, judicial proceedings or other examination of Taxes ("Tax Proceeding") by any Governmental Authority and no Tax Proceeding is pending or threatened. Company has not entered into a written agreement or waiver extending any statute of

limitations relating to the payment or collection of Taxes of Company that has not expired nor is presently contesting any Tax liability before any Governmental Authority. The Company has not received any notice of deficiency or proposed adjustment for any amount of Tax proposed, asserted or assessed by any Governmental Authority against the Company that has not been fully resolved and paid in full.

- (c) The Company (i) has withheld (or will withhold prior to Closing) from all employees, customers, independent contractors, creditors, members and any other applicable payees proper and accurate amounts for all taxable periods in compliance, in all material respects, with all Tax withholding provisions of applicable Law and (ii) has remitted, or will remit on a timely basis, such amounts to the appropriate Governmental Authority.
  - (d) The Company is not a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code.
- (e) The Company and Sellers have delivered to the Purchaser correct and complete copies of all examination reports and statements of deficiencies assessed against or agreed to by the Company that relate to any tax year for which the applicable statute of limitations has not expired.
- (f) The Company has not received any notification from a Governmental Authority in a jurisdiction where the Company does not file Tax Returns that the Company is or may be subject to taxation by that jurisdiction.
- (g) Except as disclosed on <u>Schedule 5.12(g)</u>, no power of attorney granted by the Company with respect to any Taxes is currently in force.
- (h) There are no liens for Taxes upon the Assets, other than liens for Taxes not yet due and payable or liens identified on <u>Schedule 5.12(h)</u> for Taxes being contested in good faith by appropriate proceedings and for which adequate reserves have been established clearly on the Company's financial statements in accordance with GAAP.
- (i) The Company is not a party to any Tax allocation, indemnification or sharing agreement pursuant to which it would have any obligation to make payments after Closing to any party. The Company is not and has not been a member of an affiliated, combined or unitary group for U.S. federal, state or local income tax purposes. The Company has no liability for the Taxes of any other Person as a transferee or successor, by contract, or otherwise.
- (j) The Company has not participated in, nor has any liability for the payment of any Tax resulting from a Person's participation in: a "reportable transaction" within the meaning of Section 6707A(c) of the Code and Treasury Regulations Section 1.6011-4(b).
- (k) The Company will not be required to include any material item of income in, nor will it be required to exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting for a taxable period ending on or prior to the Closing Date under Code Section 481(c) (or any corresponding or similar provision of state, local or foreign Tax Law); (ii) "closing agreement" as described in Code Section 7121 (or any corresponding or similar

provision of state, local or foreign Tax Law) executed on or prior to the Closing Date; (iii) installment sale or open transaction disposition made on or prior to the Closing Date; or (iv) prepaid amount received on or prior to the Closing Date.

- (l) No closing agreements, private letter rulings, technical advice memoranda or similar agreements or rulings relating to Taxes have been entered into or issued by an Governmental Authority with or in respect of the Company that would have a continuing effect after the Closing Date.
- (m) The Company has not constituted a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a) (1)(A) of the Code) in a distribution of shares qualifying for tax-free treatment under Section 355 of the Code (i) in the two years prior to the date of this Agreement or (ii) in a distribution that could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section 355(e) of the Code) in conjunction with this acquisition.
- (n) The Company has been a validly electing S corporation within the meaning of Code Sections 1361 and 1362, and has made valid state S corporation elections, where applicable, effective beginning January 1, 2006 and such S status will continue through the taxable year that ends as a result of the Closing.
- 5.13 <u>Condemnations</u>. Except as set forth in <u>Schedule 5.13</u>, Company has not received any written notice of any condemnation, expropriation, eminent domain or similar proceeding affecting all or any portion of any Locations, nor to Sellers' Knowledge is there any pending condemnation, expropriation, eminent domain or other similar proceeding affecting all or any portion of any Location.
- 5.14 <u>Labor Matters/Employees</u>. No Company employee is represented by any union or any collective bargaining agreement. No labor organization or group of employees of Company has made a pending demand for recognition or certification, and there are no representation or certification proceedings or petitions seeking a representation proceeding presently pending or, to the Knowledge of Sellers, threatened to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority. To the Knowledge of Sellers, Company is in compliance in all material respects with all applicable Laws relating to the employment and labor (including those relating to wages, hours, immigration and the classification of employees as exempt or not exempt from the payment of minimum wages or overtime under applicable Law, the prohibitions against discrimination and harassment, occupational safety and health, leaves of absence, and the collection and payment of withholding and social security taxes), except for violations that would not have, or be reasonably likely to have, individually or in the aggregate, a Material Adverse Effect. There are no material grievances or other material labor disputes pending or, to the Knowledge of Sellers, threatened against or involving Company. <u>Schedule 5.14</u> contains a list of the names, current pay rates and accrued vacation pay and sick leave of each employee of the Company as of the Effective Date.
- 5.15 <u>Bonds</u>. All surety or other bonds that Company maintains in connection with its operation of the Business on the Locations are listed on <u>Schedule 5.15</u> and no other surety or other bonds will be required by Purchaser at the Closing to conduct the Business. Company is

not in default under any of such surety or other bonds, and, to the Knowledge of Sellers, there has been no occurrence nor exists any condition or circumstance that, with either notice or passage of time or both, would result in or constitute a default by Company under any of such surety or other bonds.

5.16 <u>Contracts</u>. Other than those Contracts that are set forth on <u>Schedule 5.16</u> (collectively, the "Material Contracts"), Company is not a party to or bound by any Contracts of the following nature:

- (a) Contracts that contain a minimum annual purchase requirement of \$50,000 or more which have a term of more than one (1) year and that cannot be canceled on less than ninety (90) days' notice;
- (b) Contracts that materially restrain, limit or impede the Company's ability to compete with or conduct any business or line of business;
  - (c) Contracts involving Indebtedness of the Company;
- (d) Contracts (other than this Agreement and any agreement or instrument entered into pursuant to this Agreement) with (i) any Seller or any Affiliate of any Seller or (ii) any current or former officer or director of the Company;
- (e) Contracts evidencing partnerships, joint ventures or minority equity interests or that obligate Company with respect to contingent payments; and
- (f) Contracts relating to any material legal proceedings or other actions involving the Company at any time during the last two (2) years.

Each Material Contract set forth on <u>Schedule 5.16</u> is in full force and effect and is the legal, valid and binding obligation of Company. Company is not in default under any Material Contract in any material respect, and, to the Knowledge of Sellers, there has been no occurrence nor exists any condition or circumstance that, with either notice or passage of time or both, would result in or constitute a default by Company under any Material Contract

5.17 <u>Insurance. Schedule 5.17</u> hereto lists all material insurance policies or binders which currently are in effect insuring the Business, copies of which have been made available to Purchaser. No party to any such policy has repudiated any provision thereof or has provided any notice to Company or Sellers that it is denying coverage for any event covered thereby and each policy is legal, valid, binding, enforceable and in full force and effect. A schedule of all claims by Company under any of such policies in the two (2) years prior to the Effective Date of this Agreement is included in <u>Schedule 5.17</u>

#### 5.18 <u>Employee Benefits</u>.

(a) Each employee benefit plan, within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and each material fringe benefit, deferred compensation, severance, stock option, stock appreciation rights, incentive and bonus plan maintained by Company or to which the Company contributes (or has

an obligation to contribute) or is a party, in each case for the benefit of current or former Company employees (collectively, the "Employee Benefit Plans") is listed on Schedule 5.18. True and complete copies of all Employee Benefit Plans and, if applicable with respect to any Employee Benefit Plans, any trust instruments, insurance contracts, most recent determination or opinion letters, summary plan descriptions and Form 5500s for the most recent plan year have been made available to Purchaser or its counsel. Each Employee Benefit Plan that is intended to be "qualified" within the meaning of Section 401(a) of the Code, has received, or has timely requested, a favorable determination letter or opinion letter from the Internal Revenue Service, and, to the Knowledge of Sellers, no event has occurred and no condition exists that would reasonably be expected to result in the revocation of, or failure to issue, any such determination letter or opinion letter. No Employee Benefit Plan is covered by Title IV of ERISA or subject to Section 412 of the Code or Section 302 of ERISA.

(b) Except as set forth on Schedule 5.18, (i) each Employee Benefit Plan is in material compliance with applicable Law and has been administered and operated in all material respects in accordance with its terms; (ii) neither Company nor, to the Knowledge of Sellers, any other "disqualified person" or "party in interest" (as defined in Section 4975(e)(2) of the Code and Section 3(14) of ERISA, respectively) has engaged in any transactions in connection with any Employee Benefit Plan that would reasonably be expected to result in the imposition on Company of a material penalty pursuant to Section 502 of ERISA, material damages pursuant to Section 409 of ERISA or a material Tax pursuant to Section 4975 of the Code; (iii) no Employee Benefit Plan provides for post-employment or retiree welfare benefits, except as required by applicable Laws; (iv) no claim, action or litigation has been made, commenced or, to the Knowledge of Sellers, threatened, with respect to any Employee Benefit Plan (other than routine claims for benefits payable in the ordinary course, and appeals of denied such claims) that would reasonably be expected to result in a material liability of Company; (v) neither the execution of this Agreement nor the consummation of the transactions contemplated hereby will result in any excess parachute payment (within the meaning of Section 280G of the Code) under any Employee Benefit Plan or cause the acceleration of vesting in, or the time of payment of, any benefits under any Employee Benefit Plan or increase any amount payable under any Employee Benefit Plan; (vi) Company has no contractual obligation to make any gross-up payments as a result of the excise taxes imposed by Section 4999 of the Code or the additional income taxes imposed by Section 409A of the Code with respect to any Employee Benefit Plan; and (vii) Company maintains no Employee Benefit Plan outside of the United States primarily for the benefit of Company employees working outside the United States.

#### 5.19 <u>Real Properties</u>.

(a) Set forth on Schedule 5.19 is an accurate and complete list and summary description of the Owned Real Properties and the Leased Real Properties (collectively, the "Facilities") and, except as set forth on Schedule 5.19, none of the described Leases require any consent to the transactions contemplated by this Agreement. Except as set forth on Schedule 5.19, Company has not received written notice of any physical defects or conditions or violations of applicable law (representations and warranties with respect to environmental matters being set forth in Section 5.20 hereof, and being specifically excluded from this Section 5.19) with regard to any of the Facilities. Company has made available for review by Purchaser complete copies of all surveys, reports or deficiency notices concerning the Facilities which are

in the possession of Company. Company has previously delivered to Purchaser complete copies of all Leases listed and described on <u>Schedule 5.19</u>. Company has possession of each of the Facilities. Except as set forth on <u>Schedule 5.19</u>, to the Knowledge of Sellers, Company has not received written notice of any default under any of the described Leases and, to the Knowledge of Sellers, no event has occurred which, with the lapse of time or action by a third party, will result in a default under any of the Leases.

- (b) All rents or other payment obligations which have become due in respect of each of the leased Facilities have been or will be paid in the ordinary course of business and Company has complied in all material respects with its obligations under the said Leases to which it is a party. Company has not received any written notice of default under any Law, Contract or Permit, relating to the use and operation of the Facilities, which could reasonably be expected to have a material adverse effect on the operation of such Facility.
- 5.20 <u>Environmental Matters</u>. Except as disclosed on <u>Schedule 5.20</u> which shall include any Property to be sold by the Company under the Purchase and Sale Agreement, to the Knowledge of Sellers:
- (a) no written or oral notice, notification, demand, request for information, citation, summons, complaint or order has been received by Company, no complaint has been filed, no penalty has been assessed and no investigation or review is pending or, to the Knowledge of Sellers, threatened by any governmental entity or other person or entity with respect to any past or present (i) alleged violation by Company of any Environmental Laws or liability thereunder; (ii) alleged failure by Company to have any Permit required under any Environmental Laws in connection with the conduct of its business; (iii) Release of Hazardous Substances (as hereinafter defined); or (iv) damages to natural resources;
- (b) no surface impoundments, pits or lagoons in which Hazardous Substances are being or have been treated, stored, or disposed of have been located on any property owned or leased by Company since the date Company was incorporated or, to the Knowledge of Sellers, at any time before, no Hazardous Substances have been found in any potable water (excluding potable water supplied by governmental authorities) used to supply any such property since the date the Company was incorporated or, to the Knowledge of Sellers, at any time before, and no polychlorinated biphenyls, radioactive material, urea formaldehyde, lead, asbestos, asbestos-containing material or other Hazardous Substances in an amount or concentration so as to violate Environmental Laws is or was deposited or released at, under or on any property owned or leased by Company since the date Company was incorporated or, to the knowledge of Sellers, at any time before;
  - (c) there are no Environmental Liabilities (as hereinafter defined in Article XIII);
- (d) there has been no environmental investigation, study, or audit, conducted of which Company has possession or control in relation to the current business of Company or any property or facility now owned or leased by Company which has not been delivered to Purchaser;
- (e) with the exception of gasoline products, Company has not transported or arranged for the transportation of any Hazardous Substance to any location which is listed or proposed for listing on the National Priorities List under CERCLA, or on any similar state list or which is the subject of Federal, state or local actions regarding the release of Hazardous Substances or, to the Knowledge of Sellers, other investigations which may lead to claims for clean-up costs, remedial work, damages to natural resources or for personal injury claims, including but not limited to, claims under CERCLA or analogous state environmental clean-up laws; and
- (f) Company is in material compliance with all Environmental Laws relative to the construction, maintenance, compliance and use of underground storage tanks and appurtenances thereto at all Facilities.
- Intellectual Property. Except as disclosed on Schedule 5.21, Company owns or has the right to use, free and clear of all Liens, all Intellectual Property necessary to conduct the business of Company substantially as presently conducted except where the failure to so own or have such rights, or the presence of such Liens, does not have, individually or in the aggregate, a Material Adverse Effect. To the Knowledge of Sellers, the operation of the business of Company has not and does not materially infringe, misappropriate, dilute or otherwise violate the Intellectual Property of any third Person. No claim is pending or, to the Knowledge of Sellers, has been threatened against Company, and Company has not received any written threat, charge, claim, demand, notice or complaint, in each case within the last two (2) years, challenging the use or ownership by Company of any Intellectual Property used in the business of Company or alleging that the operation of the business of Company infringes, misappropriates, dilutes or otherwise violates the Intellectual Property of any third Person (including any claim that Company must license any Intellectual Property of any third Person). Notwithstanding the foregoing, Purchaser acknowledges that Company has not registered the service mark and trade name "Express Lane" with the USPTO or any state registering authority, and Sellers make no representations or warranties to the effect that Company has the exclusive right to use the mark and name "Express Lane" throughout the United States.
- 5.22 <u>Bank Accounts.</u> <u>Schedule 5.22</u> lists the names, account numbers and locations of all banks and other financial institutions at which Company has an account or safe deposit box and the names of each Person authorized to draft on or have access to any such account or safe deposit box.

#### VI. REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser represents and warrants to Sellers as follows:

6.1 <u>Organization and Good Standing</u>. Purchaser is a corporation duly organized, validly existing, and in good standing under the Laws of its state of incorporation and has the organizational power and authority to own its property and to carry on its business as now conducted and to enter into and to carry out the terms and conditions of this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by Purchaser in connection with the consummation of the transactions contemplated by this Agreement (all such other agreements, documents, instruments and certificates required to

be executed by Purchaser being hereinafter referred to, collectively, as the "Purchaser Documents") and to perform fully its obligations hereunder and thereunder.

6.2 <u>Due Authorization; Enforceability; Absence of Conflicts</u>. The execution, delivery and performance by Purchaser of this Agreement and each Purchaser Document has been duly authorized and approved by all necessary action on the part of Purchaser. This Agreement has been, and the Purchaser Documents will be at or before the Closing, duly executed and delivered by Purchaser and (assuming the due authorization, execution and delivery by the other Parties hereto and thereto) this Agreement constitutes, and the Purchaser Documents when so executed and delivered will constitute, legal, valid and binding obligations of Purchaser, enforceable against Purchaser according to their respective terms, except (i) as limited by applicable bankruptcy,

insolvency, reorganization, moratorium, fraudulent conveyance, or other laws of general application relating to or affecting the enforcement of creditors' rights generally, and (ii) as limited by laws relating to the availability of specific performance, injunctive relief, or other equitable remedies. None of the execution and delivery by Purchaser of this Agreement and the Purchaser Documents, or the consummation of the transactions contemplated hereby or thereby, or compliance by Purchaser with any of the provisions hereof or thereof, will (x) conflict with, or result in the breach of, any provision of the organizational or governing documents of Purchaser, (y) conflict with, violate, result in the breach or termination of, or constitute a default under, any agreement to which Purchaser is a party or by which it or any of its properties or assets is bound or subject or (z) constitute a violation of any Law applicable to Purchaser.

- 6.3 <u>Consents and Approvals</u>. Except as set forth on <u>Schedule 6.3</u>, no consent, waiver, approval, order, permit or authorization of, or declaration or filing with, or notification to, any person or governmental body is required on the part of Purchaser in connection with the execution and delivery of this Agreement or the Purchaser Documents or the compliance by Purchaser with any of the provisions hereof or thereof.
- 6.4 <u>Litigation</u>. There is no legal proceeding pending or, to the knowledge of the Purchaser, threatened against the Purchaser that seeks to enjoin or obtain damages in respect of the consummation of the transactions contemplated by this Agreement or that questions the validity of this Agreement, the Purchaser Documents or any action taken or to be taken by the Purchaser in connection with the consummation of the transactions contemplated hereby or thereby.
- 6.5 <u>Brokerage Fees</u>. Purchaser represents and warrants that it has not acted in a manner that could cause Sellers to incur liability to any person for brokerage commissions, finder's fees or other remuneration in connection with the sale of the Shares or the transactions contemplated by this Agreement, other than the payments by Sellers contemplated by Section 8.5 hereof.
- 6.6 <u>Financial Capacity</u>. Purchaser has the net worth, financial standing, access to required liquidity and the necessary borrowing capacity to complete the transactions contemplated by this Agreement.
- 6.7 <u>Eligibility for Alcoholic Beverage License</u>. Purchaser (or its equity owners, as applicable) has not been convicted of a felony, is not controlled, majority owned, or affiliated with, directly or indirectly, any other Person that is engaged in manufacturing distributing, or exporting alcoholic beverages under a Florida license or registration or any other state of the United States, and it knows of no reason why its application for a retail alcoholic beverage license from the Florida Division of Alcoholic Beverages and Tobacco for the Locations would not be approved.
- Investment. Purchaser is acquiring the Shares for investment purposes only and not with a view to a distribution or resale thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act"). Purchaser has not directly or indirectly offered the Shares to anyone or solicited any offer to buy the Shares from anyone, in each case in a manner that would require registration under the Securities Act. Purchaser shall not sell, convey, transfer or offer for sale any of the Shares except in compliance with the registration and other requirements of the Securities Act, applicable state securities Laws and other applicable Laws or pursuant to applicable exemption therefrom.

# VII. <u>DUE DILIGENCE AND DISCLAIMER OF WARRANTIES</u>

- Access; Due Diligence. The Parties acknowledge that the Company has (i) given Purchaser, its lenders, and its accounting, legal, business, environmental, engineering, intellectual property and other authorized representatives and advisors full access, during normal business hours, to the Facilities, and (ii) furnished Purchaser and its authorized representatives and advisors with all documents and information relating to Company as reasonably requested by Purchaser, its lenders, and its authorized representatives and advisors to review all books, records and Contracts of Company as requested by Purchaser and its authorized representatives and advisors, and made copies thereof, (iv) made available the employees and advisors of Company, including those responsible for the management of the Business, and caused Company' employees and advisors to furnish Purchaser and its authorized representatives and advisors with data and other information with respect to Company as requested by Purchaser and its authorized representatives and advisors the affairs of the Business, and (v) fully cooperated with Purchaser and its authorized representatives and advisors in its investigation and examination of Company and its Business and affairs. No investigation, or receipt of information provided by or on behalf of Company or review thereof by Purchaser, its lenders, or its representatives or advisors has diminished or obviated, or relieved Sellers from, or affected the ability or right of Purchaser to rely on, any of the representations, warranties, covenants and agreements of Sellers contained in this Agreement and the Seller Documents. Purchaser acknowledges that its obligations to consummate the transactions contemplated hereby are not subject to any further due diligence investigation by Purchaser.
- 7.2 <u>Confidential Information</u>. Purchaser acknowledges that it has received information about the Company (the "Confidential Information"). The Purchaser agrees to continue to be bound by all agreements under which it received such Confidential Information, including that certain confidentiality agreement entered into before the date hereof between Purchaser and Matrix for the benefit of Sellers and Company (the "Confidentiality Agreement").

If this Agreement is terminated before the Closing, Purchaser promptly shall return to Sellers or destroy all Confidential Information and shall not retain copies thereof, including all information, testing, studies, surveys, reports and evaluations (environmental and financial, for example), and at Sellers' request, Purchaser shall provide Sellers with complete copies of all third party engineering and environmental reports obtained by Purchaser in connection with the Owned Real Properties and Leased Real Properties.

- 7.3 Environmental Site Assessment. At its sole expense, Purchaser shall have the right to perform an Environmental Site Assessment ("ESA") at the Locations before the Closing, provided the ESA conforms to Schedule 7.3 and applicable ASTM standards, does not damage or interfere with the operation of the Locations and Purchaser provides Sellers with a copy of the ESA report promptly after the report is issued. Purchaser shall indemnify and hold Sellers harmless for the acts of Purchaser and/or Purchaser's agents, employees or contractors in performing the ESA. Purchaser shall return each Location to its previous condition upon completion of its activities in connection with the ESA. Purchaser's conduct of an ESA shall not be a condition to the Closing, and failure to perform or complete the ESA shall not extend the date for the Closing.
- 7.4 <u>Notices to Governmental Agencies</u>. If Purchaser's due diligence reveals any condition at the Locations that requires disclosure to any governmental agency or authority, Purchaser shall immediately (a) notify Sellers thereof, and (b) provide all data and analysis concerning the condition acquired during the due diligence process including without limitation the ESA. Purchaser acknowledges that applicable rules (including particularly section

62-770.250, FAC) may require notification to a governmental agency within twenty-four (24) hours of occurrence, and Purchaser agrees to take all reasonable steps necessary to support the ability of the Company or Sellers, as the case may be, to meet this obligation. Except as provided by Law, in such event, at all times prior to the Closing, Sellers, and not Purchaser's agents, or anyone acting on Purchaser's behalf, shall make such legal determinations regarding disclosures as Sellers deem appropriate.

Purchaser Acknowledgments. Purchaser specifically acknowledges that, except for Sellers' representations in Article V, and subject to Sellers' obligations as contained in this Agreement, Purchaser is not relying on any representations or warranties of any kind whatsoever, express or implied, from Sellers, Matrix, or any director, officer, employee, representative, broker or other agent of either of them, as to any matters concerning the Locations, the Assets and/or the Business conducted therewith, including: (a) the condition or safety of the Owned Real Properties and Leased Real Properties or any improvements thereon, including plumbing, sewer, heating and electrical systems, roofing, air conditioning, if any, foundations, soils and geology, lot size, or suitability of the Owned Real Properties and Leased Real Properties or its improvements for a particular purpose; (b) whether the appliances, if any, plumbing or utilities are in working order; (c) the habitability or suitability for occupancy of any structure and the quality of its construction; (d) the fitness or condition of any personal property; (e) whether the fixtures or improvements, including the Fuel Equipment, are structurally sound, in good condition, or in compliance with applicable Laws; (f) the profits or losses relating to operations at the Locations; (g) the legal or tax consequences of this Agreement or the transactions contemplated hereby; (h) the environmental condition of the Owned Real Properties and Leased Real Properties, including but not limited to the possible presence of petroleum

products and/or Hazardous Substances in, under or near the Owned Real Properties and Leased Real Properties; and (i) the completeness or accuracy of any information provided to Purchaser by Sellers or their agents. Purchaser understands the legal significance of the foregoing provisions and acknowledges that they are a material inducement to Sellers' willingness to enter into this Agreement.

THE OPPORTUNITY TO EXAMINE ALL ASPECTS OF THE COMPANY AND TO REVIEW ALL FILES CONCERNING THE LOCATIONS MAINTAINED BY COMPANY AND ITS ENVIRONMENTAL CONSULTANTS AND ALL STATE AGENCIES HAVING JURISDICTION OVER THE LOCATIONS BEFORE PURCHASER'S EXECUTION AND DELIVERY OF THIS AGREEMENT. ACCORDINGLY, PURCHASER AGREES THAT THE SHARES SHALL BE SOLD AND THAT PURCHASER SHALL ACCEPT THAT THE COMPANY OWNS THE ASSETS AT CLOSING (EXCEPT ANY PROPERTY SOLD UNDER THE PURCHASE AND SALE AGREEMENT) STRICTLY ON AN "AS IS, WHERE IS, WITH ALL FAULTS" BASIS, WITH NO RIGHT OF SET-OFF OR REDUCTION IN THE SHARE PURCHASE PRICE EXCEPT FOR ADJUSTMENTS EXPRESSLY CONTEMPLATED BY THIS AGREEMENT AND THAT, EXCEPT FOR SELLERS' LIMITED REPRESENTATIONS SET FORTH IN ARTICLE V HEREOF, THE SALE OF SHARES AND OTHER TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT SHALL BE WITHOUT REPRESENTATION OR WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF INCOME POTENTIAL, OPERATING EXPENSES, USES, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND SELLERS DO HEREBY DISCLAIM AND RENOUNCE ANY SUCH REPRESENTATION OR WARRANTY.

#### VIII. ADDITIONAL AGREEMENTS AND COVENANTS

- B.1 <u>Distributions/Dividends</u>. Nothing herein shall prohibit Company from making distributions to Sellers prior to the Closing Date of the assets set forth on <u>Schedule 8.1</u>. In particular, Purchaser acknowledges that Company is required to distribute to Sellers all proceeds which Company receives with respect to (a) the Purchase and Sale Agreement and (b) claims Company has made arising out of the Deepwater Horizon Oil Spill against BP Exploration and Production, Inc., BP American Production Company or other third parties (the "Oil Spill Claims"). In the event that, on or prior to the day before Closing, Company has not received all sums to which it is entitled related to the Oil Spill Claims, Company shall assign on the day before Closing its right to receive any additional sums related to the Oil Spill Claims to Sellers or their assignees. Subject to Section 10.2(e), Purchaser hereby agrees to cause Company to distribute to Sellers or their assignees any additional sums related to the Oil Spill claims Company receives subsequent to Closing.
- 8.2 <u>Announcements</u>. Before Closing, neither Sellers nor Purchaser shall make any public announcements concerning the execution and delivery of this Agreement or the transactions contemplated hereby without first obtaining the prior written consent of the other, except as may be required by applicable Law or agreement with any securities exchange, and

except that Sellers may disclose this Agreement to any lender holding a lien on any property owned by the Company or otherwise subject to this Agreement and to any landlord or lender of the Company in order to comply with Sellers' obligations hereunder, and Sellers or Purchaser may disclose the identity of the other Party and terms of this Agreement to their employees, vendors, outside counsel or other outside professionals on a need to know basis in connection with consummating this transaction. If a disclosure is required by a fuel supplier agreement or applicable Law or agreement with any securities exchange, the disclosing Party shall make reasonable efforts to afford the other Party an opportunity to review and comment on the proposed disclosure before the making of such disclosure.

- 8.3 <u>Insurance and Casualty.</u> Sellers covenant and agree to cause the Company to keep the Assets (except the Property sold under the Purchase and Sale Agreement) and the Business insured, at Company's expense, in such amounts as presently insured through the Closing. On any material damage to or destruction of the Owned Real Properties or Leased Real Properties or any portion thereof before Closing, the Closing shall, at Sellers' sole discretion, nevertheless proceed; *provided*, *however*, that unless before Closing the same shall have been restored by Company, Sellers or Company's landlord, as applicable, to its condition as of the date hereof, at the Closing, Company and Purchaser shall be entitled to the benefit of any insurance proceeds due Company as a result of such damage or destruction (without recourse to Sellers) and Purchaser shall assume responsibility for such repair and receive a credit against the Share Purchase Price for any applicable policy deductible or uninsured damage up to a maximum credit equal to the difference between the amount of the Share Purchase Price and the amount of the insurance proceeds payable to Company.
- 8.4 <u>Condemnation</u>. If before Closing, any part of the Owned Real Properties or Leased Real Properties is taken, or noticed for taking, by eminent domain, Sellers shall promptly give Purchaser written notice thereof and the Closing shall at Sellers' sole discretion nevertheless proceed; provided, that such provision shall not apply with respect to any such taking of or notice for the Company's principal headquarters located at 600 Ohio Ave., Lynn Haven, FL 32444 or more than two (2) individual Locations other than those Locations disclosed on <u>Schedule 5.13</u>.

- 8.5 Matrix; Payment of Brokers. Purchaser acknowledges that Sellers and/or Company have retained Matrix Private Equities, Inc. ("Matrix") to act as the financial advisor in connection with the sale of the Shares and the transactions contemplated by this Agreement. Sellers shall be responsible for any fees due to Matrix. Each Party shall be responsible for any brokerage commission or other fees legally determined to be due to any broker or advisor claiming by, through or under such Party and, as between Sellers and Purchaser, shall also be responsible for any loss, liability, damage, cost or expense (including, without limitation, reasonable attorneys' fees) paid or incurred by the other Party by reason of any claim to any such broker's, finder's or other fee in connection with the transactions contemplated by this Agreement by any such broker or advisor.
  - 8.6 Mutual Cooperation. From and after the date hereof and until the date of Closing:
- (a) Purchaser hereby covenants and agrees with Sellers that Purchaser shall use its reasonable efforts to cause the consummation of the transactions contemplated hereby

according to the terms and conditions hereof, and Sellers hereby covenant and agree with Purchaser that Sellers shall use their reasonable efforts to cause the consummation of the transactions contemplated hereby according to the terms and conditions hereof.

- (b) Sellers shall use reasonable efforts to obtain the written consent of any other party to a Contract or Lease if such party's consent may be legally required as a result of the transfer of the Shares. Purchaser agrees to cooperate fully with Sellers to secure such consents, including supplying information about Purchaser as may be requested by third parties, if applicable. Purchaser shall bear the costs of all transfer fees.
- (c) Sellers and Purchaser shall each prepare any and all documentation and supply any and all information required by any governmental authority or agency thereof to be filed by Purchaser or Sellers, as the case may be, before conveying the Shares as contemplated hereby, and shall timely make the necessary filings or applications relating thereto. Purchaser, on the one hand, and Sellers, on the other hand, each agree to cooperate with the other in the completion, execution and submission of any such filings or applications.

#### 8.7 <u>Environmental Responsibilities</u>.

If an ESA conducted by Purchaser (or any ESA conducted by Sellers) prior to the Closing reveals releases of petroleum at the Locations which are required to be reported to the FDEP, and which have not been previously reported to FDEP, then Purchaser shall advise Sellers, as applicable, and Sellers and Company shall promptly, but in all cases prior to Closing, report the same to FDEP. Company shall thereafter undertake Remedial Measures (as described below) for such releases to the extent required by Law at the Locations consistent with a commercial use standard. The Parties acknowledge that the Company is currently undertaking Remedial Measures as set forth on Schedule 5.20. Subject to the last sentence of this Section 8.7(a), Sellers agree to be financially responsible for the cost of any Remedial Measures to the extent such Remedial Measures are not covered by insurance or do not qualify for reimbursement from trust funds from the State of Florida or other applicable governmental agency, if such Remedial Measures are specifically listed on Schedule 5.20 or are related to releases identified in the ESA prior to Closing. Purchaser must request payment for actual costs of the Remedial Measures incurred by Purchaser or Company by giving written notice under the Environmental Escrow Agreement or otherwise to Seller Representative as follows: (i) on or before December 31, 2014, for those Locations listed as "Inactive" on Schedule 5.20, or (ii) on or before December 31, 2015, for those Locations listed as "Active" on Schedule 5.20 or identified in the ESA prior to Closing. Otherwise, subject to Section 8.7(g), Sellers' financial responsibility under this Section 8.7 shall terminate. (For purposes of clarity, Sellers shall have no obligation with respect to Remedial Measures at the Locations for any releases of petroleum that are first discovered after the Closing.) "Remedial Measures" shall consist of the lowest cost activities reasonably required by applicable Law and by FDEP or other applicable governmental agency to investigate, assess, remediate and/or monitor such releases to achieve a commercial use standard consistent with all applicable Law and regulatory requirements, including but not limited to the use of monitored natural attenuation and the implementation of institutional and engineering controls. "Remedial Measures" do not include activities that exceed what is required by Law, FDEP or other applicable governmental agencies for the existing uses of the Locations. Sellers and Purchaser shall cooperate with each other in the performance of the

Remedial Measures, including execution of documents that may be necessary to obtain reimbursement. Sellers shall have the right to review and approve any financial expenditures related to the Remedial Measures for which Sellers have financial responsibility hereunder.

- (b) Company's obligation to conduct Remedial Measures for the Locations shall be deemed complete for purposes of this Agreement upon receipt by Company of a written letter, order or statement from FDEP or other appropriate governmental authority stating that, subject to the agency's usual and customary reservation of rights, it will not require Company to perform further Remedial Measures with respect to the release ("Closure Letter"), other than routine monitoring. If requested to do so by FDEP or other appropriate governmental authority in connection with the performance of the Remedial Measures or the issuance of the Closure Letter, the Parties acknowledge that Company may be asked to place reasonable activity and use limitations ("AULs") on the Locations, including, if requested, prohibiting the use of groundwater under the Locations for potable purposes and/or restricting the use of the Locations to non-residential purposes. Company shall promptly provide a copy of the Closure Letter to Purchaser and Sellers upon receipt.
- (c) If the Remedial Measures required under Section 8.7(a) have not been completed at the time of Closing, Sellers agree to deposit One Hundred Thirty-one Thousand Five Hundred U.S. Dollars (\$131,500) at Closing with the Escrow Agent pursuant to the terms of an environmental escrow agreement substantially in the form attached hereto as <a href="Exhibit B">Exhibit B</a> (the "Environmental Escrow Agreement") to be applied against the cost of such Remedial Measures.
- (d) Except to the extent set forth in Section 5.20 and this Section 8.7, Sellers shall have no other obligation to Purchaser, and Purchaser hereby releases and discharges Sellers, for any liabilities, costs, expenses, losses, damages, or claims, whether arising in law or equity, known or unknown, or arising before or after the Closing, in connection with or relating to the environment at, on, under or near the Locations, including but not limited to any such matters relating to the Fuel Equipment or releases of petroleum products or Hazardous Substances.
- (e) For purposes of clarity, after Closing, if Purchaser or Company excavates or removes any soil on or from the Locations other than in connection with taking Remedial Measures for which Sellers are financially responsible, Purchaser shall be responsible for all costs and expenses related to such excavation or removal (including, without limitation, all costs of disposal of such soil). Furthermore, Purchaser or Company, as the case may be, shall be the owner and generator of such soil. Sellers shall not be responsible for increased remediation costs or expenses as a result of such excavation or removal.

- (f) Notwithstanding anything in this Section 8.7 to the contrary, the Parties agree that the provisions of this Section 8.7 apply to all Properties that may be sold by the Company pursuant to the Purchase and Sale Agreement.
- (g) Notwithstanding anything in this Section 8.7 to the contrary, Sellers agree to be solely responsible for the Remedial Measures at a location formerly leased by the Company, located at 8022 Highway 90 East, Sneads, Florida and referred to by the Company as "Site #60" pursuant to the terms of the Environmental Remediation and Indemnity Agreement.
- 8.8 <u>Covenant Not to Compete</u>. For a period of four (4) years after the Closing Date, none of the Sellers will engage directly or indirectly in the retail petroleum or convenience store business within the State of Florida by acting as an officer, director, partner, consultant or employee of, or through any equity ownership in, any Person so engaged; provided, however, that no owner of less than one percent (1%) of the outstanding stock of any publicly traded corporation shall be deemed to engage solely by reason thereof in any of its businesses. If the final judgment of a court of competent jurisdiction declares that any term or provision of this Section 8.8 is invalid or unenforceable, the Parties agree that the court making the determination of invalidity or unenforceability shall have the power to reduce the scope, duration or area of the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified after the expiration of the time within which the judgment may be appealed. The Sellers acknowledge and agree that the restrictions set forth in this Section 8.8 are reasonable in time and scope, that their agreement in this Section 8.8 is a material inducement to Purchaser's execution of this Agreement, that conduct in violation of this Section 8.8 would be a substantial detriment to Purchaser and Company, and that Purchaser and Company shall be entitled to seek equitable relief in any court of competent jurisdiction without the requirement of the posting of any bond or surety or the proof of any actual damages.
- 8.9 Confidentiality. Each Seller covenants and agrees that, for a period of four (4) years following the Closing Date, she or he will hold all information concerning Company (other than information which (a) becomes generally available to the public, (b) was available to such Seller on a non-confidential basis prior to its disclosure, or (c) becomes available to such Seller after the Closing Date on a non-confidential basis from a source other than Purchaser or Company that is not prohibited from disclosing such information to such Persons by a contractual, legal or fiduciary obligation) on a confidential basis. Each Seller also covenants and agrees to not use themselves or voluntarily disclose (other than pursuant to legal process after an opportunity to restrict or otherwise limit disclosure) to others any such information, and to, upon request by Purchaser, promptly turn over all such confidential information and any copies thereof it may have made and to destroy any summaries, compilations or similar documents it may have made or derived from such material and have its representatives promptly turn over such documents and copies and destroy such summaries, compilations or similar documents. Each Seller further covenants and agrees that she, he or it will keep confidential and not use trade secrets of Purchaser and Company. The Sellers acknowledge and agree that the restrictions set forth in this Section 8.9 are reasonable in time and scope, that their agreement in this Section 8.9 is a material inducement to Purchaser's execution of this Agreement, that conduct in violation of this Section 8.9 would be a substantial detriment to Purchaser and Company, and that Purchaser and Company shall be entitled to seek equitable relief in any court of competent jurisdiction without the requirement of the posting of any bond or surety or the proof of any actual damages.

# 8.10 Non-Solicitation.

(a) Sellers shall not and shall cause Company and their respective representatives not to, directly or indirectly, from and after the Effective Date until the earlier of the Closing or the termination of this Agreement, (i) solicit, initiate, encourage (including by way of providing information) or induce the submission or announcement of any inquiries,

proposals or offers or any other efforts or attempts that constitute, or could reasonably be expected to lead to, a Takeover Proposal, (ii) enter into, continue, participate or engage in any discussions or negotiations with, or furnish any information to, any Person with respect to a Takeover Proposal or (iii) otherwise cooperate with or assist or participate in, or facilitate or take any action that could reasonably be expected to facilitate, any such inquiries, proposals, offers, discussions or negotiations. For purposes of this Agreement, (x) the term "Acquisition Transaction" means any transaction or series of transactions, directly or indirectly, (1) involving any exchange, lease, sale, disposition, acquisition, transfer, purchase, or exclusive license of all or substantially all of Company's rights with respect to assets or businesses of the Company, (2) that, if consummated, would result in any Person or group (other than Sellers) beneficially owning, directly or indirectly, or having the right to acquire, any outstanding shares of the capital stock of Company, (3) involving any merger, consolidation, business combination, recapitalization, liquidation, dissolution, binding share exchange or similar transaction involving the Company, (4) the issuance or sale by Company (including by way of merger, consolidation, business combination, share exchange, joint venture or similar transaction) of equity interests of Company, or (5) any combination of the foregoing, other than, in each case, the transactions contemplated by this Agreement; and (y) the term "Takeover Proposal" means any inquiry, proposal or offer (whether or not in writing) from any Person or group relating to any Acquisition Transaction.

- (b) Sellers shall immediately cease and cause to be terminated any solicitation, discussion or negotiation with any Person conducted heretofore by Sellers, the Company or any of their respective representatives with respect to any actual or potential Acquisition Transaction.
- (c) Sellers acknowledge and agree that any violation of the restrictions set forth in this Section 8.10 by any Seller or Company or their respective representatives shall be deemed to be a material breach by the Company of this Section 8.10.
- 8.11 <u>Conduct of Business Prior to Closing</u>. Except for actions contemplated by this Article VIII, actions taken with the prior written consent of Purchaser or actions related to the sale of the Property contemplated by the Purchase and Sale Agreement,, from the Effective Date until the Closing Date, Sellers shall cause Company to conduct its business in the ordinary course consistent in all material respects with past practice, and shall cause Company to:
- (a) maintain its business intact, market, promote, and sell its products and services consistently in all material respects with past practice, and preserve the goodwill of its business and present relationships with and suppliers and others with whom it has business relations;
- (b) maintain its Owned Real Properties and Leased Real Properties, buildings, structures and other improvements and machinery and equipment constituting any of its assets in good operating condition and repair in all material respects;
- (c) meet its contractual obligations in all material respects and perform and pay its obligations as they mature in the ordinary course of

- (d) comply in all material respects with all judgments and Laws promulgated by any governmental body and all Permits applicable to the conduct of its business or the ownership or operation of its assets, and maintain in all material respects, and prosecute applications for, such Permits and pay all Taxes, assessments and other charges applicable thereto;
- (e) promptly advise Purchaser in writing of any material adverse change in its assets or the conduct, business, operations, properties, condition (financial or otherwise) or prospects of its business;
- (f) not take any action, or omit to take any action, that would result in any of Company's representations and warranties made herein being inaccurate in any material respect at the time of such action or omission as if made at and as of such time;
- (g) not (i) settle or compromise any Tax liability, (ii) make or change any material Tax election, (iii) make or change any Tax method of accounting other than as disclosed on Schedule 5.11, (iv) prepare any Tax Return in a manner that is inconsistent with the past custom and practice with respect to the treatment of items on such Tax Returns, (v) file any amendment to a Tax Return that would increase the Tax liability of the Company after the Closing, (vi) consent to the extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes with any Governmental Authority, or (vii) surrender any right to claim a refund of Taxes; and
- (h) give notice to Purchaser promptly upon becoming aware of any inaccuracy in any material respect of any of the representations or warranties made herein by Sellers or in the Schedules attached hereto or of any event or state of facts that would result in any such representation or warranty being inaccurate in any material respect at the time of such event or state of facts as if made at and as of such time (any such notice to describe such inaccuracy, event or state of facts in reasonable detail); provided, however, (i) for purposes of determining whether the closing conditions described in Section 9.2 have been satisfied, no such supplemental information shall be deemed to avoid or cure any misrepresentation or breach of warranty or constitute an amendment of any representation, warranty or statement in this Agreement, and (ii) unless expressly consented to in writing by Purchaser, no such supplemental information shall be deemed to avoid or cure any misrepresentation or breach of warranty for purposes of the indemnity provisions of Article X.
- Section 8.12 <u>Contractual Obligations</u>. Purchaser and Sellers hereby agree that the responsibility for any fees or expenses associated with any consents sought under <u>Schedule 5.2</u>, including without limitation assignment fees, transfer fees and any required costs of store refurbishment shall be borne by the party that is responsible for paying such fees in the applicable third party agreement. Notwithstanding the foregoing, Purchaser agrees that under no circumstances will Sellers be responsible for any "debranding", "contingent repayment" or similar obligations under or related to the Exxon or Chevron agreements set forth on <u>Schedule 5.2</u>.
- Section 8.13 <u>Third Party Consents</u>. In the event that Purchaser closes or waives the condition precedent to Closing provided in Section 9.2(g), the Sellers will not be responsible for

obtaining, and shall have no liability for failure to obtain, the third party consents and any other approvals listed on Schedule 5.2.

- Section 8.14 <u>Operation of Business</u>. Purchaser, on behalf of its Affiliate LGP, hereby agrees that the Company shall have the right to continue to operate the Business in the ordinary course on the real estate sold under the Purchase and Sale Agreement, including exercising its rights under any Contracts sold under the Purchase and Sale Agreement through the Closing hereunder.
- Section 8.15 Site 20. Sellers acknowledge that there remain certain issues of concern to Purchaser regarding the property located at 3417 Fred George Rd., Tallahassee, Florida ("Site 20") that will not be resolved by Closing. As an inducement to Purchaser to consummate Closing without resolving such issues of concern related to Site 20, the Company shall: (a) assign, transfer and convey ("Assignment") to Sellers or their Affiliate that certain Lease dated March 9, 1987 between Company and Andrews Enterprises, Inc., as amended, with respect to Site 20 and all tangible and intangible property of the Company located at or used in connection with the Business at Site 20 (collectively, the "Site 20 Property"), and the Sellers or their Affiliate shall accept the Assignment and assume all obligations related thereto at or prior to Closing; and (b) use good faith efforts to enter into a mutually acceptable management agreement whereby Company will agree to operate a convenience store at Site 20 until the resolution of the zoning issues allowing the tenant under the Lease to have a permanent right to sell beer at Site 20 ("Zoning Resolution") during which time Company will pay a fee to Sellers or their Affiliate equal in amount to the amount of rent due under the Lease and, following the Zoning Resolution, the Company will have the right to purchase the Site 20 Property for a period ending sixty (60) days following the Zoning Resolution for the sum of Ten Thousand Dollars (\$10,000.00) by giving Sellers or their Affiliate thirty (30) days written notice of Company's decision either to repurchase the Site 20 Property or terminate the management agreement, subject to any terms and conditions that the Company and Sellers or their Affiliate mutually agree upon. The provisions of this Section shall survive Closing.

# IX. CONDITIONS PRECEDENT TO CLOSING.

- 9.1 <u>Sellers' Conditions Precedent</u>. The obligations of Sellers to consummate the transactions contemplated by this Agreement are subject to each of the following conditions:
- (a) The representations and warranties made by Purchaser in this Agreement shall be true in all material respects when made and on and as of the Closing as though such representations and warranties were made on and as of Closing. Sellers shall have received from Purchaser at Closing a satisfactory certificate to such effect signed by an authorized officer of Purchaser.
- (b) Purchaser shall have performed and complied in all material respects with all provisions of this Agreement required to be performed or complied with by Purchaser before or at Closing. Sellers shall have received from Purchaser at the Closing a satisfactory certificate to such effect, signed by an authorized officer of Purchaser.

- (c) Purchaser shall have executed and delivered to Sellers at the Closing each of the Purchaser Documents and such additional documents as may be reasonably requested by Sellers in order to consummate the transactions contemplated by this Agreement.
- (d) Purchaser shall have paid or made provisions acceptable to Sellers for the payment of all fees, costs and expenses for obtaining all environmental due diligence, surveys, title examinations, inventory audits, and other inspections performed in connection with the transactions contemplated pursuant to this Agreement.
- (e) Sellers shall have been released from any personal guaranties that they have given prior to the Closing related to the operation of the Business.
  - (f) Closing (as defined in the Purchase and Sale Agreement) shall have been consummated.
- 9.2 <u>Purchaser's Conditions Precedent.</u> The obligations of Purchaser to consummate the transactions contemplated by this Agreement are subject to each of the following conditions:
- (a) The representations and warranties made by Sellers in this Agreement shall be true in all material respects when made and on and as of the date of Closing as though such representations and warranties were made on and as of Closing except to the extent those representations and warranties may be impacted by the sale of the Property pursuant to the Purchase and Sale Agreement. Purchaser shall have received from Sellers at Closing satisfactory certificates to such effect signed by an authorized officer of each Sellers.
- (b) Sellers shall have performed and complied in all material respects with all provisions of this Agreement required to be performed or complied with by Sellers before or at Closing. Purchaser shall have received from Sellers at Closing satisfactory certificates to such effect signed by each Seller.
- (c) Sellers shall have executed and delivered to Purchaser at the Closing each of the Seller Documents and such additional documents as may be reasonably requested by Purchaser in order to consummate the transactions contemplated by this Agreement.
  - (d) Company shall have paid in full all debt secured by any Assets either prior to or simultaneous with Closing.
  - (e) From the Effective Date to the Closing Date, there shall not have been any Material Adverse Effect on the Company.
- (f) Each Seller shall have executed and delivered to Purchaser a general release in favor of Company and Purchaser and their Affiliates in a form to be mutually agreed upon by the Sellers and the Purchaser.
  - (g) Sellers shall have caused the Company to obtain all consents and approvals listed on <u>Schedule 5.2</u> hereto.
  - (h) Closing (as defined in the Purchase and Sale Agreement) shall have been consummated.
- (i) The Company shall have distributed to the Sellers all proceeds received by the Company under the Purchase and Sale Agreement, including but not limited to the Purchase Price (as that term is defined in the Purchase and Sale Agreement).
- 9.3 <u>Mutual Condition Precedent.</u> The obligations of each of the Parties to consummate the transactions contemplated by this Agreement are subject to each of the following condition: as of the Closing Date, there shall not be any claim or judgment of any nature or type threatened, pending or made by or before any governmental authority that questions or challenges the lawfulness of the transactions contemplated by this Agreement under any law or regulation or seeks to delay, restrain or prevent such transactions.

#### X. SURVIVAL: INDEMNIFICATION

10.1 <u>Survival</u>. Subject to the limitations and other provisions of this Agreement, the representations and warranties of the parties contained herein shall survive the Closing Date and shall remain in full force and effect until March 31, 2014, except with respect to the representations and warranties contained in Sections 5.1, 5.3, 5.4, and 5.12, and Sections 6.1 and 6.2, which shall survive until three (3) months after the expiration of the applicable statute of limitations (including any waivers or extensions thereof).

#### 10.2 Indemnification for Taxes.

- (a) To the extent not accrued as a liability in the Closing Balance Sheet, Sellers, jointly and severally (reserving unto each other any rights of contribution), shall indemnify Purchaser, Company and their respective Affiliates and hold each of them harmless (on an after-Tax basis) from and against (i) any and all Taxes of Company (or any predecessor company thereto or any Subsidiary) in respect of any period ending on or before the Closing Date, or in the case of a period that includes but does not end at the Closing Date, the portion thereof prior to and including the Closing Date (such period or portion, a "Pre-Closing Period"), except for any Taxes imposed pursuant to Section 1374 of the Code (and any analogous state and local provisions) resulting from the sale of the Property pursuant to the Purchase and Sale Agreement; (ii) any and all Taxes for which the Company may be or become liable by reason of (1) being a member of an affiliated, combined, consolidated, or unitary group at any time prior to the Closing, including under Treasury Regulation Section 1.1502-6 or any analogous or similar provision under any state, local or foreign Tax Law or (2) being a successor-in-interest or transferee of any other Person as a result of an event or transaction occurring prior to Closing; or (iii) the effect of any breach of a representation in Section 5.12 or covenant in Section 12.2 or other covenant with respect to Taxes, in each case, all reasonable costs (including reasonable attorneys' fees and related disbursements and expenses) incurred by Company, Purchaser or any of their Affiliates in connection therewith or in enforcing its rights hereunder.
- (b) The indemnities provided for in this Section 10.2 (i) shall apply notwithstanding any investigation made by Purchaser in connection with the transactions contemplated by this Agreement or its receipt or review of or comments on, any Return, (ii) shall

be separate and independent of any other indemnity provision contained herein and (iii) anything in this Agreement to the contrary notwithstanding, shall survive until three (3) months after the expiration of the applicable statute of limitations, including extensions or waivers thereof. The provisions of Sections 10.5 and 10.6 shall not apply to claims for indemnification under this Section 10.2.

- (c) Sellers shall promptly forward to Purchaser a copy of all written communications from a Tax authority received by any of the Sellers that relates to Company, its income, assets, payroll or operations, including any notice of a Tax Proceeding with respect to the Taxes of the Company. Purchaser shall promptly forward to Sellers' Representative a copy of all written communications from a Tax authority received by it for which the Sellers may be liable under this Section 10.2, including any notice of a Tax Proceeding.
- (d) Purchaser agrees not to settle or make any payment of an amount claimed to be due with respect to a proposed adjustment or undertake the defense or control of any Tax Proceeding for which Sellers may be liable under this Section 10.2 for at least fifteen (15) days after giving notice to Sellers pursuant to Section 10.2(c). If, within such fifteen (15)-day period, Purchaser receives a written request from Sellers that the proposed adjustments or Tax Proceeding will be contested or defended, which includes a statement of a reasonable basis in fact and in law for such contest, which includes an acknowledgement that the claim or Tax Proceeding is one to which the indemnity herein applies, Sellers shall have the right thereafter to undertake, conduct and control, through counsel of its own choosing and at its expense, the settlement and defense of such matter, and agree to keep Purchaser informed as to the progress of the matter, provided, however, that Sellers shall not enter into any settlement or compromise of any such matter without prior written consent of Purchaser, which consent shall not be unreasonably withheld, delayed or conditioned. Purchaser shall reasonably cooperate with Seller in connection with any such contest or Tax Proceeding. Notwithstanding the foregoing, Purchaser shall have the right thereafter to undertake, conduct and control, through counsel of its own choosing and at its expense, the settlement and defense of any matter that relates to a Straddle Period, and agree to keep Sellers' Representative informed as to the progress of the matter, provided, however, that Purchaser shall not enter into any settlement or compromise of any such matter which settlement or compromise would result in an indemnification obligation of Sellers under this Agreement without prior written consent of Sellers' Representative, which consent shall not be unreasonable withheld, delayed or conditioned. Sellers' Representative shall reasonably cooperate with Purchaser in connection with any such contest or Tax Pr
- (e) If Company (any successor thereto), Purchaser or their Affiliates are subject to any Taxes in any taxable period (or portion thereof) after the Closing Date with respect to the Oil Spill Claims (including any deemed or constructive receipt of the proceeds of such Oil Spill Claims), then Sellers hereby agree to indemnify Company (any successor thereto), Purchaser and their Affiliates from and against any and all incremental Taxes, if any, incurred by Company (any successor), Purchaser or their Affiliates related to the Oil Spill Claim. The Company shall also be permitted to reduce the amount of the distribution to be made under Section 8.1(e) to satisfy such indemnification obligation of Sellers. For example, if, in a taxable period, upon the receipt of proceeds of Oil Spill Claims, the Company must recognize the proceeds in its income for Tax purposes but is allowed a full deduction for Tax purposes for the distribution of such proceeds under Section 8.1 and, as a result, neither the Company, Purchaser

nor their Affiliates incur any Tax with respect to the Oil Spill Claims for that taxable period, then the Sellers would not have any obligations under this Section 10.2(e) for that taxable period; conversely, if the Company, Purchaser or their Affiliates incurs such a Tax, the Company shall be permitted to reduce the amount of the distribution to be made under Section 8.1 by the amount of such Tax and/or seek indemnification from the Sellers, as the case may be.

#### 10.3 <u>General Indemnity</u>.

- (a) Subject to the terms and conditions of this Article X, Sellers, jointly and severally (reserving unto each other any rights of contribution), hereby agree to indemnify and hold Company, Purchaser and LGP and their respective Affiliates harmless from and against all damages to and liabilities of Company, Purchaser and LGP, as the case may be, (including those resulting from or relating to demands, claims, actions or causes of action, assessments or other losses, costs and expenses relating thereto, interest and penalties thereon and reasonable attorneys' fees and related disbursements and other expenses in respect thereof) by reason of or resulting from (i) a breach of any representation or warranty of (A) Sellers contained in or made pursuant to this Agreement or the Seller Documents or (B) the Company contained in or made pursuant to the Purchase and Sale Agreement or the Seller Documents (as defined in the Purchase and Sale Agreement), (ii) the failure of (A) the Sellers duly to perform or observe any term, provision, covenant or agreement to be performed or observed by them pursuant to this Agreement or the Seller Documents or (B) the Company duly to perform or observe any term, provision, covenant or agreement to be performed or observed by the Company pursuant to the Purchase and Sale Agreement or the Seller Documents (as defined in the Purchase and Sale Agreement), or (iii) any claim by St. Joe Timberland Company of Delaware, L.L.C. (together with its successors and assigns, "Former Owner"), the former owner of 8140 West Highway 388, Panama City Beach, Florida (the "PCB Property"), in connection with that certain Purchase and Sale Agreement dated September 8, 2011 between Former Owner and the Company, a memorandum of which is recorded in the Public Records of Bay County, Florida in OR Book 3361, Page 1331, that the Former Owner is entitled to any profits arising from the sale of the PCB Property under the Purchase and Sale Agreement; provided, however, under no circumstances will Sellers have any obli
- (b) Subject to the terms and conditions of this Article X, Purchaser hereby agrees to indemnify, defend and hold Sellers harmless from and against all damages to and liabilities of Sellers (including those resulting from or relating to demands, claims, actions or causes of action, assessments or other losses, costs and expenses relating thereto, interest and penalties thereon and reasonable attorneys' fees and related disbursements and other expenses in respect thereof) by reason of or resulting from (i) a breach of any representation or warranty of Purchaser contained in or made pursuant to this Agreement, (ii) any failure of Purchaser duly to perform or observe any term, provision, covenant or agreement to be performed or observed by Purchaser pursuant to this Agreement, (iii) any personal guaranties or indemnities given by Sellers in favor of Hardee's Food Systems, Inc. or its Affiliates related to the operation of the Business that have not been released under Section 9.1(e) above, (iv) any liability of Sellers related to any third party consents or approvals listed on Schedule 5.2, or (v) any liability of Sellers incurred post-Closing as a result of any of them remaining on record as an officer or director of the Company with the Florida Division of Alcoholic Beverages and Tobacco pending

the approval of the application for a retail alcoholic beverage license submitted by Purchaser or its Affiliate.

(c) The parties hereby acknowledge and agree that their sole and exclusive remedy with respect to any and all claims relating to the subject matter of this Agreement (other than a claim for fraud or for specific performance of the terms of this Agreement) shall be pursuant to the indemnification provisions set forth in this Article X.

- (d) The parties shall take all reasonable steps to mitigate all liabilities and damages upon and after becoming aware of any event that could reasonably be expected to give rise to such liabilities and damages. In no event shall any party be liable for consequential, incidental, exemplary or punitive damages other than such damages awarded to a third party.
- 10.4 <u>Specific Indemnity</u>. Subject to the terms and conditions of this Article X, Sellers, jointly and severally (reserving unto each other any rights of contribution), hereby agree to indemnify and hold Company and Purchaser and their respective Affiliates harmless from and against all damages to and liabilities of Company and Purchaser, as the case may be, (including those resulting from or relating to demands, claims, actions or causes of action, assessments or other losses, costs and expenses relating thereto, interest and penalties thereon and reasonable attorneys' fees and related disbursements and other expenses in respect thereof) by reason of or resulting from the termination of that certain Deferred Compensation Agreement, dated on or about December 13, 2005 by and between the Company and James E. Lewis, Jr.
- 10.5 Third Party Claims. If any claim, assertion or proceeding by or in respect of a third party is made against an indemnified party or any event in respect of a third party occurs, and if the indemnified party intends to seek indemnity with respect thereto under this Article X or to apply any damage or liability arising therefrom to the U.S. Dollar amounts referred to in Section 10.6, the indemnified party shall promptly notify the indemnifying party of such claim in writing, provided that the failure by the indemnified party to give such notice shall not relieve the indemnifying party from its indemnification obligations hereunder, except if and to the extent that the indemnifying party is actually materially prejudiced thereby. The indemnifying party shall have thirty (30) days after receipt of such notice to undertake, conduct and control, through counsel of its own choosing and at its expense, the settlement or defense thereof, and the indemnified party shall cooperate with it in connection therewith; provided, however, that (a) the indemnifying party shall permit the indemnified party to participate in such settlement or defense through counsel chosen by the indemnified party, provided that the fees and expenses of such counsel shall be borne by the indemnified party, (b) the indemnifying party shall promptly reimburse the indemnified party for the full amount of any liability resulting from such claim and all related and reasonable expenses (other than the fees and expenses of counsel as aforesaid) incurred by the indemnified party within the limits of this Article X and subject to the U.S. Dollar amounts referred to in Section 10.6, (c) the indemnified party shall not, without the prior written consent of the indemnifying party, settle or compromise any claim or consent to the entry of any judgment which does not include as an unconditional term thereof the giving by the claimant or the plaintiff to the indemnified party a release from all liability in respect of such claim and (d) nothing herein shall

such claim in good faith, the indemnified party shall have the right to pay or settle any such claim; provided, however, that in such event it shall waive any right to indemnity therefor by the indemnifying party. If representation of the indemnified party, on the one hand, and the indemnifying party, on the other, by the same counsel would otherwise be inappropriate due to actual or potential differing interests between them, then the indemnified parties shall be entitled to engage separate legal counsel to participate in the defense of such claim at the sole expense of indemnifying party. If the indemnifying party does not notify the indemnified party within thirty (30) days after the receipt of the indemnified party's written notice of a claim of indemnity hereunder that it elects to undertake the defense thereof, the indemnified party shall have the right to contest, settle or compromise the claim in the exercise of its reasonable judgment at the expense of the indemnifying party.

- Limitation on Indemnities. Except with respect to the representations and warranties contained in Sections 5.1, 5.3, 5.4, 5.9, 5.12 or 5.18, or Sections 6.1 or 6.2, no claim for indemnification will be made by Purchaser or by the Sellers under Sections 10.3(a)(i), 10.3(b)(i) or 10.4 hereof, as the case may be, with respect to any individual item of liability or damage unless and to the extent that the aggregate of all such claims by Purchaser and LGP under, respectively, this Agreement and the Purchase and Sale Agreement, on the one hand, or by Sellers, on the other hand, shall be in excess of \$200,000, whereupon Purchaser or Sellers, as the case may be, shall be liable for all such claims, damages and liabilities, and the maximum aggregate liability of the Sellers, on the one hand, and Purchaser and LGP, on the other hand for such claims, damages and liabilities shall be the Escrow Amount. Payments by an indemnifying party pursuant to Section 10.3 shall be limited to the amount of any liability or damage that remains after deducting therefrom any insurance proceeds and any indemnity, contribution or other similar payment actually recovered by the indemnified party from any third party with respect thereto. Notwithstanding anything to the contrary contained in this Agreement, no claim by any party hereto may be asserted, nor may any action be commenced against any party hereto, for breach of any representation, warranty, covenant or agreement unless notice thereof is received in writing describing in reasonable detail the facts or circumstances with respect to the subject matter of such claim on or before the date on which the representation, warranty, covenant or agreement on which such claim or action is based ceases to survive as set forth in Section 10.1, irrespective of whether the subject matter of such claim or action shall have occurred before, on or after such date.
- 10.7 <u>Effect on the Share Purchase Price</u>. Any payment made under this Article X or Section 12.2(g) shall constitute an adjustment to the Share Purchase Price for all purposes, including federal, state and local Tax as well as financial accounting purposes, except as otherwise required by GAAP for financial accounting purposes only. Notwithstanding anything in this Agreement to the contrary, Purchaser shall have the right to offset any indemnification amount under this Agreement that has not been paid to Purchaser or its Affiliates against payment of any amounts that may be payable pursuant to any agreement between or among the parties.
- 10.8 <u>Investigation</u>. The indemnities provided for in this Agreement shall apply notwithstanding any investigation made by Purchaser in connection with the transactions contemplated by this Agreement or its receipt or review of or comments on, any Return.

#### XI. <u>TERMINATION</u>

- 11.1 <u>Termination by Sellers or Purchaser</u>. In the event that the closing of the transactions contemplated herein has not occurred on or before December 22, 2012, either Sellers or Purchaser shall have the right, exercisable by written notice to the other(s) delivered by telecopier and overnight delivery, to terminate this Agreement. Upon the termination of this Agreement neither party shall have any further obligations hereunder.
- 11.2 <u>Termination by Sellers</u>. Notwithstanding anything herein to the contrary, in the event that this Agreement has not prior to such date been terminated pursuant to its terms or by mutual consent of the parties, and either (a) the Closing (as defined in the Purchase and Sale Agreement) has not occurred by December 22, 2012 as a result of LGP's non-performance, default or breach of the Purchase and Sale Agreement or (b) the Closing has not occurred on or before December 22, 2012 as a result of Purchaser's non-performance, default or breach of this Agreement, including the failure or refusal of Purchaser to satisfy any of the conditions precedent to the obligations of the Company set forth in Section 9.1 above, Purchaser and Sellers agree that based upon the circumstances now existing, known and unknown, it would be impractical or extremely difficult to establish Sellers' damages by reason of LPG's

non-performance, default or breach of the Purchase and Sale Agreement or Purchaser's non-performance, default or breach of this Agreement prior to Closing. Accordingly, Purchaser and Sellers agree that it would be reasonable in the event of such non-performance, default or breach by LPG or Purchaser to award Sellers "liquidated damages" equal to the amount of the Deposit. Sellers and Purchaser acknowledge and agree that the foregoing amount is reasonable as liquidated damages and shall be the sole and exclusive remedy of the Company and Sellers in lieu of any other relief, right or remedy, at law or in equity, to which the Company or Sellers might otherwise be entitled by reason of such non-performance, default or breach by LGP under the Purchase and Sale Agreement or Purchaser hereunder.

- 11.3 Termination by Purchaser. Notwithstanding anything herein to the contrary, in the event that either (a) the Purchase and Sale Agreement has not prior to December 22, 2012 been terminated pursuant to its terms or by the mutual consent of the Company and LGP, and the Closing (as defined in the Purchase and Sale Agreement) has not occurred on or before December 22, 2012 as a result of the Company's non-performance, default or breach of the Purchase and Sale Agreement or (b) this Agreement has not prior to December 22, 2012 been terminated pursuant to its terms or by mutual consent of the parties, and the Closing has not occurred on or before December 22, 2012 as a result of Sellers' non-performance, default or breach of this Agreement, including the failure or refusal of Sellers to satisfy any of the conditions precedent to the obligations of Purchaser set forth in Section 9.2 above, LGP and Purchaser may, at their option and as their sole and exclusive remedy, either (a) terminate the Purchase and Sale Agreement and/or this Agreement in which event the Deposit shall immediately be returned to Purchaser or (b) bring an action to enforce specific performance of the Purchase and Sale Agreement and/or this Agreement (including the provisions of Article X) according to the terms thereof and hereof by all means available at law or in equity.
- 11.4 Acknowledgment as to Liquidated Damages. PURCHASER AND SELLERS HEREBY ACKNOWLEDGE AND AGREE THAT THE AMOUNT OF SELLERS' DAMAGES IN THE EVENT OF A BREACH OF THIS AGREEMENT BY PURCHASER

WOULD BE DIFFICULT OR IMPOSSIBLE TO DETERMINE AND THAT AN AMOUNT EQUAL TO THE AGREED DAMAGES IS THE PARTIES' BEST AND MOST ACCURATE ESTIMATE OF THE DAMAGES SELLERS WOULD SUFFER IN THE EVENT THE TRANSACTION PROVIDED FOR IN THIS AGREEMENT FAILS TO CLOSE AS A RESULT OF THE BREACH OR DEFAULT BY PURCHASER AND THAT SUCH ESTIMATE IS REASONABLE UNDER THE CIRCUMSTANCES EXISTING AS OF THE DATE OF THIS AGREEMENT AND UNDER THE CIRCUMSTANCES THAT SELLERS AND PURCHASER REASONABLY ANTICIPATE WOULD EXIST AT THE TIME OF SUCH BREACH. PURCHASER AND SELLERS HEREBY ACKNOWLEDGE THAT PURSUANT TO THE ESCROW AGREEMENT, ON SUCH A BREACH BY PURCHASER, ESCROW AGENT SHALL DELIVER THE DEPOSIT TO SELLERS, WHO SHALL APPLY THE DEPOSIT AGAINST THE AGREED DAMAGES.

#### XII. POST CLOSING AGREEMENTS.

- 12.1 <u>Further Accommodations</u>. In case at any time after the Closing any further action is necessary, appropriate or convenient to carry out the purposes of this Agreement, each of the Parties will take such further action as any other Party reasonably may request, all at the sole cost and expense of the requesting Party.
- 12.2 <u>Certain Tax Matters</u>. The following provisions shall govern the allocation of responsibility as between Purchaser and Company, on the one hand, and Sellers, on the other hand, for certain tax matters following the Closing:
- (a) Tax Returns. For any tax periods ending on or before the Closing Date, Sellers' Representative shall prepare or cause to be prepared, at Sellers' expense, and timely file all Tax Returns for Company which are required to be filed after the Closing Date with respect to such tax periods (the "Pre-Closing Returns"). Subject to the requirements of applicable Tax Law, each Pre-Closing Return shall be prepared in a manner consistent with past practices of the Company, but in all cases shall be in conformity with the Code, the United States Treasury Regulations and other primary authority. Sellers and Purchaser agree that any income or gain recognized by the Company from the sale of the Property pursuant to the Purchase and Sale Agreement shall be reported as long-term capital gain (except as otherwise required under Section 1245 and Section 1250 of the Code or with respect to the Property placed in service by the Company within one year prior to the closing under the Purchase and Sale Agreement) and not subject to any Taxes imposed under Section 1374 of the Code (any analogous state or local provision) unless the sale of any Real Property acquired prior to January 1, 2006 results in a gain that was unrealized as of December 31, 2005 (the "Reporting Position"). The Sellers' Representative shall deliver any Pre-Closing Return (along with associated tax workpapers) to Purchaser at least thirty (30) days prior to the date on which such Pre-Closing Return is required to be filed (taking into account extension) and in the case of a return due within 30 days after the Closing Date as soon as practical. If Purchaser disputes any item on any such Pre-Closing Return prepared by the Sellers' Representative, it shall, within ten (10) days of receiving such Pre-Closing Return, notify the Sellers' Representative of such disputed item (or items) and the basis for its objection. Sellers' Representative and Purchaser shall act in good faith to resolve any such dispute prior to the date on which the relevant Pre-Closing Return is

filed. If Sellers' Representative and Purchaser cannot resolve any disputed item, the item in question shall be resolved by the Independent Auditor. The fees and expenses of the Independent Auditor attributable to such dispute shall be borne equally by the Sellers and the Purchaser. If the Independent Auditor is unable to resolve the dispute no later than 3 days prior to the filing date of the Pre-Closing Return at issue (taking into account applicable extensions), then such Pre-Closing Return shall be filed as prepared by the Sellers' Representative, subject to subsequent amendment, if any, necessary to reflect Independent Auditor's final resolution of the disputed items. Sellers' Representative shall provide a copy of such Pre-Closing Returns to Purchaser promptly after the filing of such Pre-Closing Returns.

(b) Straddle Periods. For a taxable period that begins on or before the Closing Date and ends after Closing Date (a "Straddle Period"), Purchaser shall prepare or cause to be prepared, at Purchaser's expense, and timely file all Tax Returns for the Company which are required to be filed after the Closing Date with respect to such Straddle Periods (the "Straddle Returns"). Subject to the requirements of applicable Tax Law, each Straddle Return shall be prepared in a manner consistent with past practices of the Company, but in all cases shall be in conformity with the Code, the United States Treasury Regulations and other primary authority, and in accordance with the Reporting Position. The Purchaser shall deliver any Straddle Return (along with associated tax workpapers) relating to Straddle Period which shows a Tax owing allocable to a Pre-Closing Period to the Sellers' Representative for its review and comment at least thirty (30) days prior to the date on which such Straddle Return is required to be filed (taking into account extensions) or, in the case of a Straddle Return due within thirty days after the end of the taxable period to which that return relates, as soon as practical. If the Sellers' Representative disputes any item on any such Straddle Return, it shall, within ten (10) days of receiving such Straddle Return, notify the Purchaser of such disputed item (or items) and the basis for its objection. Sellers' Representative and Purchaser shall act in good faith to resolve any such dispute prior to the date on which the

relevant Straddle Return is required to be filed. If Sellers' Representative and Purchaser cannot resolve any disputed item, the item in question shall be resolved by the Independent Auditor. The fees and expenses of the Independent Auditor attributable to such dispute shall be borne equally by the Sellers and the Purchaser. If the Independent Auditor is unable to resolve the dispute no later than 3 days prior to the filing date of the Straddle Return at issue (taking into account applicable extensions), then such Straddle Return shall be filed as prepared by Purchaser, subject to subsequent amendment, if any, necessary to reflect Independent Auditor's final resolution of the disputed items. Purchaser shall provide a copy of such Tax Returns to Sellers' Representative promptly after the filing of such Tax Returns.

(c) Payment of Taxes. All Taxes shown as due and owing on any Pre-Closing Return or on any Straddle Return allocable to a Pre-Closing Period (as determined under Section 10.2(a)(i)) shall be borne by the Sellers to the extent that such Taxes exceed the amount of such Taxes accrued as a liability in the Closing Balance Sheet and shall be borne by the Purchaser to the extent that such Taxes are equal to or less than the Taxes accrued as a liability in the Closing Balance Sheet. If such Tax Returns will be filed by the Purchaser, the amount of such Taxes for which Sellers are responsible shall be paid by the Sellers' Representative (on behalf of the Sellers) to the Purchaser no later than three (3) Business Days prior to the due date for filing such Tax Returns and Purchaser shall remit such Taxes to the appropriate Governmental Authority. If such Tax Returns will be filed by the Sellers' Representative, the amount of such Taxes for

which Purchaser is responsible shall be paid by the Purchaser to the Sellers' Representative no later than three (3) Business Days prior to the due date for filing such Tax Returns and Sellers' Representative shall remit such Taxes to the appropriate Governmental Authority.

- (d) Tax Periods. For purposes of this Agreement, if the Company is permitted but not required under applicable state, local, or foreign Tax Laws to treat the end of the Closing Date as the last day of a taxable period with respect to any Tax, then the Parties shall treat that day as the last day of a taxable period. The Parties acknowledge that, for federal and applicable state and local income and franchise tax purposes, pursuant to Treasury Regulations Section 1.1362-3(b)(3), the Company's items of income, gain, loss, deduction and credit will not be allocated pro rata and will be allocated to each short taxable year resulting from the transaction on the basis of the Company's method of accounting during each short taxable year as determined under Section 446 of the Code. In the event Treasury Regulations Section 1.1362-3(b)(3) is for any reason not applicable, the Sellers and the Purchaser agree to make all necessary elections to apply the "closing of the books" method for allocating items between the short years. For purposes of this Agreement, in the case of any Tax (other than Property Taxes) for a Straddle Period for which an election is not available to terminate the tax year or apply a "closing of the books" method as of the Closing Date, such Taxes shall be allocated between the Pre-Closing Period and the balance of the period based on a closing of the books as of the end of the Closing Date. Property Taxes shall be allocated in accordance with Section 4.5.
- (e) <u>Amendments of Tax Returns</u>. Except as otherwise required under applicable Tax Law, permitted in Section 12.2(a) or (b) or in settlement of a Tax Proceeding, Purchaser shall not file, nor cause to be filed, any amended Pre-Closing Return or any amended Tax Return for any Tax period ending before the Closing Date and filed before the Closing Date without the written consent of Sellers' Representative, which consent shall not be unreasonably withheld, delayed or conditioned.
- (f) <u>Cooperation on Tax Matters.</u> Sellers and Purchaser shall (i) each provide the other, and Purchaser shall cause Company to provide Sellers with such assistance as may reasonably be requested by any of them or their professional advisors in connection with the preparation of any Tax Return or the audit or other examination by any taxing authority, or judicial or administrative proceedings, relating to liability for Taxes for which Sellers may possibly be responsible under this Agreement, (ii) each retain and provide the other, and Purchaser shall cause Company to retain and provide Sellers, with any records or other information in the possession of such Party or his, her or its professional advisors or other representative(s) which may be relevant to such Tax Return, audit or examination, proceeding or determination, and (iii) each provide the other with any final determination of any such audit or examination, proceeding or determination that affects any amount required to be shown on any Tax Return of Company for any period ending on or prior to the Closing Date. Without limiting the generality of the foregoing, Purchaser shall retain, and shall cause Company to retain, and Sellers shall retain until the applicable statutes of limitations (including any extensions) have expired, copies of all Tax Returns, supporting work schedules and other records or information (to the extent such items are the possession of such Party or his, her or its professional advisors or other representative(s)) which may be relevant to such Tax Returns for all tax periods or portions thereof ending on, before or including the Closing Date and shall not destroy or

otherwise dispose of any such records without first providing the other Party with a reasonable opportunity to review and copy the same.

(g) <u>Special Tax Indemnity</u>. Purchaser agrees to indemnify and hold the Sellers harmless from and against, any additional federal income Taxes (taking into account any subsequent adjustments made by any Governmental Authority) incurred by Sellers on their share of the income or gain resulting from the sale of the Property pursuant to the Purchase and Sale Agreement being subject to a federal income tax rate greater than a 15% rate ("Additional Shareholder Taxes"), plus an additional amount ("Tax Gross-Up Amount") as may be necessary so that the Sellers' net proceeds from the foregoing indemnification payments equal the amount of the Additional Shareholder Taxes after taking into account the federal income taxes imposed on the Sellers on the receipt of the Additional Shareholder Taxes and Tax Gross-Up Amount. The Additional Shareholder Taxes and the Tax Gross-Up Amount are collectively referred to as the "Total Shareholder Taxes". At Closing, Purchaser will deposit \$623,349 (the "Tax Escrow") with Escrow Agent which represents a preliminary estimate of the Total Shareholder Taxes.

No later than 60 days after the Closing Date, Sellers' accountants shall determine the amount of Total Shareholder Taxes in accordance with the Reporting Position and taking into account 2012 and 2013 federal income tax rates, as applicable ("Updated Total Shareholder Taxes") and shall provide to the Sellers' Representative and Purchaser a calculation of the updated amount of the Additional Shareholder Taxes and the Tax Gross-Up Amount ("Closing Tax Schedule"), along with supporting workpapers. Each Party shall provide to Sellers' accountants comments to the Closing Tax Schedule no later than 10 days after that Party's receipt of the Closing Tax Schedule. Purchaser and Sellers' Representative shall mutually agree upon any changes to the Closing Tax Schedule, and Sellers' accountants thereafter shall promptly issue a final Closing Tax Schedule that shall be binding on the Parties ("Final Closing Tax Schedule"). If no comments are submitted by the Parties within the 10-day period, the Closing Tax Schedule shall be become the Final Closing Tax Schedule (and shall be deemed issued as final as of the end of that 10-day period). In the event the Purchaser and Sellers' Representative cannot mutually agree upon changes to the Closing Tax Schedule, then Purchaser and Sellers' Representative mutually shall select an independent accountant (the "Independent Accountant") to calculate the Updated Total Shareholder Taxes and prepare the Final Closing Tax Schedule based on the comments received above. The Parties agree to prepare all Tax Returns of the Company and each Seller agrees to prepare their individual income tax returns in a manner consistent with the Final Closing Tax Schedule (unless required to do so under applicable Tax Law). On or before April 10, 2013, in the event the Updated Total Shareholder Taxes equal or exceed the Tax Escrow,

the Purchaser and the Sellers' Representative shall provide notice to the Escrow Agent to release the Tax Escrow to the Sellers' Representative (on behalf of the Sellers), and the Purchaser shall pay directly to the Sellers' Representative (on behalf of the Sellers) an amount equal to the excess, if any, of (i) the amount of the Updated Total Shareholder Taxes as reflected on the Final Closing Tax Schedule over (ii) the Tax Escrow released by the Escrow Agent. On or before April 10, 2013, in the event the Tax Escrow exceeds the Updated Total Shareholder Taxes, the Purchaser and the Sellers' Representative shall provide notice to the Escrow Agent to release (i) to the Sellers' Representative (on behalf of the Sellers) an amount equal to the Updated Total Shareholder Taxes reflected on the Final Closing Tax Schedule; and (ii) to the Purchaser an amount equal to the excess of (A) the Tax

Escrow over (B) the amount of the Updated Total Shareholder Taxes reflected on the Final Closing Tax Schedule. Purchaser shall be responsible for and pay the costs and fees of the Independent Accountant, if any, up to \$50,000 in determining the amount of the Updated Total Shareholder Taxes, and Sellers shall be responsible for and pay any costs and fees in excess of \$50,000. The indemnities provided for in this Section 12.2(g) shall survive until three (3) months after the expiration of the applicable statute of limitations, including extensions or waivers thereof.

- (h) Tax Audit. If there is any proposed claim or adjustment in any audit, examination or similar proceedings with respect to the federal, state, or local income tax returns of the Company or Sellers, which, if successful, might result in an indemnity obligation owed by the Purchaser pursuant to Section 12.2(g) or result in the imposition of Taxes under Section 1374 of the Code (and any analogous state or local provision) against the Company with respect to any income or gain recognized upon the sale of the Property pursuant to the Purchase and Sale Agreement, then Sellers' Representative shall promptly and in any event no more than fifteen (15) days following the Sellers' receipt of such proposed claim or adjustment, give written notice to the Purchaser of such claim or adjustment (a "Special Tax Claim"); provided, however, the failure of the Sellers' Representative to give such notices shall only relieve the Purchaser from its indemnification obligations hereunder to the extent it is actually prejudiced by such failure. Purchaser shall control all proceedings and may make all decisions taken in connection with the Special Tax Claim (including any settlement and selection of counsel) at its own expense. Sellers and Sellers' Representative shall cooperate with Purchaser, as reasonably requested by Purchaser, with respect to the contest of any Special Tax Claim. The procedures of this Section 12.2(h) shall apply to Special Tax Claims and not Section 10.2(d). In the event that a settlement or final resolution of the Special Tax Claim results in the imposition of additional Taxes owed by the Sellers greater than the Additional Shareholder Taxes on the Final Closing Tax Schedule, the Purchaser promptly will make a payment to Sellers' Representative (on behalf of the Sellers) for such difference upon written request from the Sellers' Representative. For purposes of clarity, it is the intent of the Parties that under no circumstances shall Sellers have any responsibility for any taxes incurred by the Company as a resu
- (i) Transfer Taxes. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees including, without limitation, any interest, penalties or additions thereto ("Transfer Taxes") incurred in connection with this Agreement and the transactions contemplated hereby and any other similar Tax that may be imposed, shall be paid by Sellers when due, and Sellers will file all necessary Tax Returns and other documentation with respect to all such transfer, documentary, sales, use, stamp, registration and other Taxes and fees, and, if required by applicable Tax Law, the Purchaser will join in the execution of any such Tax Returns and other documentation. The parties shall take any reasonable action that may be available to reduce or eliminate any Transfer Taxes.

Section 12.3 Store 99. Purchaser and Sellers shall cooperate post-Closing to seek the acknowledgment of Former Owner or its Affiliates that (a) the Improvements (as defined in that certain Memorandum of Agreement recorded on October 31, 2011 at OR Book 3361, page 1331 of the official records of Bay County, Florida) have been completed in accordance with Section 12 of the Memorandum of Agreement and the underlying purchase and

sale agreement and (b) the repurchase option set forth in Section 16.1 of the Memorandum of Agreement and the underlying purchase and sale agreement is null and void.

# XIII. <u>DEFINITIONS</u>

The following capitalized terms used in this Agreement shall have the meanings set forth below:

- "Acquisition Transaction" has the meaning set forth in Section 8.10(a).
- "Additional Deposit" has the meaning set forth in Section 2.3.
- "Additional Shareholder Taxes" has the meaning set forth in Section 12.2(g).
- "Adjusted Share Purchase Price" has the meaning set forth in Section 2.1.
- "Affiliate" means, with respect to any specified Person, any other Person who, directly or indirectly, controls, is controlled by, or is under common control with such Person.
  - "Agreed Damages" has the meaning set forth in Section 10.2.
  - "Agreement" has the meaning set forth in the Preamble.
- "Assets" means the assets used by the Company in the operation of the Business and owned by the Company on the Closing Date including the Inventory, the Tangible Personal Property, the Owned Real Properties, the Leased Real Properties, and the Fuel Equipment.
  - "AULs" has the meaning set forth in Section 8.7(b).
  - "Balance Sheet Date" has the meaning set forth in Section 5.11.
  - "Business" means the ownership and operation of the convenience stores at the Locations.

- "Cash Drawer" means an operating cash balance of \$300.00 at the Locations.
- "CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.
- "Closing" has the meaning set forth in Section 4.1.
- "Closing Balance Sheet" has the meaning set forth in Section 2.2(b).
- "Closing Date" has the meaning set forth in Section 4.1.
- "Closing Tax Schedule" has the meaning set forth in Section 12.2(g).
- "Closure Letter" has the meaning set forth in Section 8.7.
- "Code" means the U.S. Internal Revenue Code of 1986, as amended.
- "Company" has the meaning set forth in the Preamble.
- "Confidential Information" has the meaning set forth in Section 7.2.
- "Confidentiality Agreement" has the meaning set forth in Section 7.2.
- "Contract" means any agreement, contract, note, mortgage, indenture, Lease, arrangement or other obligation, excluding purchase orders.
- "Current Assets" means the book value of the current assets listed on <u>Schedule 13</u>; provided, however, the book value of any Inventory items shall be adjusted as contemplated by Section 3.3.
  - "Current Liabilities" means the book value of the current liabilities listed on Schedule 13.
  - "Deposit" has the meaning set forth in Section 2.3.
  - "Effective Date" has the meaning set forth in the Preamble.
  - "Employee Benefit Plan" has the meaning set forth in Section 5.18(a).
  - "Environmental Escrow Agreement" has the meaning set forth in Section 8.7(c).
  - "Environmental Escrow Amount" has the meaning set forth in Section 8.7(c)(ii).
- "Environmental Laws" means all (i) Laws relating to protection of the environment, natural resources or human and occupational health and safety, including Laws relating to Hazardous Substances, or Laws otherwise relating to the manufacture, formulation, generation, processing, distribution, use, treatment, storage, release, transport, remediation, abatement, cleanup or handling of Hazardous Substances, (ii) Laws with regard to recordkeeping, notification, disclosure and reporting requirements respecting Hazardous Substances, and (iii) Laws relating to the condition, management or use of natural resources.
- "Environmental Liabilities" means all known liabilities of the Company arising in connection with the Owned Real Properties or Leased Real Properties, which arise under or relate to a violation by Company of Environmental Laws.
- "Environmental Remediation and Indemnity Agreement" means that certain Environmental Remediation and Indemnity Agreement by and among the Company and the Sellers dated on the Closing Date and delivered at Closing.
  - "ERISA" has the meaning set forth in Section 5.18(a).
  - "ESA" has the meaning set forth in Section 7.3.
  - "Escrow Account" has the meaning set forth in Section 4.4(a).
  - "Escrow Agent" has the meaning set forth in Section 2.3.
  - "Escrow Agreement" has the meaning set forth in Section 2.3.
  - "Escrow Amount" has the meaning set forth in Section 4.4(a).
  - "Estimated Total Shareholder Taxes" has the meaning set forth in Section 12.2(g).
  - "Estimated Tax Schedule" has the meaning set forth in Section 12.2(g).
  - "Facilities" has the meaning set forth in Section 5.19.
  - "FDEP" means the Florida Department of Environmental Protection.

- "Final Closing Tax Schedule" has the meaning set forth in Section 12.2(g).
- "Financial Statements" has the meaning set forth in Section 5.11.
- "FIRPTA" has the meaning set forth in Section 5.9.
- "Fiscal Year" has the meaning set forth in Section 4.5.
- "Food Service Inventory" means all items used in the preparation or dispensing at the Locations of food items for immediate consumption, including such resulting food items.
  - "Former Owner" has the meaning set forth in Section 10.3(a).
- "Fuel Equipment" means all fuel fixtures and equipment now attached to or used in connection with the Owned Real Properties and Leased Real Properties, including without limitation, all petroleum pumps and dispensers, underground and aboveground fuel storage tanks, canopies, fuel lines, fittings and connections used in the Ordinary Course of Business at the Owned Real Properties and Leased Real Properties to receive, store and/or dispense fuels.
  - "GAAP" shall mean generally accepted accounting principles of the United States of America consistently applied, as in effect from time to time.
  - "Gasoline Taxes" has the meaning set forth in Section 3.3(e).
- "Governmental Authority" means any government or political subdivision, whether federal, state, local or foreign, or any agency or instrumentality of any such government or political subdivision, or any federal, state, local or foreign court or arbitrator.
- "Hazardous Substances" means (a) any and all radioactive materials or wastes, petroleum (including crude oil or any fraction thereof) or petroleum distillates, asbestos or asbestos containing materials, urea formaldehyde foam, and (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "contaminants," "toxic pollutants," or words of similar meaning and regulatory effect under any applicable Environmental Law.
  - "Independent Accountant" has the meaning set forth in Section 12.2(g).
  - "Independent Auditor" has the meaning set forth in Section 3.1.
  - "Independent Consultant" has the meaning set forth in Section 8.7(c).
  - "Initial Deposit" has the meaning set forth in Section 2.3.
- "Inventory" means all Merchandise Inventory, Supplies Inventory, Food Service Inventory and Petroleum Inventory that are present at the Locations at Closing.
- "Intellectual Property" shall mean any of the following: (a) patents and patent applications, (b) registered and unregistered trademarks and service marks, trademark registrations, pending trademark and service mark registration applications, and similar reservations of marks, (c) registered and unregistered copyrights, copyright registrations, and applications for copyright registrations, (d) Internet domain names and (e) trade secrets.
- "Knowledge" means, with respect to Sellers, actual knowledge of James E. Lewis, Jr., James E. Lewis, III and Reid D. Lewis after reasonable inquiry of employees of the Company who are responsible for such area of inquiry.
  - "Law" or "Laws" shall mean all applicable local, state, or federal laws, statutes, ordinances, or administrative or judicial decisions.
- "Leased Real Properties" means Leases, together with all of Company's interest under such Leases in the buildings, improvements, easements and appurtenances thereon and thereto.
- "Leases" means any real property leases with Company as lessee covering the Locations more particularly described as leased real properties on <u>Schedule 5.19</u>.
  - "LGP" shall mean LGP Realty Holdings LP, a Delaware limited partnership.
- "Liabilities" shall mean any and all debts, liabilities and obligations, whether accrued or fixed, known or unknown, absolute or contingent, matured or unmatured or determined or determinable.
  - "Liens" shall mean any lien, security interest, mortgage, encumbrance or charge of any kind.
  - "Locations" has the meaning set forth in the Preamble.
- "Material Adverse Effect" means a material adverse effect on the Business, Assets or financial condition of Company except for any adverse effect resulting from general economic conditions applicable to the securities and banking industry.
  - "Material Contracts" has the meaning set forth in Section 5.16.
  - "Matrix" has the meaning set forth in Section 8.5.

"Merchandise Inventory" means saleable items of merchandise of every type and description at the Locations as of the Closing excluding Petroleum Inventory, Supplies Inventory and Food Service Inventory. For purposes of determining Merchandise Inventory, "saleable" shall mean all items other than (1) items that violate applicable government code specifications; (2) spoiled or damaged items, including packaging thereof; and (3) items that are stale dated as of the date of the inventory audit.

"Notice of Disagreement" has the meaning set forth in Section 2.2(b).

"Oil Spill Claims" has the meaning set forth in Section 8.1.

"Ordinary Course of Business" means the ordinary course of business consistent with past custom and practice (including with respect to quantity and frequency).

"Owned Real Properties" means all real property owned by the Company comprising the Locations more particularly described as owned real properties on <u>Schedule 5.19</u>, together with all buildings, improvements, easements and appurtenances thereon and thereto, subject only to the Permitted Encumbrances.

"Outside Date" has the meaning set forth in Section 4.1.

"Parties" means Purchaser and Sellers.

"Party" means either the Purchaser, on the one hand, or the Sellers, on the other hand.

"PCB Property" has the meaning set forth in Section 10.3(a).

"Permits" means all operating permits, Fuel Equipment permits or tank permits, underground and aboveground storage tank notifications or registrations, if any, and other permits, licenses, filings and other governmental authorizations, agreements, contracts, and approvals.

"Permitted Encumbrances" means (i) real estate taxes assessed for the Fiscal Year in which the Closing takes place and not yet due and payable, which taxes shall be adjusted as provided in this Agreement; (ii) existing encroachments, easements, rights of way, covenants, reservations and restrictions in the chain of title to the Owned Real Properties or existing thereon which do not substantially interfere with the use of the property for the retail sale of petroleum products and food stuffs; (iii) all building, zoning and historical Laws, rules and regulations affecting the Owned Real Properties and Leased Real Properties; (iv) parties in possession other than Company pursuant to any third party tenant leases; and (v) matters of survey which do not substantially interfere with the use of the Owned Real Properties for the retail sale of petroleum products and food stuffs.

"Person" means any individual, partnership, limited liability company, corporation, cooperative, trust or other entity.

"Petroleum Inventory" means saleable gasoline, kerosene and diesel fuel at the Locations as of the Closing. For purposes of determining Petroleum Inventory, "saleable" shall mean

conforming to the octane, brand and applicable requirements for sales of fuel from the Fuel Equipment at the Locations.

"Physical Inventory" has the meaning set forth in Section 3.1.

"Pre-Closing Period" has the meaning set forth in Section 10.2

"Pre-Closing Returns" has the meaning set forth in Section 12.2(a).

"Property" has the same meaning as that term is used in the Purchase and Sale Agreement.

"Property Taxes" has the meaning set forth in Section 4.5.

"Purchase and Sale Agreement" means that certain Sale and Purchase Agreement by and between the Company and LGP which will be entered into on the Effective Date.

"Purchaser" has the meaning set forth in the Preamble.

"Purchaser Documents" has the meaning set forth in Section 6.1.

"QSRs" means quick service restaurants.

"Release" has the meaning specified in 42 U.S.C. §9601(22).

"Remedial Measures" has the meaning set forth in Section 8.7.

"Reporting Position" has the meaning set forth in Section 12.2(a).

"Securities Act" has the meaning set forth in Section 6.8.

"Seller Documents" has the meaning set forth in Section 5.1.

- "Sellers" has the meaning set forth in the Preamble.
- "Sellers' Representative" means James E. Lewis, III.
- "Share Purchase Price" has the meaning set forth in Section 2.1.
- "Share Purchase Price Calculation" has the meaning set forth in Section 2.2(b).
- "Shares" has the meaning set forth in the Preamble.
- "Special Tax Claim" has the meaning set forth in Section 12.2(h).
- "Straddle Period" has the meaning set forth in Section 12.2(b).
- "Straddle Returns" has the meaning set forth in Section 12.2(b).
- "Supplies Inventory" means consumable operating items not intended for retail sale at the Locations as of the Closing and excluding any forms.
- "Takeover Proposal" has the meaning set forth in Section 8.10(a).
- "Tangible Personal Property" means all furniture, fixtures, equipment and other tangible personal property owned by Seller and now situated at the Owned Real Properties and Leased Real Properties.
  - "Tax Escrow" has the meaning set forth in Section 12.2(g).
  - "Tax Gross-Up Amount" has the meaning set forth in Section 12.2(g).
  - "Tax Law" means any Law with respect to Taxes.
  - "Tax Proceeding" has the meaning set forth in Section 5.12(b).
- "Tax Return" means any return, declaration, report, estimate, information return and statement (including any attachment, amendment or schedule thereto) required to be filed, or which is otherwise filed, with any Governmental Authority in respect of any Tax.
  - "Tax Schedule" has the meaning set forth in Section 12.2(i).
- "Taxes" shall mean (i) all taxes, assessments, charges, duties, fees, levies or other governmental charges including all United States federal, state, local, foreign and other income, franchise, profits, capital gains, capital stock, transfer, sales, use, occupation, property, excise, severance, windfall profits, stamp, license, payroll, withholding and other taxes, assessments, charges, duties, fees, levies or other governmental charges, all estimated taxes, deficiency assessments, additions to tax, penalties and interest and (ii) any liability for the payment of or in respect of any amounts of the type described in clause (i) of this definition as a result of being a member of an affiliated, combined or unitary group for Tax purpose or, as a result of any tax sharing or tax allocation agreement, arrangement or understanding or as a result of being liable for another Person's taxes as a transferee or successor, by contract or otherwise.
  - "Total Shareholder Taxes" has the meaning set forth in Section 12.2(g).
  - "Transfer Taxes" has the meaning set forth in Section 12.2(k).
  - "Treasury Regulations" means the income tax Treasury Regulations promulgated under the Code, as the same may be amended from time to time.
  - "Updated Total Shareholder Taxes" has the meaning set forth in Section 12.2(g).
  - "USPTO" means the United States Patent and Trademark Office.
  - "Valuation" has the meaning set forth in Section 12.2(h).

#### XIV. MISCELLANEOUS

- 14.1 <u>Payment of Expenses and Fees</u>. Except as otherwise provided in this Agreement, Purchaser and Sellers shall each bear their own costs and expenses, including attorneys' fees, incurred in connection with the transactions contemplated by this Agreement.
- 14.2 <u>Entire Agreement</u>. This Agreement, including the exhibits and other writings referred to herein or delivered pursuant hereto, together with the Purchase and Sale Agreement, constitutes the entire agreement between Sellers and Purchaser with respect to the subject matter hereof and supersedes all prior oral or written agreements, commitments or understandings with respect thereto, except the Confidentiality Agreement. No amendment hereof shall be binding on the Parties unless in writing and signed by authorized representatives of all Parties hereto.
- 14.3 <u>No Third Party Beneficiaries</u>. This Agreement is not intended to and does not confer any rights or obligations on any party that is not a signatory to this Agreement.
- 14.4 <u>Business Days</u>. If the day for performance of any action described in this Agreement shall fall on a Saturday, Sunday or a day on which the banks are closed in the State of Florida the time for such action shall be extended to the next business day after such Saturday, Sunday or day on which the

banks are closed.

- 14.5 <u>Governing Law</u>. This Agreement shall be deemed to be a contract entered into in the State of Florida and it and all matters arising out of the transactions contemplated hereby or related thereto shall be governed, construed and interpreted in all respects according to the Laws of the State of Florida, without reference to principles of conflicts of law thereof.
- 14.6 <u>Venue</u>. Any action to enforce this Agreement, or to interpret or construe the meaning of this Agreement, shall be brought in the Circuit Court for Bay County, Florida or for matters that require Federal Jurisdiction, the Federal District Court for the Northern District of Florida, Panama City Division, and Purchaser hereby expressly waives venue in any other Federal or State court other than in the State of Florida.

#### 14.7 Obligations of Parties; Successors and Assigns

- (a) Subject to following provisions, this Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, personal representatives, successors and assigns. If more than one person or entity is named as Purchaser, the term "Purchaser" shall refer to each person or entity so named and any one or more of them in any combination, and the representations, warranties, covenants, obligations and liabilities of Purchaser herein shall constitute their joint and several representations, warranties, covenants, obligations and liabilities.
- (b) Without the prior written consent of Sellers and in Sellers' sole discretion, Purchaser shall not, directly or indirectly, assign this Agreement or any of its rights hereunder. Any attempted assignment in violation hereof shall, at the election of Sellers, be of no force or effect and shall constitute a default by Purchaser.
  - (c) The obligations of Sellers under this Agreement shall be joint and several.
- 14.8 <u>Waiver</u>. The excuse or waiver of the performance by a Party of any obligation of the other Party under this Agreement shall only be effective if evidenced by a written statement signed by the Party so excusing or waiving. No delay in exercising any right or remedy shall constitute a waiver thereof, and no waiver by Sellers or Purchaser of the breach of any covenant of this Agreement shall be construed as a waiver of any preceding or succeeding breach of the same or any other covenant or condition of this Agreement.
- 14.9 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile or attachment of a PDF file to e-mail shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties transmitted by facsimile or attachment of a PDF file to e-mail shall be deemed to be their original signatures for all purposes.
- 14.10 <u>Attorneys' Fees</u>. In the event of a judicial or administrative proceeding or action by one Party against the other Party with respect to the interpretation of, enforcement of, or any action under this Agreement, the prevailing Party shall be entitled to recover reasonable costs and expenses including reasonable attorneys' fees and expenses, whether at the investigative, pretrial, trial or appellate level. The prevailing Party shall be determined by the court based upon an assessment of which Party's major arguments or position prevailed.
- 14.11 <u>Descriptive Headings; Word Meaning</u>. The descriptive headings of the paragraphs of this Agreement are inserted for convenience only and shall not control or affect the meaning or construction of any provisions of this Agreement. Words such as "herein," "hereinafter," "hereof" and "hereunder" when used in reference to this Agreement, refer to this Agreement as a whole and not merely to a subdivision in which such words appear, unless the context otherwise requires. The singular shall include the plural and the masculine gender shall include the feminine and neuter, and vice versa, unless the context otherwise requires. The word "including" shall not be restrictive and shall be interpreted as if followed by the words "without limitation." Each exhibit referenced herein shall be deemed part of this Agreement and incorporated herein wherever any reference is made thereto. Unless otherwise defined therein, capitalized terms used in the exhibits to this Agreement shall have the meanings given to such terms respectively in the body of this Agreement.
- 14.12 <u>Time of the Essence</u>. **TIME IS OF THE ESSENCE WITH RESPECT TO EACH PROVISION OF THIS AGREEMENT.** Without limiting the foregoing, Purchaser and Sellers hereby confirm their intention and agreement that time shall be of the essence with respect to each and every provision of this Agreement, notwithstanding any subsequent modification or extension of any date or time period that is provided for under this Agreement. The agreement of Purchaser and Sellers that time is of the essence with respect to each and every provision of this Agreement shall not be waived or modified by any conduct of the Parties, and the agreement of Purchaser and Sellers that time is of the essence with respect to each and every provision of this Agreement may only be modified or waived by the express written agreement of Purchaser and Sellers that time shall not be of the essence with respect to a particular date or time period, or any modification or extension thereof, which is provided under this Agreement.
- 14.13 <u>Construction of Contract</u>. This Agreement shall not be construed more strictly against one Party than against the other merely by virtue of the fact that it may have been prepared primarily by counsel for one of the Parties, it being recognized that both Purchaser and Sellers have contributed substantially and materially to the preparation of this Agreement.
- 14.14 Severability. The Parties hereto intend and believe that each provision in this Agreement comports with all applicable Laws. If, however, any provision in this Agreement is found by a court of law to be in violation of any applicable Law or public policy, or if in any other respect such a court declares any such provision to be illegal, invalid, unlawful, void or unenforceable as written, then it is the intent of all Parties hereto that, consistent with and with a view towards preserving the economic and legal arrangements among the Parties hereto as expressed in this Agreement, such provision shall be given force and effect to the fullest possible extent, and that the remainder of this Agreement shall be construed as if such illegal, invalid, unlawful, void, or unenforceable provision were not contained herein, and that the rights, obligations, and interests of the Parties under the remainder of this Agreement shall continue in full force and effect.
- 14.15 <u>No Implied Contract</u>. Neither Sellers nor Purchaser shall have any obligations in connection with the transaction contemplated by this Agreement unless (a) all the Sellers, on the one hand each acting in his or her sole discretion, and Purchaser, on the other hand acting in its sole discretion, elect to execute and deliver this Agreement to the other Party and (b) the Company and LGP, each acting in its sole discretion, elect to execute and deliver the

Purchase and Sale Agreement to the other. No correspondence, course of dealing, or submission of drafts or final versions of this Agreement between Sellers and Purchaser shall be deemed to create any binding obligations in connection with the transaction contemplated hereby, and no contract or obligation on the part of Sellers or Purchaser shall arise unless and until a counterpart of this Agreement is fully executed by all the Sellers and Purchaser. Subject to Section 14.15(b), once so executed and delivered by Sellers and Purchaser, this Agreement shall be binding upon them.

Notices. All notices and consents to be given hereunder shall be in writing and shall be deemed to have been duly given if (a) (i) delivered personally, (ii) mailed (postage prepaid) by certified mail (in this case, notice to be deemed given three days after mailing), or (iii) delivered by a recognized commercial courier to the Party entitled thereto at the address set forth below or such other address as such Party shall have designated by five days' notice to the other; and (b) delivered by email, as set forth below:

If to Sellers, to: James E. Lewis, III

2101 Northside Drive, Suite 104 Panama City, FL 32405 Email: jelfish@comcast.net

With a copy to: Glenn A. Adams, Esq.

Holland & Knight LLP

200 South Orange Avenue, Suite 2600

Orlando, FL 32801

Email: glenn.adams@hklaw.com

If to Purchaser, to: Lehigh Gas Wholesale Services, Inc.

1425 Mountain Drive Bethlehem, PA 18015 Telephone Number:

Email: jtopper@lehighgas.com Attention: Joseph V. Topper, Jr.

With a copy to: Richard A. Silfen, Esq.

Duane Morris LLP 30 South 17th Street Philadelphia, PA 19103

Telephone Number: 215-979-1225 Email: rasilfen@duanemorris.com

[Signatures appear on the next succeeding page.]

IN WITNESS WHEREOF, the Parties hereto have executed this Stock Purchase Agreement as of the date first above written.

WITNESS

/s/ GLENN A. ADAMS	/s/ JAMES E. LEWIS, JR.		
Print Name: GLENN A. ADAMS	JAMES E. LEWIS, JR.		
WITNESS			
/s/ GLENN A. ADAMS	/s/ LIDA N. LEWIS		
Print Name: GLENN A. ADAMS	LIDA N. LEWIS		
WITNESS			
/s/ GLENN A. ADAMS	/s/ JAMES E. LEWIS, III		
Print Name: GLENN A. ADAMS	JAMES E. LEWIS, III		
WITNESS			
/s/ GLENN A. ADAMS	/s/ REID D. LEWIS		
Print Name: GLENN A. ADAMS	REID D. LEWIS		

WITNESS

/s/ DAVID HR	INAK	By:	/s/ JOSEPH V. TOPPER, JR.	
Print Name:	DAVID HRINAK	Name:	JOSEPH V. TOPPER, JR.	
		Title:	President	

#### PURCHASE AND SALE AGREEMENT

This PURCHASE AND SALE AGREEMENT (this "Agreement") is entered into as of the 21<sup>st</sup> day of December, 2012 ("Effective Date") between **EXPRESS LANE, INC.**, a Florida corporation ("Seller"), and **LGP REALTY HOLDINGS LP**, a Delaware limited partnership ("Buyer"), with reference to the following:

#### **RECITALS**

- A. Seller owns certain real properties described on attached **Exhibit "A"**, together with Seller's right, title and interest in and to all appurtenant rights, privileges and servitudes (collectively, the "Parcels of Land" and each a "Parcel of Land"), together with all buildings, structures, or other improvements owned by Seller thereon (the "Improvements") and certain personal property (the "Personal Property") as described in the Bill of Sale attached hereto as **Exhibit "D"** (the Parcels of Land, Improvements, Personal Property, and the Distribution Agreements (as hereinafter defined) being collectively referred to as the "Property").
  - B. Seller agrees to sell and Buyer agrees to buy the Property upon the terms and conditions set forth below.

NOW, THEREFORE, in consideration of the sum of One Dollar (\$1.00), and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

#### 1. PURCHASE AND SALE.

- **1.1** Seller agrees to sell and convey the Property to Buyer, and Buyer agrees to purchase the Property from Seller, subject to the terms and conditions set forth in this Agreement. In addition, at Closing (as defined in Section 3.1 below), Seller shall assign to Buyer all of Seller's right, title and interest in and to those certain agreements listed on **Schedule 1.1** attached hereto (the "Distribution Agreements").
- **1.2** The purchase price for the Property ("Purchase Price") shall be Eighteen Million U.S. Dollars (\$18,000,000), payable on the Closing Date (as defined in Section 3.1 below) by Federal Funds wire transfer of immediately available funds, subject to prorations and adjustments at Closing as provided in this Agreement. The Purchase Price shall be allocated to each portion of the Property as set forth on **Exhibit "B"**.
- 1.3 The Closing shall take place one (1) business day prior to the Closing (as defined in the SPA (as hereinafter defined)) under the SPA. The term "SPA" shall mean that certain Stock Purchase Agreement dated the date hereof among James E. Lewis, Jr., Lida N. Lewis, James E. Lewis, III and Reid D. Lewis (collectively, "Shareholders") and Lehigh Gas Wholesale Services, Inc. ("LGWS").
- **1.4** Capitalized terms not defined elsewhere in this Agreement are defined in Section 10 below.

#### 2. <u>TITLE REVIEW AND SURVEY</u>.

- 2.1 Title to the Property (both legal and beneficial) shall be good, marketable and indefeasible and free and clear of all liens, restrictions, easements, encumbrances and title objections, except for the Permitted Encumbrances (as defined in Section 2.2 below), and title to the Real Properties shall be insurable as such at standard rates by Stewart Title Guaranty Company ("Title Company").
- 2.2 The following items shall be deemed "Permitted Encumbrances": (a) real estate taxes assessed for the Tax Year (as defined below in Section 3.8) in which the Closing takes place and not yet due and payable, which taxes shall be adjusted as provided in this Agreement; (b) existing encroachments, easements, rights of way, covenants, reservations and restrictions in the chain of title to the Real Properties or existing thereon which do not substantially interfere with the use of the Real Properties; and (d) matters of survey which do not substantially interfere with the use of the Real Properties for the retail sale of petroleum products and food stuffs; (c) all building, zoning and historical Laws, rules and regulations affecting the Real Properties; and (d) matters of survey which do not substantially interfere with the use of the Real Properties for the retail sale of petroleum products and food stuffs.
- **2.3** On or prior to the Closing, Seller shall deliver to Stewart Title Guaranty Company, as escrow agent ("Escrow Agent"), all customary and reasonable documentation required by Title Company to issue a 2006 Form ALTA Owner's Title Policy (the "Owner's Title Policy").
- **2.4** The descriptions to be included in the Deeds (as defined in Section 3.3 below), to be delivered from Seller to Buyer at Closing, shall be the **Exhibit "A"** legal descriptions attached to this Agreement.

#### 3. CLOSING, DELIVERIES AND ADJUSTMENTS

- 3.1 The purchase and sale of the Property (the "Closing") shall occur on date and time as designated in writing by Buyer to Seller, but in no event later than December 21, 2012 (the "Outside Date"), but in all events the Closing shall be one (1) day prior to the date of the Closing (as defined in the SPA) under the SPA. The Closing will be coordinated by and held at the offices of Holland & Knight LLP, in Orlando, Florida, and to the extent feasible, the Parties will use Fedex (or similar overnight carrier) or permit the electronic exchange of documents in PDF format or by facsimile (with original documents to follow), without the principals present; it being agreed, however, that all Deeds and other documents to be recorded will be delivered to Escrow Agent on or before the Closing Date (as defined in this Section 3.1). The actual date as of which the Closing occurs is referred to herein as the "Closing Date."
- 3.2 All proceedings to be taken and any documents to be executed and delivered by Seller and Buyer at the Closing shall be deemed to have been taken, executed and delivered simultaneously, and no proceedings shall be deemed taken nor any documents executed or delivered until all have been taken, executed and delivered.

- 3.3 At Closing, Seller shall execute and/or deliver (or cause to be executed and delivered as appropriate) to Escrow Agent:
  - **a.** A special warranty deed in the form attached hereto as **Exhibit "C"** for each of the Real Properties (the "Deeds").
  - **b.** A bill of sale in the form attached hereto as **Exhibit "D"** transferring the Personal Property (the "Bill of Sale").
  - **c.** An Assignment and Assumption of Intangible Rights and Obligations in the form attached hereto as **Exhibit "E"** (the "Assignment and Assumption").
  - **d.** An Assignment and Assumption Agreement in the form attached hereto as **Exhibit "F"** with respect to the Distribution Agreements (the "Distribution Agreements").
  - **e.** A Tax Certification in the form attached as **Exhibit "G"**, pursuant to Internal Revenue Code Section 1445 (the "FIRPTA Certification").
  - **f.** True, correct and complete copies of any plans and specifications and development plans in Seller's possession, and all keys, security and pass codes for each of the Real Properties.
  - **g.** Authorizing resolutions to evidence the authorization of the sale of the Property by Seller and the delivery by Seller of all of the Seller Documents (as defined in Section 4.1(a) below).
  - h. Such standard form affidavits as Title Company may reasonably require in order to issue the Owner's Title Policy as required herein and to insure title to the Real Properties after the Closing; it being agreed that the affidavit for the property located at 8140 West Highway 388, Panama City Beach, Florida shall include the following representation: "[Seller] acknowledges that (a) the Improvements (as defined in Memorandum of Agreement recorded on October 31, 2011 at OR Book 3361, page 1331 of the official records of Bay County, Florida) have been completed in accordance with Section 12 of the Memorandum of Agreement and the underlying Purchase and Sale Agreement, and (b) the repurchase option set forth in section 16.1 of the Memorandum of Agreement and the underlying Purchase and Sale Agreement is null and void."
  - i. <u>Intentionally Omitted.</u>
  - j. A closing statement prepared by Escrow Agent and approved by Seller and Buyer (the "Closing Statement").
- **3.4** At Closing, Buyer shall execute and/or deliver to the Escrow Agent:
  - **a.** The Purchase Price, by wire transfer of immediately available U.S.

Federal Funds, to be received at or before 2:00 p.m., Eastern Time, on the Closing Date in the Escrow Agent's escrow account.

- **b.** The Assignment and Assumption.
- **c.** The Distribution Agreements Assignment.
- **d.** The Closing Statement.
- **3.5** Buyer may, at Buyer's election, direct Seller to assign the Distribution Agreements to Buyer's designee, Lehigh Gas Wholesale LLC, a Delaware limited liability company.
- **3.6** At Closing, Seller shall pay the following costs and expenses:
  - **a.** All fees and costs to release liens (including, but not limited to any existing financing) against the Property.
  - **b.** Any recording fees relating to its obligations to deliver title to the Property as required by Section 2.1 above.
  - **c.** With regard to any existing financing, all prepayment penalties and termination or filing fees for any UCC financing statements.
  - **d.** One-half (1/2) of escrow and/or closing fees of Escrow Agent.
  - **e.** Seller's attorneys' fees and expenses.
- **3.7** At Closing, Buyer shall pay the following costs and expenses:
  - **a.** The Purchase Price.
  - **b.** Any transfer taxes and/or conveyance fees and stamps on the Deeds.
  - **c.** All expenses relating to its due diligence and investigation of the Property.
  - **d.** The premium for the Owner's Title Policy, including any endorsements requested by Buyer.
  - **e.** All premiums and fees for title examination and all related charges.

- **f.** Except as expressly provided in Section 3.6, any recording fees for documentation to be recorded in connection with the transactions contemplated by this Agreement.
- **g.** One-half (1/2) of escrow and/or closing fees of Escrow Agent.
- **h.** All UCC searches.

- i. All fees and costs for any survey.
- **j.** All costs of any financing obtained by Buyer.
- **k.** Buyer's attorneys' fees and expenses.
- As applicable, all real estate taxes, ad valorem personal property taxes, fire district taxes, and other such charges constituting a lien or encumbrance on any of the Property, including special or betterment assessments against any of the Property which are payable in installments over more than one year (collectively "Property Taxes"), shall be adjusted on the basis of the calendar year (regardless of the date of assessment) (the "Tax Year"). All unpaid Property Taxes that relate to Tax Years ending on or before the Closing Date shall be the responsibility of Seller. Property Taxes that relate to the Tax Year in which the Closing occurs shall be prorated between Seller and Buyer, with Seller being responsible for the portion of such Tax Year up to and including the Closing Date and Buyer being responsible for the portion of such Tax Year after the Closing Date. Property Taxes that relate to a Tax Year commencing after the Closing Date shall be the responsibility of the Buyer. All installments of Property Taxes having a due date on or before the Closing Date shall be paid by Seller on or before the Closing Date, and Buyer shall pay all installments of Property Taxes having a due date after the Closing Date. At the Closing, the net amount of all Property Tax adjustments computed according to this Section 3.8, based upon which Party is responsible for such Property Taxes and which Party is to pay such Property Taxes, shall be added to or deducted from the Purchase Price, as the case may be, as shown on the Closing Statement for the applicable Real Property. If the amount of any Property Taxes is not fixed and determined as of the Closing, the foregoing Closing adjustment shall be based on the amount thereof as reasonably estimated at Closing.
- 3.9 Charges for water, gas, power, light and other utility service shall be the responsibility of Seller with respect to service up to the Closing Date and shall be Buyer's responsibility with respect to service on and after the Closing Date. The Parties shall endeavor to obtain meter readings or other evidence of the amounts due for utilities before Closing. If such readings are not obtained (and if such readings are obtained, then with respect to any period between such reading and the Closing Date), water rates and charges, and charges for gas, power, light and other utility services, if any, shall be apportioned based upon the last meter readings, subject to reapportionment when readings for the relevant period are obtained after the Closing Date. The provisions of this Section 3.9 shall survive the Closing.
- **3.10** At Closing, Seller shall deliver full possession of the Real Properties free and clear of all occupants and subject to the Permitted Encumbrances, such Real Properties to be in substantially the same condition as on the Effective Date with the following exceptions: (a) ordinary wear and tear, (b) any damages caused by the acts or omissions of Buyer, its agents or contractors, and (c) subject to the provisions of Sections 6.2 and 6.3, damages by casualty or affected by any taking.

5

**3.11** Not less than one (1) business day prior to Closing, Buyer and Seller shall agree on a Closing Statement prepared by the Escrow Agent for the transactions contemplated by this Agreement in anticipation of Closing.

#### 4. REPRESENTATIONS AND WARRANTIES.

- **4.1 Seller's Representations and Warranties.** Seller represents and warrants to Buyer the following as of the Effective Date and as of the Closing Date as though made on and as of each such date:
  - a. Seller is a corporation duly organized, validly existing, and in good standing under the Laws of its state of incorporation and has the corporate power and authority to own the Property and to enter into and to carry out the terms and conditions of this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by Seller in connection with the consummation of the transactions contemplated by this Agreement (all such other agreements, documents, instruments and certificates required to be executed by Seller being hereinafter referred to, collectively, as the "Seller Documents") and to perform fully its obligations hereunder and thereunder. This Agreement has been, and the Seller Documents will be at or before the Closing, duly executed and delivered by Seller and, assuming the due authorization, execution and delivery by Buyer, this Agreement constitutes, and the Seller Documents when so executed and delivered will constitute, legal, valid and binding obligations of Seller, enforceable against Seller according to their respective terms, except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, or other laws of general application relating to or affecting the enforcement of creditors' rights generally, and (ii) as limited by Laws relating to the availability of specific performance, injunctive relief, or other equitable remedies. None of the execution and delivery by Seller of this Agreement and the Seller Documents, or the consummation of the transactions contemplated hereby or thereby, or compliance by Seller with any of the provisions hereof or thereof, will (x) conflict with, violate, result in the breach or termination of, or constitute a default under, any agreement to which Seller is a party or by which Seller or any of the Property is bound or subject or (y) constitute a violation of any Law applicable to Seller or the Property.
  - **b.** Except as set forth on <u>Schedule 4.1(b)</u>, no consent, waiver, approval, order, permit or authorization of, or declaration or filing with, or notification to, any person or governmental body is required on the part of Seller in connection with the execution and delivery of this Agreement or the Seller Documents or the compliance by Seller with any of the provisions hereof or thereof.

of the transactions contemplated by this Agreement or that questions the validity of this Agreement, the Seller Documents or any action taken or to be taken by Seller in connection with the consummation of the transactions contemplated hereby or thereby.

- **d.** Except as described on attached <u>Schedule 4.1(d)</u>, Seller has not received written notice of, and to the best of Seller's knowledge, there does not exist, any violation of (i) any applicable federal, state or local laws, codes, rules, ordinances or orders, including without limitation those relating to environmental, health, safety, zoning, platting, fire, seismic or other land use requirements, which has not been corrected to the satisfaction of the appropriate governmental authority, or which would affect the use or income of the Property, or (ii) any covenant, condition, restriction, instrument or agreement affecting or relating to the use, occupation and construction of the Property.
- e. Seller has good and marketable title to the Property free and clear of all liens except the Permitted Encumbrances.
- **f.** Except as set forth in Section 6.4, Seller is not a party to any contract or undertaking to pay any broker's, finder's or financial advisor's fee in connection with the origin, negotiation, execution or performance of this Agreement.
- g. Seller is not a "foreign person" as defined in the Foreign Investment in Real Property Tax Act, 26 U.S.C. 1445(f)(3).
- h. Seller holds in its name all applicable Permits required by Law for the operation of Seller's business at the Real Properties, all of which Permits are listed on Schedule 4.1(h). Except as disclosed in Schedule 4.1(h), Seller has not received official notice of any alleged violation of, nor citation for noncompliance with, any Law or Permit (except for minor violations or events of noncompliance which have been cured or remedied) including, without limitation, all applicable Laws relating to employment, employee benefits or the sale of tobacco, and the Americans With Disabilities Act, relating to the Property and, to Seller's Knowledge, except as set forth on Schedule 4.1(h), Seller is in material compliance with all Laws and Permits relating to the Property.
- **i.** Except as set forth in <u>Schedule 4.1(i)</u>, Seller has not received any written notice of any condemnation, expropriation, eminent domain or similar proceeding affecting all or any portion of any Real Property, nor to Seller's Knowledge is there any pending condemnation, expropriation, eminent domain or other similar proceeding affecting all or any portion of any Real Property.
- **j.** All surety or other bonds that Seller maintains in connection with its operation of the Real Properties are listed on <u>Schedule 4.1(j)</u> and no other surety or other bonds will be required by Buyer at the Closing to operate the Real Properties. Seller is not in default under any of such surety or other bonds, and, to

7

the Knowledge of Seller, there has been no occurrence nor exists any condition or circumstance that, with either notice or passage of time or both, would result in or constitute a default by Seller under any of such surety or other bonds.

- **k.** Except as set forth on Schedule 4.1(k), Seller has not received written notice of any physical defects or conditions or violations of applicable law (representations and warranties with respect to environmental matters being set forth in Section 4.1(1) hereof, and being specifically excluded from this Section 4.1(k)) with regard to any of the Real Properties. Seller has made available for review by Buyer complete copies of all surveys, reports or deficiency notices concerning the Real Properties which are in the possession of Seller. There are no Leases affecting any Real Property, and Seller has the exclusive right of possession of each Real Property, subject only to the Permitted Encumbrances. Seller has not received any written notice of default under any Law or Permit which could reasonably be expected to have a material adverse effect on the operation of the Real Properties.
- **l.** Except as disclosed on **Schedule 4.1(!)**, to the Knowledge of Seller:
- (i) no written or oral notice, notification, demand, request for information, citation, summons, complaint or order has been received by Seller, no complaint has been filed, no penalty has been assessed and no investigation or review is pending or, to the Knowledge of Seller, threatened by any governmental entity or other person or entity with respect to any past or present (A) alleged violation by Seller of any Environmental Laws or liability thereunder; (B) alleged failure by Seller to have any Permit required under any Environmental Laws in connection with the conduct of its business at the Real Properties; (C) the Release of Hazardous Substances at the Real Properties; or (D) damages to natural resources;
- (ii) no surface impoundments, pits or lagoons in which Hazardous Substances are being or have been treated, stored, or disposed of have been located on any Real Property since the date of Seller's acquiring such Real Property or, to the Knowledge of the Seller, at any time before, no Hazardous Substances have been found in any potable water (excluding potable water supplied by governmental authorities) used to supply any such Real Property since the date of Seller's acquiring such Real Property or, to the Knowledge of Seller, at any time before, and no polychlorinated biphenyls, radioactive material, urea formaldehyde, lead, asbestos, asbestos containing material or other Hazardous Substances in an amount or concentration so as to violate Environmental Laws is or was deposited or released at, under or on any Real Property since the date of Seller's acquiring such Real Property or, to the knowledge of Seller, at any time before;
  - (iii) there are no Environmental Liabilities;

- (iv) there has been no environmental investigation, study, or audit, conducted of which Seller has possession or control in relation to any Real Property which has not been delivered to Buyer; and
- (v) Seller is in material compliance with all Environmental Laws relative to the construction, maintenance, compliance and use of underground storage tanks and appurtenances thereto at all of the Real Properties.
- **4.2 Buyer's Representations and Warranties.** Buyer represents and warrants to Seller the following as of the Effective Date and as of the Closing Date as though made on and as of each such date:
  - a. Buyer is a limited partnership duly organized, validly existing, and in good standing under the Laws of its state of formation and has the partnership power and authority to enter into and to carry out the terms and conditions of this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by Buyer in connection with the consummation of the transactions contemplated by this Agreement (all such other agreements, documents, instruments and certificates required to be executed by Buyer being hereinafter referred to, collectively, as the "Buyer Documents") and to perform fully its obligations hereunder and thereunder. This Agreement has been, and the Buyer Documents will be at or before the Closing, duly executed and delivered by Buyer and, assuming the due authorization, execution and delivery by Seller, this Agreement constitutes, and the Buyer Documents when so executed and delivered will constitute, legal, valid and binding obligations of Buyer, enforceable against Buyer according to their respective terms, except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, or other laws of general application relating to or affecting the enforcement of creditors' rights generally, and (ii) as limited by Laws relating to the availability of specific performance, injunctive relief, or other equitable remedies. None of the execution and delivery by Buyer of this Agreement and the Buyer Documents, or the consummation of the transactions contemplated hereby or thereby, or compliance by Buyer with any of the provisions hereof or thereof, will (x) conflict with, violate, result in the breach or termination of, or constitute a default under, any agreement to which Buyer is a party or by which Buyer is bound or subject or (y) constitute a violation of any Law applicable to Buyer.
  - **b.** No consent, waiver, approval, order, permit or authorization of, or declaration or filing with, or notification to, any person or governmental body is required on the part of Buyer in connection with the execution and delivery of this Agreement or the Buyer Documents or the compliance by Buyer with any of the provisions hereof or thereof.
  - **c.** There is no legal proceeding pending or, to the knowledge of Buyer, threatened against the Buyer that seeks to enjoin or obtain damages in respect of

the consummation of the transactions contemplated by this Agreement or that questions the validity of this Agreement, the Buyer Documents or any action taken or to be taken by the Buyer in connection with the consummation of the transactions contemplated hereby or thereby.

- **d.** Buyer represents and warrants that Buyer has not acted in a manner that could cause Seller to incur liability to any person for brokerage commissions, finder's fees or other remuneration in connection with the sale of the Property the transactions contemplated by this Agreement, other than the payments by Seller contemplated by Section 6.4 hereof.
- **e.** Buyer and Buyer's Affiliates have the net worth, financial standing, access to required liquidity and the necessary borrowing capacity to complete the transactions contemplated by this Agreement.

#### 5. <u>DUE DILIGENCE AND DISCLAIMER OF WARRANTIES</u>

- 5.1 Buyer acknowledges that Seller has (i) given Buyer, its lenders, and its accounting, legal, business, environmental, engineering, intellectual property and other authorized representatives and advisors full access, during normal business hours, to the Real Properties, and (ii) furnished Buyer and its authorized representatives and advisors with all documents and information relating to the Property as reasonably requested by Buyer, its lenders, and its authorized representatives and advisors, (iii) permitted Buyer and its authorized representatives and advisors to review all books, records of Seller as requested by Buyer and its authorized representatives and advisors, and made copies thereof, and (iv) fully cooperated with Buyer and its authorized representatives and advisors in its investigation and examination of the Property. No investigation, or receipt of information provided by or on behalf of Seller or review thereof by Buyer, its lenders, or its representatives or advisors has diminished or obviated, or relieved Seller from, or affected the ability or right of Buyer to rely on, any of the representations, warranties, covenants and agreements of Seller contained in this Agreement and the Seller Documents. Buyer acknowledges that its obligations to consummate the transactions contemplated hereby are not subject to any further due diligence investigation by Buyer.
- 5.2 Buyer acknowledges that it has received information about the Property (the "Confidential Information"). Buyer agrees to continue to be bound by all agreements under which it received such Confidential Information, including that certain confidentiality agreement entered into before the date hereof between Lehigh Gas Corporation and Matrix (as defined in Section 6.4 below) for the benefit of Seller (the "Confidentiality Agreement"). If this Agreement is terminated before the Closing, Buyer promptly shall return to Seller or destroy all Confidential Information and shall not retain copies thereof, including all information, testing, studies, surveys, reports and evaluations (environmental and financial, for example), and at no cost to Seller Buyer shall provide Seller with complete copies of all third party engineering and environmental reports obtained by Buyer in connection with the Real Properties.

- 5.3 At its sole expense, Buyer shall have the right to perform an Environmental Site Assessment ("ESA") at the Real Properties before the Closing, provided the ESA conforms to Schedule 7.3 of the SPA and applicable ASTM standards, does not damage or interfere with the operation of the Real Properties and Buyer provides Seller with a copy of the ESA report promptly after the report is issued. Buyer shall indemnify and hold Seller harmless for the acts of Buyer and/or Buyer's agents, employees or contractors in performing the ESA. Buyer shall return each Real Property to its previous condition upon completion of its activities in connection with the ESA. Buyer's conduct of an ESA shall not be a condition to the Closing, and failure to perform or complete the ESA shall not extend the date for the Closing.
- 5.4 If Buyer's due diligence reveals any condition at the Real Properties that requires disclosure to any governmental agency or authority, Buyer shall immediately (a) notify Seller thereof, and (b) provide all data and analysis concerning the condition acquired during the due diligence process including without limitation, the ESA. Buyer acknowledges that applicable rules (including particularly Rule 62-770.250, FAC) may require notification to a governmental agency within twenty-four (24) hours of occurrence, and Buyer agrees to take all reasonable steps necessary to support the ability of Seller to meet this obligation. Except as provided by Law, in such event, at all times prior to the Closing, Seller, and not Buyer, Buyer's agents, or anyone acting on Buyer's behalf, shall make such legal determinations regarding disclosures as Seller deems appropriate.
- Seller's representations in this Agreement and the Seller Documents, and subject to Seller's obligations as contained in this Agreement, Buyer is not relying on any representations or warranties of any kind whatsoever, express or implied, from Seller, Matrix, Shareholders, or any director, officer, employee, representative, broker or other agent of Seller or Matrix, as to any matters concerning the Property, including: (a) the condition or safety of the Real Properties, including plumbing, sewer, heating and electrical systems, roofing, air conditioning, if any, foundations, soils and geology, lot size, or suitability of the Real Properties for a particular purpose; (b) whether the appliances, if any, plumbing or utilities are in working order; (c) the habitability or suitability for occupancy of any structure and the quality of its construction; (d) the fitness or condition of any Personal Property; (e) whether the Improvements, including the Fuel Equipment, are structurally sound, in good condition, or in compliance with applicable Laws; (f) the profits or losses relating to operations at the Real Properties; (g) the legal or tax consequences of this Agreement or the transactions contemplated hereby; (h) the environmental condition of the Real Properties, including but not limited to the possible presence of petroleum products and/or Hazardous Substances in, under or near the Real Properties; and (i) the completeness or accuracy of any information provided to Buyer by Seller or Seller's agents. Buyer understands the legal significance of the foregoing provisions and acknowledges that they are a material inducement to Seller's willingness to enter into this Agreement.
- 5.6 BUYER ACKNOWLEDGES THAT BUYER HAS BEEN GIVEN THE

OPPORTUNITY TO EXAMINE ALL ASPECTS OF THE PROPERTY AND TO REVIEW ALL FILES CONCERNING THE PROPERTY MAINTAINED BY SELLER AND ITS ENVIRONMENTAL CONSULTANTS AND ALL STATE AGENCIES HAVING JURISDICTION OVER THE REAL PROPERTIES BEFORE BUYER'S EXECUTION AND DELIVERY OF THIS AGREEMENT. ACCORDINGLY, BUYER AGREES THAT THE PROPERTY SHALL BE SOLD AND THAT BUYER SHALL ACCEPT THE PROPERTY AT CLOSING STRICTLY ON AN "AS IS, WHERE IS, WITH ALL FAULTS" BASIS, WITH NO RIGHT OF SET-OFF OR REDUCTION IN THE PURCHASE PRICE EXCEPT FOR ADJUSTMENTS EXPRESSLY CONTEMPLATED BY THIS AGREEMENT AND THAT, EXCEPT FOR SELLER'S REPRESENTATIONS SET FORTH IN THIS AGREEMENT AND THE SELLER DOCUMENTS, THE SALE OF THE PROPERTY AND OTHER TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT SHALL BE WITHOUT REPRESENTATION OR WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF INCOME POTENTIAL, OPERATING EXPENSES, USES, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND SELLER DOES HEREBY DISCLAIM AND RENOUNCE ANY SUCH REPRESENTATION OR WARRANTY.

5.7 RADON IS A NATURALLY OCCURRING RADIOACTIVE GAS THAT, WHEN IT HAS ACCUMULATED IN A BUILDING IN SUFFICIENT QUANTITIES, MAY PRESENT HEALTH RISKS TO PERSONS WHO ARE EXPOSED TO IT OVER TIME. LEVELS OF RADON THAT EXCEED FEDERAL AND STATE GUIDELINES HAVE BEEN FOUND IN BUILDINGS IN FLORIDA. ADDITIONAL INFORMATION REGARDING RADON AND RADON TESTING MAY BE OBTAINED FROM YOUR COUNTY PUBLIC HEALTH DEPARTMENT.

#### 6. COVENANTS

- 6.1 Before Closing, neither Seller nor Buyer shall make any public announcements concerning the execution and delivery of this Agreement or the transactions contemplated hereby without first obtaining the prior written consent of the other, except as may be required by applicable Law or agreement with any securities exchange, and except that (a) Seller may disclose this Agreement to any lender holding a lien on any Real Property in order to comply with Seller's obligations hereunder, (b) Buyer may disclose this Agreement to any lender providing financing to Buyer, and (c) Seller or Buyer may disclose the identity of the other Party and terms of this Agreement to their employees, vendors, outside counsel or other outside professionals on a need to know basis in connection with consummating this transaction. If a disclosure is required by applicable Law or agreement with any securities exchange, the disclosing Party shall make reasonable efforts to afford the other Party an opportunity to review and comment on the proposed disclosure before the making of such disclosure.
- 6.2 Seller shall keep the Property insured, at Seller's expense, in such amounts as presently insured through the Closing. On any material damage to or destruction of the Real Properties or any portion thereof before Closing, the Closing shall, at Seller's sole

12

discretion, nevertheless proceed; provided, however, that unless before Closing the same shall have been restored by Seller to its condition as of the Effective Date, at the Closing, Buyer shall be entitled to the benefit of any insurance proceeds due Seller as a result of such damage or destruction and Buyer shall assume responsibility for such repair and receive a credit against the Purchase Price for any applicable policy deductible or uninsured damage up to a maximum credit equal to the difference between the amount of the Purchase Price and the amount of the insurance proceeds payable to Seller.

- **6.3** If before Closing, any part of the Real Properties is taken, or noticed for taking, by eminent domain, Seller shall promptly give Buyer written notice thereof and the Closing shall at Seller's sole discretion nevertheless proceed; provided, that such provision shall not apply with respect to any such taking more than two (2) individual Real Properties other than the Real Properties listed on **Schedule 4.1(i)**.
- 6.4 Buyer acknowledges that Seller and/or Shareholders have retained Matrix Private Equities, Inc. ("Matrix") to act as the financial advisor in connection with the transactions contemplated by this Agreement. Shareholders shall be responsible for any fees due to Matrix. Each Party shall be responsible for any brokerage commission or other fees legally determined to be due to any broker or advisor claiming by, through or under such Party and, as between Seller and Buyer, shall also be responsible for any loss, liability, damage, cost or expense (including, without limitation, reasonable attorneys' fees) paid or incurred by the other Party by reason of any claim to any such broker's, finder's or other fee in connection with the transactions contemplated by this Agreement by any such broker or advisor.

#### 6.5 Environmental.

a. If an ESA conducted by Buyer (or any ESA conducted by Seller) prior to the Closing reveals releases of petroleum at the Real Properties which are required to be reported to the FDEP, and which have not been previously reported to FDEP, then Buyer shall advise Seller, and Seller shall promptly, but in all cases prior to Closing, report the same to FDEP. Seller shall thereafter undertake Remedial Measures (as described below) for such releases to the extent required by Law at the Real Properties consistent with a commercial use standard. The Parties acknowledge that Seller is currently undertaking Remedial Measures as set forth on Schedule 4.1(1). Subject to the last sentence of this Section 6.5(a), Seller agrees to be financially responsible for the cost of any Remedial Measures to the extent such Remedial Measures are not covered by insurance or do not qualify for reimbursement from trust funds from the State of Florida or other applicable governmental agency, if such Remedial Measures are specifically listed on Schedule 4.1(1) or are related to releases identified in the ESA prior to Closing. Buyer must request payment for actual costs of the Remedial Measures incurred by Buyer or Seller by giving written notice under the Environmental Escrow Agreement (as defined below) or otherwise to Shareholder Representative as follows: (i) on or before December 31, 2014, for Real Properties listed as "Inactive" on Schedule 4.1(1), or (ii) on or before December 31, 2015, for those

13

Real Properties listed as "Active" on Schedule 4.1(1) or identified in the ESA prior to Closing. Otherwise, Seller's financial responsibility under this Section 6.5(a) shall terminate. (For purposes of clarity, Seller and Shareholders shall have no obligation with respect to Remedial Measures at the Real Properties for any releases of petroleum that are first discovered after the Closing.) "Remedial Measures" shall consist of the lowest cost activities reasonably required by applicable Law and by FDEP or other applicable governmental agency to investigate, assess, remediate and/or monitor such releases to achieve a commercial use standard consistent with all applicable Law and regulatory requirements, including but not limited to the use of monitored natural attenuation and the implementation of institutional and engineering controls. "Remedial Measures" do not include activities that exceed what is required by Law, FDEP or other applicable governmental agencies for the existing uses of the Real Properties. Seller and Buyer shall cooperate with each other in the performance of the Remedial Measures, including execution of documents that may be necessary to obtain reimbursement. Shareholders shall have the right to review and approve any financial expenditures related to the Remedial Measures for which Seller has financial responsibility hereunder.

- b. Seller's obligation to conduct Remedial Measures for the Real Properties shall be deemed complete for purposes of this Agreement upon receipt by Seller of a written letter, order or statement from FDEP or other appropriate governmental authority stating that, subject to the agency's usual and customary reservation of rights, it will not require Seller to perform further Remedial Measures with respect to the release ("Closure Letter"), other than routine monitoring. If requested to do so by FDEP or other appropriate governmental authority in connection with the performance of the Remedial Measures or the issuance of the Closure Letter, the Parties acknowledge that Seller may be asked to place reasonable activity and use limitations ("AULs") on the Real Properties, including, if requested, prohibiting the use of groundwater under the Real Properties for potable purposes and/or restricting the use of the Real Properties to non-residential purposes, and Buyer shall join in and consent to same. Seller shall promptly provide a copy of the Closure Letter to Buyer and Shareholders upon receipt.
- c. If the Remedial Measures required under Section 6.5(a) have not been completed at the time of Closing, Seller agrees to cause Shareholders to deposit One Hundred Thirty-One Thousand Five Hundred U.S. Dollars (\$131,500) at Closing (as defined in the SPA) with SPA Escrow Agent pursuant to the terms of an environmental escrow agreement substantially in the form attached as Exhibit B to the SPA (the "Environmental Escrow Agreement") to be applied against the cost of such Remedial Measures.
- **d.** Except to the extent set forth in Section 4.1(l) and this Section 6.5, Seller shall have no other obligation to Buyer, and Buyer hereby releases and discharges Seller, for any liabilities, costs, expenses, losses, damages, or claims, whether

14

arising in law or equity, known or unknown, or arising before or after the Closing, in connection with or relating to the environment at, on, under or near the Real Properties, including but not limited to any such matters relating to the Fuel Equipment or releases of petroleum products or Hazardous Substances.

- **e.** For purposes of clarity, after Closing, if Seller or Buyer excavates or removes any soil on or from the Real Properties other than in connection with taking Remedial Measures for which Shareholders are financially responsible, Buyer shall be responsible for all costs and expenses related to such excavation or removal (including, without limitation, all costs of disposal of such soil). Furthermore, Buyer or Seller, as the case may be, shall be the owner and generator of such soil. Shareholders shall not be responsible for increased remediation costs or expenses as a result of such excavation or removal.
- **6.6** Except for actions contemplated by this Section 6 or taken with the prior written consent of Buyer, from the Effective Date until the Closing Date, Seller shall conduct its business in the ordinary course consistent in all material respects with past practice, and shall:
  - a. maintain the Real Properties and the other Property in good operating condition and repair in all material respects;

- **b.** meet its contractual obligations in all material respects and perform and pay its obligations as they mature in the ordinary course of business;
- c. comply in all material respects with all judgments and Laws promulgated by any governmental body and all Permits applicable to the conduct of its business at the Real Properties or the ownership or operation of the Property, and maintain in all material respects, and prosecute applications for, such Permits;
- **d.** promptly advise Buyer in writing of any material adverse change to the Property or the conduct, business, operations, properties, condition (financial or otherwise) or prospects of its business at the Real Properties; and
- e. give notice to Buyer promptly upon becoming aware of any inaccuracy in any material respect of any of the representations or warranties made herein by Sellers or in the Schedules attached hereto or of any event or state of facts that would result in any such representation or warranty being inaccurate in any material respect at the time of such event or state of facts as if made at and as of such time (any such notice to describe such inaccuracy, event or state of facts in reasonable detail); provided, however, (i) for purposes of determining whether the closing conditions described in Section 7.2 have been satisfied, no such supplemental information shall be deemed to avoid or cure any misrepresentation or breach of warranty or constitute an amendment of any representation, warranty or statement in this Agreement, and (ii) unless expressly consented to in writing by Buyer, no such supplemental information shall be deemed to avoid or cure any

misrepresentation or breach of warranty for purposes of the indemnity provisions of Section 8.

- 6.7 <u>Contractual Obligations</u>. Buyer and Seller hereby agree that the responsibility for any fees or expenses associated with any consents sought under <u>Schedule 4.1(b)</u>, including without limitation assignment fees, transfer fees and any required costs of store refurbishment shall be borne by the party that is responsible for paying such fees in the applicable third party agreement. Notwithstanding the foregoing, Buyer agrees that under no circumstances will Seller, its shareholders, officers, directors or assigns be responsible for any "debranding", "contingent repayment" or similar obligations under or related to the Exxon or Chevron agreements set forth on <u>Schedule 4.1(b)</u>.
- 6.8 Operation of Business. Buyer hereby agrees that Seller shall have the right to continue to operate the Business (as defined in the SPA) in the ordinary course on the Real Property, including exercising its rights under the Distribution Agreements, from the Closing Date through the Closing (as defined in the SPA) under the SPA. The provisions of this Section 6.8 shall survive Closing until the Closing (as defined in the SPA).

#### 7. <u>CONDITIONS PRECEDENT TO CLOSING</u>

- 7.1 The obligations of Seller to consummate the transactions contemplated by this Agreement are subject to each of the following conditions:
  - **a.** The representations and warranties made by Buyer in this Agreement shall be true in all material respects when made and on and as of the Closing as though such representations and warranties were made on and as of the Closing. Seller shall have received from Buyer at Closing a satisfactory certificate to such effect signed by an authorized officer of Buyer.
  - **b.** Buyer shall have performed and complied in all material respects with all provisions of this Agreement required to be performed or complied with by Buyer before or at Closing. Seller shall have received from Buyer at the Closing a satisfactory certificate to such effect, signed by an authorized officer of Buyer.
  - **c.** Buyer shall have executed and delivered to Seller at the Closing each of the Buyer Documents and such additional documents as may be reasonably requested by Seller or the Title Company in order to consummate the transactions contemplated by this Agreement.
  - **d.** Buyer shall have paid or made provisions acceptable to Seller for the payment of all fees, costs and expenses for obtaining all environmental due diligence, surveys, title examinations, inventory audits, and other inspections performed in connection with the transactions contemplated pursuant to this Agreement.
  - e. The SPA shall have not been terminated and LGWS is not in default under the terms of the SPA.

16

- 7.2 The obligations of Buyer to consummate the transactions contemplated by this Agreement are subject to each of the following conditions:
  - **a.** The representations and warranties made by Seller in this Agreement shall be true in all material respects when made and on and as of the date of Closing as though such representations and warranties were made on and as of Closing. Buyer shall have received from Seller at Closing a satisfactory certificate to such effect signed by an authorized officer of Seller.
  - **b.** Seller shall have performed and complied in all material respects with all provisions of this Agreement required to be performed or complied with by Seller before or at Closing. Buyer shall have received from Seller at Closing a satisfactory certificate to such effect signed by an authorized officer of Seller.
  - **c.** Seller shall have executed and delivered to Buyer at the Closing each of the Seller Documents and such additional documents as may be reasonably requested by Buyer or the Title Company in order to consummate the transactions contemplated by this Agreement.
  - d. The SPA shall have not been terminated and Shareholders are not in default under the terms of the SPA.

7.3 The obligations of each of Seller and Buyer to consummate the transactions contemplated by this Agreement are subject to each of the following condition: as of the Closing Date, there shall not be any claim or judgment of any nature or type threatened, pending or made by or before any governmental authority that questions or challenges the lawfulness of the transactions contemplated by this Agreement under any law or regulation or seeks to delay, restrain or prevent such transactions.

#### 8. <u>SURVIVAL; INDEMNIFICATION</u>

- **8.1** Subject to the limitations and other provisions of this Agreement, the representations and warranties of the parties contained herein shall survive the Closing Date and shall remain in full force and effect until March 31, 2014, except with respect to the representations and warranties contained in Sections 4.1(a) and 4.2(a), which shall survive until three (3) months after the expiration of the applicable statute of limitations (including any waivers or extensions thereof).
- 8.2 Subject to the terms and conditions of Article X of the SPA, Shareholders have, jointly and severally (reserving unto each other any rights of contribution), agreed to indemnify and hold Seller, Buyer and their respective Affiliates harmless from and against all damages to and liabilities of Seller and Buyer, as the case may be, (including those resulting from or relating to demands, claims, actions or causes of action, assessments or other losses, costs and expenses relating thereto, interest and penalties thereon and reasonable attorneys' fees and related disbursements and other expenses in respect thereof) by reason of or resulting from (i) a breach of any representation or warranty of Seller contained in or made pursuant to this Agreement or the Seller

17

Documents, or (ii) the failure of Seller duly to perform or observe any term, provision, covenant or agreement to be performed or observed by them pursuant to this Agreement or the Seller Documents.

- **8.3** Subject to the terms and conditions of this Section 8, Buyer hereby agrees to indemnify, defend and hold Seller harmless from and against all damages to and liabilities of Seller (including those resulting from or relating to demands, claims, actions or causes of action, assessments or other losses, costs and expenses relating thereto, interest and penalties thereon and reasonable attorneys' fees and related disbursements and other expenses in respect thereof) by reason of or resulting from (i) a breach of any representation or warranty of Buyer contained in or made pursuant to this Agreement or the Buyer Documents, or (ii) any failure of Buyer duly to perform or observe any term, provision, covenant or agreement to be performed or observed by Buyer pursuant to this Agreement or the Buyer Documents.
- **8.4** The Parties hereby acknowledge and agree that their sole and exclusive remedy with respect to any and all claims relating to the subject matter of this Agreement (other than a claim for fraud or for specific performance of the terms of this Agreement) shall be pursuant to the indemnification provisions set forth in this Section 8 and Article X of the SPA.
- **8.5** The Parties shall take all reasonable steps to mitigate all liabilities and damages upon and after becoming aware of any event that could reasonably be expected to give rise to such liabilities and damages. In no event shall any Party be liable for consequential, incidental, exemplary or punitive damages other than such damages awarded to a third party.
- 8.6 If any claim, assertion or proceeding by or in respect of a third party is made against an indemnified party or any event in respect of a third party occurs, and if the indemnified party intends to seek indemnity with respect thereto under this Section 8 (or in the case of Buyer being the indemnified party, under Article X of the SPA) to apply any damage or liability arising therefrom to the U.S. Dollar amounts referred to in Section 10.5 of the SPA, the indemnified party shall promptly notify the indemnifying party of such claim in writing, provided that the failure by the indemnified party to give such notice shall not relieve the indemnifying party from its indemnification obligations hereunder, except if and to the extent that the indemnifying party is actually materially prejudiced thereby. The indemnifying party shall have thirty (30) days after receipt of such notice to undertake, conduct and control, through counsel of its own choosing and at its expense, the settlement or defense thereof, and the indemnified party shall cooperate with it in connection therewith; provided, however, that (a) the indemnifying party shall permit the indemnified party to participate in such settlement or defense through counsel chosen by the indemnified party, provided that the fees and expenses of such counsel shall be borne by the indemnified party, (b) the indemnifying party shall promptly reimburse the indemnified party for the full amount of any liability resulting from such claim and all related and reasonable expenses (other than the fees and expenses of counsel as aforesaid) incurred by the indemnified party within the limits of this Section 8

18

and subject to the U.S. Dollar amounts referred to in Section 10.5 of the SPA, (c) the indemnified party shall not, without the prior written consent of the indemnifying party, settle or compromise any claim or consent to the entry of any judgment which does not include as an unconditional term thereof the giving by the claimant or the plaintiff to the indemnified party a release from all liability in respect of such claim and (d) nothing herein shall require any indemnified party to consent to the entry of any order, injunction or consent decree materially affecting its ability to conduct its business operations after the date thereof. So long as the indemnifying party is reasonably contesting any such claim in good faith, the indemnified party shall have the right to pay or settle any such claim; provided, however, that in such event it shall waive any right to indemnity therefor by the indemnifying party. If representation of the indemnified party, on the one hand, and the indemnifying party, on the other, by the same counsel would otherwise be inappropriate due to actual or potential differing interests between them, then the indemnified parties shall be entitled to engage separate legal counsel to participate in the defense of such claim at the sole expense of indemnifying party. If the indemnifying party does not notify the indemnified party within thirty (30) days after the receipt of the indemnified party's written notice of a claim of indemnity hereunder that it elects to undertake the defense thereof, the indemnified party shall have the right to contest, settle or compromise the claim in the exercise of its reasonable judgment at the expense of the indemnifying party.

8.7 Except with respect to the representations and warranties contained in Section 4.1(a), or Section 4.2(a), no claim for indemnification will be made by Buyer or by Seller under Section 8.2 or 8.3 hereof, as the case may be, with respect to any individual item of liability or damage unless and to the extent that the aggregate of all claims by Buyer and LGWS under, respectively, this Agreement and the SPA, on the one hand, or by Shareholders, on the other hand, shall be in excess of \$200,000, whereupon Buyer or Shareholders, as the case may be, shall be liable for all such claims, damages and liabilities, and the maximum aggregate liability of the Shareholders, on the one hand, and Buyer and LGWS, on the other hand

for such claims, damages and liabilities shall be the Escrow Amount. Payments by an indemnifying party pursuant to Section 8.2 or 8.3, as applicable, shall be limited to the amount of any liability or damage that remains after deducting therefrom any insurance proceeds and any indemnity, contribution or other similar payment actually recovered by the indemnified party from any third party with respect thereto. Notwithstanding anything to the contrary contained in this Agreement, no claim by any Party hereto may be asserted, nor may any action be commenced against Buyer or the Shareholders, for breach of any representation, warranty, covenant or agreement unless notice thereof is received in writing describing in reasonable detail the facts or circumstances with respect to the subject matter of such claim on or before the date on which the representation, warranty, covenant or agreement on which such claim or action is based ceases to survive as set forth in Section 8.1, irrespective of whether the subject matter of such claim or action shall have occurred before, on or after such date.

#### 9. TERMINATION

**9.1** In the event that the closing of the transactions contemplated herein has not

19

occurred on or before December 21, 2012, either Seller or Buyer shall have the right, exercisable by written notice to the other delivered by telecopier and overnight delivery, to terminate this Agreement. Upon the termination of this Agreement neither Party shall have any further obligations hereunder.

- 9.2 Notwithstanding anything herein to the contrary, in the event that this Agreement has not prior to such date been terminated pursuant to its terms or by mutual consent of the parties, and the Closing has not occurred on or before December 21, 2012 as a result of Buyer's non-performance, default or breach of this Agreement, Buyer and Seller agree that based upon the circumstances now existing, known and unknown, it would be impractical or extremely difficult to establish Seller's damages by reason of Buyer's non-performance, default or breach of this Agreement prior to Closing. Accordingly, Buyer and Seller agree that it would be reasonable in the event of such non-performance, default or breach by Buyer to pay to Seller as "liquidated damages" the Deposit. Seller and Buyer acknowledge and agree that the amount of the Deposit is reasonable as liquidated damages and shall be Seller's sole and exclusive remedy in lieu of any other relief, right or remedy, at law or in equity, to which Seller might otherwise be entitled by reason of such non-performance, default or breach by Buyer.
- 9.3 Notwithstanding anything herein to the contrary, in the event that this Agreement has not prior to such date been terminated pursuant to its terms or by mutual consent of the parties, and the Closing has not occurred on or before December 21, 2012 as a result of Seller's non-performance, default or breach of this Agreement, Buyer may, at its option and as its sole and exclusive remedy, either (a) terminate this Agreement in which event the Deposit shall immediately be returned to LGWS under the SPA or (b) bring an action to enforce specific performance of this Agreement (including the provisions of Section 8) according to the terms hereof by all means available at law or in equity.
- 9.4 BUYER AND SELLER HEREBY ACKNOWLEDGE AND AGREE THAT THE AMOUNT OF SELLER'S DAMAGES IN THE EVENT OF A BREACH OF THIS AGREEMENT BY BUYER WOULD BE DIFFICULT OR IMPOSSIBLE TO DETERMINE AND THAT AN AMOUNT EQUAL TO THE DEPOSIT IS THE PARTIES' BEST AND MOST ACCURATE ESTIMATE OF THE DAMAGES SELLER WOULD SUFFER IN THE EVENT THE TRANSACTION PROVIDED FOR IN THIS AGREEMENT FAILS TO CLOSE AS A RESULT OF THE BREACH OR DEFAULT BY BUYER AND THAT SUCH ESTIMATE IS REASONABLE UNDER THE CIRCUMSTANCES EXISTING AS OF THE DATE OF THIS AGREEMENT AND UNDER THE CIRCUMSTANCES THAT SELLER AND BUYER REASONABLY ANTICIPATE WOULD EXIST AT THE TIME OF SUCH BREACH. BUYER AND SELLER HEREBY ACKNOWLEDGE THAT PURSUANT TO THE ESCROW AGREEMENT, ON SUCH A BREACH BY BUYER, SPA ESCROW AGENT SHALL DELIVER THE DEPOSIT TO SELLER, WHO SHALL APPLY THE DEPOSIT AGAINST THE LIQUIDATED DAMAGES.

20

#### 10. <u>DEFINITIONS</u>

The following capitalized terms used in this Agreement shall have the meanings set forth below:

"Affiliate" means, with respect to any specified Person, any other Person who, directly or indirectly, controls, is controlled by, or is under common control with such Person.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.

"Deposit" shall have the meaning set forth in the SPA.

"Environmental Laws" means all (i) Laws relating to protection of the environment, natural resources or human and occupational health and safety, including Laws relating to Hazardous Substances, or Laws otherwise relating to the manufacture, formulation, generation, processing, distribution, use, treatment, storage, release, transport, remediation, abatement, cleanup or handling of Hazardous Substances, (ii) Laws with regard to recordkeeping, notification, disclosure and reporting requirements respecting Hazardous Substances, and (iii) Laws relating to the condition, management or use of natural resources.

"Environmental Liabilities" means all known liabilities of Seller arising in connection with the Real Properties, which arise under or relate to a violation by Seller of Environmental Laws.

"Escrow Amount" shall have the meaning set forth in the SPA.

"Escrow Agreement" shall have the meaning set forth in the SPA.

"FDEP" means the Florida Department of Environmental Protection.

"Fuel Equipment" means all fuel fixtures and equipment now attached to or used in connection with the Real Properties, including without limitation, all petroleum pumps and dispensers, underground and aboveground fuel storage tanks, canopies, fuel lines, fittings and connections used in the Ordinary Course of Business at the Real Properties to receive, store and/or dispense fuels.

"Hazardous Substances" means (a) any and all radioactive materials or wastes, petroleum (including crude oil or any fraction thereof) or petroleum distillates, asbestos or asbestos containing materials, urea formaldehyde foam, and (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "contaminants," "toxic pollutants," or words of similar meaning and regulatory effect under any applicable Environmental Law.

21

"Knowledge" means, with respect to Seller, actual knowledge of James E. Lewis, Jr., James E. Lewis, III and Reid D. Lewis after reasonable inquiry of employees of Seller who are responsible for such area of inquiry.

"Law" or "Laws" shall mean all applicable local, state, or federal laws, statutes, ordinances, or administrative or judicial decisions.

"Leases" shall mean all unexpired leases, subleases, occupancy agreements, and any other agreements, including all modifications or amendments thereto, for the use, possession, or occupancy of any portion of any Real Property.

"Ordinary Course of Business" means the ordinary course of business consistent with past custom and practice (including with respect to quantity and frequency).

"Parties" means Buyer and Seller.

"Party" means either the Buyer, on the one hand, or the Seller, on the other hand.

"Permits" means all operating permits, Fuel Equipment permits or tank permits, underground and aboveground storage tank notifications or registrations, if any, and other permits, licenses, filings and other governmental authorizations, agreements, contracts, and approvals.

"Person" means any individual, partnership, limited liability company, corporation, cooperative, trust or other entity.

"Real Properties" means all of the Parcels of Land and the Improvements thereon.

"Real Property" means any Parcel of Land and the Improvements thereon.

"Release" has the meaning specified in 42 U.S.C. §9601(22).

"Shareholder Representative" means James E. Lewis, III.

"SPA Escrow Agent" shall mean First American Title Insurance Company.

#### 11. MISCELLANEOUS.

- **11.1** Except as otherwise provided in this Agreement, Buyer and Seller shall each bear their own costs and expenses, including attorneys' fees, incurred in connection with the transactions contemplated by this Agreement.
- 11.2 This Agreement, including the exhibits and other writings referred to herein or delivered pursuant hereto, together with the SPA, constitutes the entire agreement between Seller and Buyer with respect to the subject matter hereof and supersedes all prior oral or written agreements, commitments or understandings with respect thereto, except the Confidentiality Agreement. No amendment hereof shall be binding on the

22

Parties unless in writing and signed by authorized representatives of all Parties hereto.

- **11.3** Except for Shareholders, this Agreement is not intended to and does not confer any rights or obligations on any party that is not a signatory to this Agreement.
- 11.4 If the day for performance of any action described in this Agreement shall fall on a Saturday, Sunday or a day on which the banks are closed in the State of Florida the time for such action shall be extended to the next business day after such Saturday, Sunday or day on which the banks are closed.
- 11.5 This Agreement shall be deemed to be a contract entered into in the State of Florida and it and all matters arising out of the transactions contemplated hereby or related thereto shall be governed, construed and interpreted in all respects according to the Laws of the State of Florida, without reference to principles of conflicts of law thereof.
- 11.6 Any action to enforce this Agreement, or to interpret or construe the meaning of this Agreement, shall be brought in the Circuit Court for Bay County, Florida or for matters that require Federal Jurisdiction, the Federal District Court for the Northern District of Florida, Panama City Division, and Buyer hereby expressly waives venue in any other Federal or State court other than in the State of Florida.
- 11.7 Subject to following provisions, this Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and assigns. Without the prior written consent of Seller and in Seller's sole discretion, Buyer shall not, directly or indirectly, assign this

Agreement or any of its rights hereunder. Any attempted assignment in violation hereof shall, at the election of Seller, be of no force or effect and shall constitute a default by Buyer. Notwithstanding the foregoing, prior to Closing, Buyer may without the consent of Seller, partially assign this Agreement to one or more Affiliates of Buyer that Buyer forms to take title to each Real Property and the Distribution Agreements. No assignment by Buyer shall reliever Buyer of Buyer's obligations under this Agreement.

- 11.8 The excuse or waiver of the performance by a Party of any obligation of the other Party under this Agreement shall only be effective if evidenced by a written statement signed by the Party so excusing or waiving. No delay in exercising any right or remedy shall constitute a waiver thereof, and no waiver by Seller or Buyer of the breach of any covenant of this Agreement shall be construed as a waiver of any preceding or succeeding breach of the same or any other covenant or condition of this Agreement.
- 11.9 This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile or attachment of a PDF file to e-mail shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of

23

the Parties transmitted by facsimile or attachment of a PDF file to e-mail shall be deemed to be their original signatures for all purposes.

- **11.10** In the event of a judicial or administrative proceeding or action by one Party against the other Party with respect to the interpretation of, enforcement of, or any action under this Agreement, the prevailing Party shall be entitled to recover reasonable costs and expenses including reasonable attorneys' fees and expenses, whether at the investigative, pretrial, trial or appellate level. The prevailing Party shall be determined by the court based upon an assessment of which Party's major arguments or position prevailed.
- 11.11 The descriptive headings of the paragraphs of this Agreement are inserted for convenience only and shall not control or affect the meaning or construction of any provisions of this Agreement. Words such as "herein," "hereinafter," "hereof" and "hereunder" when used in reference to this Agreement, refer to this Agreement as a whole and not merely to a subdivision in which such words appear, unless the context otherwise requires. The singular shall include the plural and the masculine gender shall include the feminine and neuter, and vice versa, unless the context otherwise requires. The word "including" shall not be restrictive and shall be interpreted as if followed by the words "without limitation." Each exhibit referenced herein shall be deemed part of this Agreement and incorporated herein wherever any reference is made thereto. Unless otherwise defined therein, capitalized terms used in the exhibits to this Agreement shall have the meanings given to such terms respectively in the body of this Agreement.
- 11.12 TIME IS OF THE ESSENCE WITH RESPECT TO EACH PROVISION OF THIS AGREEMENT. Without limiting the foregoing, Buyer and Seller hereby confirm their intention and agreement that time shall be of the essence with respect to each and every provision of this Agreement, notwithstanding any subsequent modification or extension of any date or time period that is provided for under this Agreement. The agreement of Buyer and Seller that time is of the essence with respect to each and every provision of this Agreement shall not be waived or modified by any conduct of the Parties, and the agreement of Buyer and Seller that time is of the essence with respect to each and every provision of this Agreement may only be modified or waived by the express written agreement of Buyer and Seller that time shall not be of the essence with respect to a particular date or time period, or any modification or extension thereof, which is provided under this Agreement.
- 11.13 This Agreement shall not be construed more strictly against one Party than against the other merely by virtue of the fact that it may have been prepared primarily by counsel for one of the Parties, it being recognized that both Buyer and Seller have contributed substantially and materially to the preparation of this Agreement.
- **11.14** The Parties hereto intend and believe that each provision in this Agreement comports with all applicable Laws. If, however, any provision in this Agreement is found by a court of law to be in violation of any applicable Law or public policy, or if in any other respect such a court declares any such provision to be illegal, invalid, unlawful,

24

void or unenforceable as written, then it is the intent of all Parties hereto that, consistent with and with a view towards preserving the economic and legal arrangements among the Parties hereto as expressed in this Agreement, such provision shall be given force and effect to the fullest possible extent, and that the remainder of this Agreement shall be construed as if such illegal, invalid, unlawful, void, or unenforceable provision were not contained herein, and that the rights, obligations, and interests of the Parties under the remainder of this Agreement shall continue in full force and effect.

- 11.15 Neither Seller nor Buyer shall have any obligations in connection with the transaction contemplated by this Agreement unless (a) Seller and Buyer, each acting in its sole discretion, elect to execute and deliver this Agreement to the other Party, and (b) Shareholders and LGWS, each acting in their or its sole discretion, elect to execute and deliver the SPA to the other party thereto. No correspondence, course of dealing, or submission of drafts or final versions of this Agreement between Seller and Buyer shall be deemed to create any binding obligations in connection with the transaction contemplated hereby, and no contract or obligation on the part of Seller or Buyer shall arise unless and until a counterpart of this Agreement is fully executed by all the Seller and Buyer. Subject to Section 11.15(b), once so executed and delivered by Seller and Buyer, this Agreement shall be binding upon them.
- 11.16 All notices and consents to be given hereunder shall be in writing and shall be deemed to have been duly given if (a) (i) delivered personally, (ii) mailed (postage prepaid) by certified mail (in this case, notice to be deemed given three days after mailing), or (iii) delivered by a recognized commercial courier to the Party entitled thereto at the address set forth below or such other address as such Party shall have designated by five days' notice to the other; and (b) delivered by email, as set forth below:

If to Seller, to: Express Lane, Inc. 2101 Northside Drive, Suite 104

Panama City, FL 32405 Attention: James E. Lewis, III Email: jelfish@comcast.net

With a copy to: Glenn A. Adams, Esq.

Holland & Knight LLP

200 South Orange Avenue, Suite 2600

Orlando, FL 32801

Email: glenn.adams@hklaw.com

If to Buyer, to: LGP Realty Holdings LP

702 Hamilton Street, Suite 203

Allentown, PA 18101

Email: jtopper@lehighgas.com Attention: Joseph V. Topper, Jr.

25

With a copy to: Richard A. Silfen, Esq.

Duane Morris LLP 30 South 17th Street Philadelphia, PA 19103

Email: rasilfen@duanemorris.com

**11.17** Following Closing, Seller and Buyer each agree to execute any and all documents and take such further actions as may be reasonably necessary to effectuate the purposes of this Agreement.

# [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

Signature Page Follows

26

IN WITNESS WHEREOF, the parties have executed this Agreement as of the last date set forth below.

SELLER:

EXPRESS LANE, INC., a Florida corporation

By: /s/ James E. Lewis III
Name: James E. Lewis III

Title: President

BUYER:

LGP REALTY HOLDINGS LP, a Delaware limited partnership

By: LGP Realty Holdings GP LLC, a Delaware limited liability company,

its General Partner

By: /s/ Joseph V. Topper, Jr.

Name: Joseph V. Topper, Jr.

Title: President



# Lehigh Gas Partners LP Completes Acquisition of 45 Sites in Florida Panhandle region

**ALLENTOWN, PA (December 24, 2012)** — Lehigh Gas Partners LP (NYSE:LGP), announced today it has completed the acquisition of Express Lane, Inc. of Lynn Haven, Florida, including Express Lane's 45 motor fuel sites in the Florida "panhandle" region. The aggregate purchase price for the Express Lane property and the Express Lane Shares is \$43 million, subject to certain post-closings adjustments. The acquisition was fully financed with Lehigh Gas Partners' revolving credit facility.

The 45 fuel sites are concentrated in the Tallahassee/Panama City metropolitan area and along the Interstate 10 corridor of the "panhandle" region located in northwest Florida. Seven of the sites are fee simple interests and 38 are leasehold interests.

Forty-three of the 45 sites sell motor fuel, of which 41 sites sell branded motor fuel, including 25 Chevron sites and 16 ExxonMobil sites. Two sites sell unbranded motor fuel. The remaining two sites are real estate acquisitions. The motor fuel volume for the 43 sites totaled more than 42 million gallons in 2011.

Lehigh Gas — Ohio, LLC, an entity managed by Joe Topper, Chairman and CEO of Lehigh Gas Partners, will operate Express Lane's gasoline and diesel retail outlet business, and its related convenience store business. Lehigh Gas — Ohio will lease the Express Lane sites from Lehigh Gas Partners. Lehigh Gas — Ohio paid Lehigh Gas Partners \$1,000,000 in exchange for Express Lane's retail business assets.

Lehigh Gas Partners estimates it will receive aggregate rental income, net of expenses, of approximately \$4.6 million per year, with an average lease term of up to 20 years. In addition, 14 of the sites have quick serve restaurants with the following national brands: Subway, Domino's, Pizza Hut and Hardee's.

"The acquisition of Express Lane helps us accomplish two key goals pursuant to our overall growth strategy," Topper said. "We achieve expansion beyond our core geographic footprint in the Northeast and establish a foothold in a highly-trafficked metro market.

"We also welcome Chevron, a major fuel brand, to our portfolio," Topper said. "Chevron has a strong market presence in the southeast."

#### **About Lehigh Gas Partners**

Lehigh Gas Partners, headquartered in Allentown, PA, was formed to engage in the wholesale distribution of motor fuels and to own and lease real estate used in the retail distribution of motor fuels. Lehigh Gas Partners owns and leases sites located in Pennsylvania, New Jersey, Ohio, New York, Massachusetts, Kentucky, New Hampshire and Maine. For additional information, please visit www.lehighgaspartners.com

# Forward-Looking Statements

This news release contains "forward-looking statements," such as statements regarding the anticipated impact of acquiring operating assets and real estate currently owned by Express Lane, Inc. These forward-looking statements are based on current plans and expectations and involve a number of risks and uncertainties that could cause actual results and events to vary materially. For a full discussion of these risks and uncertainties, please refer to the "Risk Factors" section of the Registration Statement on Form S-1 initially filed by Lehigh Gas Partners LP on May 11, 2012 and the information included in subsequent amendments and other filings. These forward-looking statements are based on and include our expectations as of the date hereof. Subsequent events and market developments could cause our expectations to change. While we may elect to update these forward-looking statements at some point in the future, we specifically disclaim any obligation to do so, even if new information becomes available, except as may be required by applicable law.

###

#### **CONTACTS:**

#### **INVESTORS**

Karen Yeakel Vice President, Investor Relations Lehigh Gas Partners 610-625-8126 kyeakel@lehighgas.com

#### MEDIA

Pete Waldron Vice President, Public Affairs, Corporate Communications & Philanthropy Lehigh Gas Partners (610) 625-8073 pwaldron@lehighgas.com



#### Lehigh Gas Partners LP Completes Acquisition of 24 Sites in the Scranton/Wilkes-Barre Market

**ALLENTOWN, PA (December 24, 2012)** — Lehigh Gas Partners LP (NYSE:LGP) announced today it has completed its acquisition of 24 sites in the metro area of Scranton and Wilkes-Barre, Pennsylvania from Dunmore Oil Company, Inc. and JoJo Oil Company, Inc., for \$28.5 million cash. Dunmore Oil Company, Inc. and JoJo Oil Company, Inc. are related companies headquartered in Dunmore, Pennsylvania. The acquisition was fully financed with Lehigh Gas Partners' revolving credit facility.

The acquired sites are strategically located along highly-traveled interstate intersections on Routes 84, 81 and 380 in northeast Pennsylvania. The sites market either ExxonMobil or Valero branded fuel. Twenty-three of the sites are fee simple interests and one is a leasehold interest.

The 24 sites, which previously served as sub-jobber locations for Lehigh Gas Partners, sold an aggregate of approximately 28 million gallons of motor fuel in 2011. Lehigh Gas Partners expects the sites will have margins comparable to other Lehigh Gas Partners' controlled fuel locations. In addition, Lehigh Gas Partners estimates it will receive aggregate rental income, net of expenses, of approximately \$1.7 million per year from these 24 sites.

Purchasing the Dunmore and JoJo sites helps Lehigh Gas Partners achieve a key goal pursuant to its overall growth strategy: acquiring additional fuel locations within its current geographic footprint in the northeast United States. Lehigh Gas Partners, headquartered in Allentown, Pa., owns and leases sites located in Pennsylvania, New Jersey, Ohio, New York, Massachusetts, Kentucky, New Hampshire and Maine.

"This transaction illustrates our commitment to the execution of our overall growth strategy," Joe Topper, Chairman and CEO of Lehigh Gas Partners LP, said. "We believe that the Dunmore and JoJo sites fit nicely into our existing portfolio and strengthen our presence in the highly-traveled corridor of northeastern Pennsylvania."

#### **About Lehigh Gas Partners**

Lehigh Gas Partners, headquartered in Allentown, PA, was formed to engage in the wholesale distribution of motor fuels and to own and lease real estate used in the retail distribution of motor fuels. Lehigh Gas Partners owns and leases sites located in Pennsylvania, New Jersey, Ohio,

New York, Massachusetts, Kentucky, New Hampshire and Maine. For additional information, please visit www.lehighgaspartners.com

#### **Forward-Looking Statements**

This news release contains "forward-looking statements," such as statements regarding the anticipated impact of acquiring the operating assets and real estate previously owned by Dunmore Oil Company and JoJo Oil Company. These forward-looking statements are based on current plans and expectations and involve a number of risks and uncertainties that could cause actual results and events to vary materially. For a full discussion of these risks and uncertainties, please refer to the "Risk Factors" section of the Registration Statement on Form S-1 initially filed by Lehigh Gas Partners LP on May 11, 2012 and the information included in subsequent amendments and other filings. These forward-looking statements are based on and include our expectations as of the date hereof. Subsequent events and market developments could cause our expectations to change. While we may elect to update these forward-looking statements at some point in the future, we specifically disclaim any obligation to do so, even if new information becomes available, except as may be required by applicable law.

###

# **CONTACTS:**

#### **INVESTORS**

Karen Yeakel Vice President, Investor Relations Lehigh Gas Partners 610-625-8126 kyeakel@lehighgas.com

#### **MEDIA**

Pete Waldron Vice President, Public Affairs, Corporate Communications & Philanthropy Lehigh Gas Partners (610) 625-8073 pwaldron@lehighgas.com